Royal





Volume 230, No. 29 July 21, 2021 Halifax, Nova Scotia Published by Authority

Part I

Index of Notices

Orders in Council OIC 2021-204 (Proclamation, dissolution of General Assembly and fixing dates for Writs of Election and ordinary polling day)......1533 **Companies Act** 3251177 Nova Scotia Limited......1534 AMAG Pharmaceuticals Canada Corporation1534 Gerbro Holdings Company1534 Jjohn Enterprises Ltd......1535 Lumarr Property Improvement Limited1535 Murray Auto Sales Yarmouth Ltd......1535 PAH Real Estate Investments Nova Scotia ROVOP Canada Inc.1535 **Fisheries and Coastal Resources Act Health Protection Act** Order of the Chief Medical Officer of Health (July 14, 2021)......1576 Order of the Chief Medical Officer of Health (July 16, 2021)......1583 **Notaries and Commissioners Act** Commissioner Appointments.......1541 **Probate Act** Harnish, Edith Read1536 MacDonald, Ronald Joseph......1536 McLaughlin, Gail Arlene1536 Morrone, Enrico Caruso......1537 **Probate Act – Estate Notices** Estate Notices (subsequent)......1555 **Ouieting Titles Act**

ORDER IN COUNCIL 2021-204 Dated July 17, 2021

The Governor in Council on the report and recommendation of the President of the Executive Council dated July 17, 2021, and pursuant to Section 29 of Chapter 5 of the Acts of 2011, the *Elections Act*, is pleased to order that the General Assembly of Nova Scotia be dissolved and that an election be instituted in all of the Electoral Districts of Nova Scotia, and is further pleased to fix the date of the writ of election as Saturday July 17, 2021, to fix the date of election day as Tuesday, August 17, 2021, and to order that a Proclamation be issued accordingly.

Canada Province of Nova Scotia

Elizabeth the Second, by the Grace of God, of the United Kingdom, Canada, and Her Other Realms and Territories, Queen, Head of the Commonwealth, Defender of the Faith.

To all whom these presents shall come, or whom the same may in any wise concern,

Greeting!

A Proclamation

Whereas we have thought fit by and with the advice of Our Executive Council for Nova Scotia to dissolve the present General Assembly. We do for that end publish this Our Royal Proclamation and do hereby dissolve the said General Assembly accordingly, and the Members of the House of Assembly are discharged from their meeting and attendance.

And We being desirous and resolved as soon as may be to meet Our People of Our Province of Nova Scotia, and to have their advice in General Assembly, do hereby make known Our Royal Will and Pleasure to call a General Assembly, and do hereby further declare that by and with the advice of our said Executive Council, We have this day given orders for the issuing of Our Writs in due form for the election of Members to serve in the House of Assembly for the several Electoral Districts of the Province, which Writs are to bear the date the 17th day of July, 2021, and the date of election day be Tuesday the 17th day of August 2021.

In Testimony Whereof, We have cause these Our Letters to be made Patent and the Great Seal of the Province of Nova Scotia to be hereunto affixed.

Witness, Our Trusty and Well Beloved, Arthur J. LeBlanc, Chancellor of Our Order of Nova Scotia, one of Her Majesty's Counsel learned in the law in the Province of Nova Scotia, Lieutenant Governor in and of Our Province of Nova Scotia.

Given at **Government House** in the Halifax Regional Municipality this 17th day of July in the year of Our Lord two thousand and twenty-one and in the 70th year of Our Reign.

By Command

Honourable Randy Delorey PROVINCIAL SECRETARY ATTORNEY GENERAL AND MINISTER OF JUSTICE

RG July 21, 2021

IN THE MATTER OF: The Companies Act, R.S.N.S.
 1989, c. 81 - and - An Application by 3007947 Nova
 Scotia Limited for Leave to Surrender its Certificate of Incorporation

3007947 Nova Scotia Limited hereby gives notice pursuant to the provisions of Section 137 of the *Companies Act* that it intends to make application to the Nova Scotia Registrar of Joint Stock Companies for leave to surrender its Certificate of Incorporation.

Dated the 16th day of July, 2021.

Alexandria Barnes McInnes Cooper 1300-1969 Upper Water Street Purdy's Wharf Tower II Halifax NS B3J 3R7 Solicitor for 3007947 Nova Scotia Limited

RG July 21, 2021

IN THE MATTER OF: The Companies Act, R.S.N.S.
 1989, c. 81 - and - An Application by 3251177 Nova
 Scotia Limited for Leave to Surrender its Certificate of Incorporation

3251177 Nova Scotia Limited (the "Company") hereby gives notice pursuant to the provisions of Section 137 of the *Companies Act* (Nova Scotia) that it intends to make application to the Registrar of Joint Stock Companies for the Province of Nova Scotia for leave to surrender its Certificate of Incorporation of the Company.

Dated at Halifax, Nova Scotia, on the 13th day of July, 2021.

Jared B. Schwartz
Patterson Law
Solicitor for 3251177 Nova Scotia Limited

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Advantage New Homes Limited** for Leave to Surrender its Certificate of Incorporation

NOTICE is hereby given that Advantage New Homes Limited intends to make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Incorporation.

Dated this 14th day of July 2021.

Brian S. Creighton Solicitor for Advantage New Homes Limited

RG July 21, 2021

IN THE MATTER OF: The Companies Act, R.S.N.S. 1989, c. 81 - and - An Application by AMAG Pharmaceuticals Canada Corporation for Leave to Surrender its Certificate of Incorporation

NOTICE is hereby given that AMAG Pharmaceuticals Canada Corporation intends to make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Incorporation.

Dated July 21, 2021.

Kimberly Bungay Stewart McKelvey Solicitor for AMAG Pharmaceuticals Canada Corporation

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Gerbro Holdings Company** for Leave to Surrender its Certificate of Amalgamation

NOTICE is hereby given that Gerbro Holdings Company intends to make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Amalgamation.

Dated this July 21, 2021.

Charles S. Reagh Stewart McKelvey Solicitor for Gerbro Holdings Company

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Green Cove Lobster Ltd.** for Leave to Surrender its Certificate of Incorporation

Green Cove Lobster Ltd., a body corporate, of Yarmouth, Nova Scotia, hereby gives notice that it intends to apply to the Registrar of Joint Stock Companies of the Province of Nova Scotia, for leave to surrender its Certificate of Incorporation and leave to have its name struck from the Companies Register, pursuant to the provisions of Section 137 of the *Companies Act*.

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Jjohn Enterprises Ltd.** for Leave to Surrender its Certificate of Incorporation

NOTICE is hereby given that Jjohn Enterprises Ltd. intends to make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Incoroporation.

Dated this 13th day of July 2021.

Brian S. Creighton Solicitor for Jjohn Enterprises Ltd.

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Lumarr Property Improvement Limited** for Leave to Surrender its Certificate of Incorporation

NOTICE is hereby given that Lumarr Property Improvement Limited will make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Incorporation.

Dated at Halifax, Nova Scotia this 15th day of July, 2021.

Lu Feng, on behalf of Lumarr Property Improvement Limited

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Murray Auto Sales Yarmouth Ltd.** for Leave to Surrender its Certificate of Incorporation

NOTICE is hereby given that Murray Auto Sales

Yarmouth Ltd. intends to make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Incorporation.

Dated this 14th day of July, 2021.

Gregory D. Barro, QC Solicitor for Murray Auto Sales Yarmouth Ltd.

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **PAH Real Estate Investments Nova Scotia Limited** for Leave to Surrender its Certificate of Incorporation

PAH Real Estate Investments Nova Scotia Limited (the "Company") hereby gives notice pursuant to the provisions of Section 137 of the *Companies Act* (Nova Scotia) that it intends to make application to the Registrar of Joint Stock Companies for the Province of Nova Scotia for leave to surrender its Certificate of Incorporation of the Company.

Dated at Truro, Nova Scotia, on the 19th day of July, 2021.

Jennifer J. Hamilton Upham Patterson Law Solicitor for PAH Real Estate Investments Nova Scotia Limited

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **ROVOP**Canada Inc. for Leave to Surrender its Certificate of Incorporation

ROVOP Canada Inc. hereby gives notice pursuant to the provisions of Section 137 of the *Companies Act* that it intends to make application to the Registrar of Joint Stock Companies of the Province of Nova Scotia for leave to surrender the Certificate of Incorporation of the Company.

Dated this 15th day of July 2021.

Marc T. LeClair BOYNECLARKE LLP 99 Wyse Road, Suite 600 Dartmouth NS B3A 4S5 Solicitor for ROVOP Canada Inc.

RG July 21, 2021

IN THE MATTER OF: The *Companies Act*, R.S.N.S. 1989, c. 81 - and - An Application by **Wardour Learning Inc.** for Leave to Surrender its Certificate of Incorporation

NOTICE is hereby given that Wardour Learning Inc. intends to make an application to the Registrar of Joint Stock Companies for leave to surrender its Certificate of Incorporation.

Dated at Kentville, Nova Scotia, this 21st day of July, 2021.

RG July 21, 2021

IN THE MATTER OF: The *Probate Act*, S.N.S. 2000, c. 31 - and - In the Estate of **Edith Read Harnish**, Deceased

IN THE COURT OF PROBATE FOR NOVA SCOTIA Notice of Application (S. 64(3)(a))

The applicant, Blake Hirtle, Executor in an unsigned draft Last Will and Testament sought to be proven has applied to the Registrar of the Probate Court of Nova Scotia, at the Probate District of Lunenburg, 141 High Street, Bridgewater, Nova Scotia B4V 1W2 for proof of the said unsigned Last Will and Testament of Edith Read Harnish in solemn form prepared in October 2020 to be heard on the 30th day of September, 2021 at 9:30 AM.

The affidavit of Blake Hirtle sword on the 17th day of June, 2021 in Form 46, a copy of which is attached to this Notice of Application, is filed in support of this application. Other materials may be filed and will be delivered to you or your lawyer before the hearing.

NOTICE: If you contest any part of the application you must complete and file a notice of objection in Form 47 with the court, and then serve the notice of objection on the applicant and the personal representative.

If you do not file and serve a notice of objection you will not be entitled to any notice of further proceedings and you may only make representations at the hearing with the permission of the registrar or judge.

If you do not come to the hearing in person or as represented by your lawyer, the court may give the applicant what they want in your absence. You will be bound by any order the court makes.

Therefore, if you contest any part of this application, you or your lawyer must file and serve a notice of objection in Form 47 and come to the hearing.

Dated June 17th, 2021.

David St. Clair Bond 85 Fleet Road

Hubbards NS B0J 1T0 Telephone: 902-858-3066; Fax: 902-820-3957

Email: dbond@davidbondlaw.com

RG July 21, 2021 (3 issues)

IN THE MATTER OF: The Probate Act, R.S. c. 359, s. 1

IN THE COURT OF PROBATE FOR NOVA SCOTIA

Notice of Citation to Close

IN THE ESTATE OF: Ronald Joseph MacDonald, Deceased

To: HEIRS, LEGATEES, CREDITORS, NEXT OF KIN AND PARTIES In any way interested in the Estate of Ronald Joseph MacDonald, late of Pleasant Valley, Antigonish County, Nova Scotia, deceased

WHEREAS Jane Adams, of 5948 Highway 337, Cape George, in the County of Antigonish, Province of Nova Scotia, B2G 2L2 Executrix of the above named deceased, presented her petition praying that a day may be fixed for the passing of her account as such Executrix and for the final settlement of the said Estate.

YOU ARE THEREFORE entitled to appear before the Court of Probate, at Antigonish, in the County of Antigonish, on Wednesday the 18th day of August A.D., 2021, at 10:00 o'clock in the forenoon to attend the adjudication of the claims of creditors, or other persons if any, the taking of the accounts of the Executor and the distribution of the estate according to law, and to show cause, if you have any, why said account should not be so passed and the said estate finally settled.

Given under the Seal of the said Court this 23rd day of June A.D., 2021.

Elizabeth MacLeod Registrar of Probate

RG July 7, 2021 (5 issues)

IN THE MATTER OF: The *Probate Act*, S.N.S. 2000, c. 31 - and - In the Estate of **Gail Arlene McLaughlin**, Deceased

IN THE COURT OF PROBATE FOR NOVA SCOTIA Notice of Application (S. 64(3)(a))

The applicants, William Stuart McLaughlin and Michael Grant McLaughlin, co-executors and co-trustees and the being [the] residuary beneficiaries of the Last Will and Testament of Gail Arlene McLaughlin have applied to the Registrar of the Probate Court of Nova Scotia, at the

Probate District of Halifax, Court Administration Office, The Law Courts, 2nd Floor, 1815 Upper Water Street, Halifax, Nova Scotia, B3J 1S7, for an Order waiving the service requirements pursuant to Regulation 65(1) of the *Probate Act*, and proving the Last Will and Testament of the deceased in Solemn Form pursuant to Section 31 of the *Probate Act* and Regulation 69 (1) and (2) enacted pursuant to the *Probate Act* to be heard on August 30, 2021 at 9:30 a.m.

The affidavits of William Stuart McLaughlin, Michael Grant McLaughlin and James Scott McLaughlin, who are all of the persons who have an interest in this estate, in Form 46, copies of which are attached to this Notice of Application, is [are] filed in support of this application. Other materials may be filed and will be delivered to you or your lawyer before the hearing.

NOTICE: If you contest any part of the application you must complete and file a notice of objection in Form 47 with the court, and then serve the notice of objection on the applicant and the personal representative.

If you do not file and serve a notice of objection you will not be entitled to any notice of further proceedings and you may only make representations at the hearing with the permission of the registrar or judge.

If you do not come to the hearing in person or as represented by your lawyer, the court may give the applicant what they want in your absence. You will be bound by any order the court makes.

Therefore, if you contest any part of this application, you or your lawyer must file and serve a notice of objection in Form 47 and come to the hearing.

Dated July 19, 2021.

Cassandra P. Taylor BOYNECLARKE LLP 99 Wyse Road, Suite 600 PO Box 876 Dartmouth NS B2Y 3Z5

Telephone: 902-469-9500; Fax: 902-463-7500

Email: ctaylor@boyneclarke.ca

RG July 21, 2021 (3 issues)

IN THE MATTER OF: The *Probate Act*, S.N.S. 2000, c. 31 - and - In the Estate of **Enrico Caruso Morrone**, Deceased

IN THE COURT OF PROBATE FOR NOVA SCOTIA Amended Notice of Application (S. 64(3)(a)) (Amended July 20, 2021)

The applicant Cristoforo Columbo Morrone, the

Executor named in the attached Will and the brother of the deceased, has applied to the Regsitrar of Probate of Nova Scotia, at the Probate District of Halifax, 1815 Upper Water Street, Halifax, Nova Scotia for an order granting to him Probate of the Estate of Enrico Caruso Morrone based on proof in solemn form of the Will, a photocopy of which is attached hereto, to be heard on August 6, 2021 at 9:30 a.m.

The affidavit of Cristoforo Columbo Morrone in Form 46, and the affidavit of Wanda L. Evans, <u>have been filed with the Court.</u> copy of which is attached to this Notice of Application, is filed in support of this application. Other materials may be filed and will be delivered to you or your lawyer before the hearing.

NOTICE: If you contest any part of the application you must complete and file a notice of objection in Form 47 with the court, and then serve the notice of objection on the applicant and the personal representative.

If you do not file and serve a notice of objection you will not be entitled to any notice of further proceedings and you may only make representations at the hearing with the permission of the registrar or judge.

If you do not come to the hearing in person or as represented by your lawyer, the court may give the applicant what they want in your absence. You will be bound by any order the court makes.

Therefore, if you contest any part of this application, you or your lawyer must file and serve a notice of objection in Form 47 and come to the hearing.

Dated July 19, 2021.

Andrew Christofi Noseworthy Di Costanzo Diab 6470 Chebucto Road Halifax NS B3L 1L4

Telephone: 902-444-4747 ext. 36; Fax: 902-444-4301

Email: achristofi@nddlaw.com

RG July 21, 2021 (3 issues)

IN THE MATTER OF: The *Quieting Titles Act*, R.S.N.S. 1989, c. 382

Supreme Court of Nova Scotia

Between: Timothy Vincent Delorey and Kelly Ann McGrath-Delorey of 10564 Highway 4, Monastery, Nova Scotia, Plaintiffs - and - Chantel Rachelle O'Neil of 9225 Highway 4, Tracadie, Nova Scotia, and Majesty The Queen in Right of the Province as Represented by the Attorney General of Nova Scotia, Defendants

Notice of Claim under Quieting Titles Act

In the captioned proceeding, Timothy Vincent Delorey and Kelly Ann McGrath-Delorey claim a certificate of title for the following land:

All AND SINGULAR that certain lot situated, lying and being at 9195 Highway 4, Tracadie, Antigonish County Nova Scotia being PID number 01243088 in the Land Registration Office of the County of Antigonish.

A person who fears they may be adversely affected by such a certificate may contest the claim by making a motion in court before the 13th day of August, 2021 to be

joined as a defendant. The motion documents may be filed with the prothonotary at the Courthouse, 11 James Street, Antigonish, Nova Scotia. Copies must be delivered to the undersigned and the other parties at the same time as they are filed.

Signed July 12th, 2021.

Harold A. MacIsaac, Solicitor for the Plaintiffs 409 Granville Street Port Hawkesbury NS B9A 2M5

RG July 14, 2021 (4 issues)

Administrative Decisions

Nova Scotia Department of Fisheries and Aquaculture *Fisheries and Coastal Resources Act*, S.N.S. 1996, c. 25

The Nova Scotia Department of Fisheries and Aquaculture invites members of the public to provide written comment on applications for Aquaculture Licences and Leases. Selected details of the application(s) are as follows, with further information also found at http://novascotia.ca/fish/aquaculture/public-information/.

Future Seafoods Inc.

AO#1351

Amendment Application

Location: Caribou Harbour, Pictou County

Type: Marine Size: 17.25 HA

Current Cultivation Method: Bottom shellfish without gear

Current Species: American oyster

Proposed Amendment:

1. Authorize Bottom Cultivation with Gear Method

2. Authorize Suspended Shellfish Cultivation

3. Add additional species – Bay quahog (Mercenaria mercenaria)

4. Change Site Boundaries (no increase in size)

Submission Period: Thursday June 24, 2021 to Friday, July 23, 2021

Summary of Application: Aquaculture Site No. 1351 has been an issued aquaculture site since 2010. The proposed boundary amendment will not result in an increase in the size of the site. The purpose of this amendment is to relocate the portion of the site that is not useable (due to depth and presence of marine plants) to the adjacent area that is deemed more viable to the growth of shellfish, through the use of both suspended gear and bottom cultivation with gear. Bottom cultivation without gear would continue in some areas of the site.

Please note that this application is being reviewed pursuant to the *Canadian Navigable Waters Act* by Transport Canada. Written comments regarding the effect of this work on marine navigation may be submitted as follows:

1. Online at: http://cps.canada.ca/ under the Registry/NPP#2010-200164

2. By Mail at: Manager

Transport Canada - Navigation Protection Program

P.O. Box 42

Moncton NB E1C 8K6

Michael A. MacIntosh

AQ#1360

Amendment Application

Location: Merigomish Harbour, Pictou County

Type: Marine **Size:** 7.91 HA

Current Cultivation Method: Bottom cultivation without gear

Current Species: American oyster, Bay Quahog

Proposed Amendment: Authorize bottom shellfish cultivation with gear method and authorize suspended shellfish cultivation method

Submission Period: Thursday June 24, 2021 to Friday, July 23, 2021

Summary of Application: AQ#1360 Licence and Lease were issued to Michael A. MacIntosh on September 12, 2011 for a ten year term from August 1, 2012 to August 1, 2022. A review of the applicant's Farm Management Plan indicated that the applicant was conducting bottom shellfish aquaculture with gear, without the appropriate approvals in place for this type of aquaculture. The Farm Management Plan was conditionally approved for a six-month period, during which time the applicant was required to submit a gear amendment application for review. The intent of the amendment application to "authorize bottom shellfish cultivation with gear method" is for the applicant to become fully compliant with all relevant regulations and licensing and leasing requirements related to bottom shellfish aquaculture with gear. Should the application not be approved, the applicant will be required to adhere to the current licensing and leasing provisions in place and ensure the removal of any unauthorized gear.

The applicant has also requested the amendment to Licence and Lease AQ#1360 to authorize the Suspended Shellfish Cultivation method for American oyster and Bay quahog, and currently is not conducting this cultivation method.

Brian Lewis and Dale Webb

AQ#1380

Amendment Application

Location: Caribou Harbour, Pictou County

Type: Marine **Size:** 11.47 HA

Current Cultivation Method: Bottom shellfish without gear

Current Species: American oyster

Proposed Amendment:

1. Authorize Bottom Cultivation with Gear Method

- 2. Authorize Suspended Shellfish Cultivation
- 3. Change Site Boundaries (no increase in size)

Submission Period: Thursday June 24, 2021 to Friday, July 23, 2021

Summary of Application: Aquaculture Site No. 1380 has been an issued aquaculture site since 2014. The purpose of this amendment is to relocate the portion of the site that is not useable (due to heavy marine traffic that is reportedly creating too much wave action to allow sustainable development) to the adjacent area that is deemed more viable to the growth of shellfish, through the use of both suspended gear and bottom cultivation with gear. Bottom cultivation without gear would continue in some areas of the site.

Please note that this application is being reviewed pursuant to the *Canadian Navigable Waters Act* by Transport Canada. Written comments regarding the effect of this work on marine navigation may be submitted as follows:

1. Online at: http://cps.canada.ca/ under the Registry/NPP#2013-200042

2. By Mail at: Manager

Transport Canada - Navigation Protection Program

P.O. Box 42

Moncton NB E1C 8K6

Future Seafoods Inc.

AQ#1386

Amendment Application

Location: Tracadie, West Arm, Antigonish County

Type: Marine Size: 8.16 HA

Current Cultivation Method: Bottom shellfish without gear

Current Species: American oyster

Proposed Amendment: Authorize Bottom Shellfish with Gear Cultivation Method

Submission Period: Thursday June 24, 2021 to Friday, July 23, 2021

Summary of Application: Aquaculture Site No. 1386 has been an issued aquaculture site since November 1, 2014. A review of the previous applicant's Farm Management Plan indicated that the previous applicant was conducting bottom shellfish aquaculture

with gear, without the appropriate approvals in place for this type of aquaculture. The Farm Management Plan was conditionally approved on September 7, 2018 for a sixmonth period, during which time the previous applicant was required to submit a gear amendment application for review. The intent of this amendment application is for the current applicant to become fully compliant with all relevant regulations and licensing and leasing requirements related to bottom shellfish aquaculture with gear. Should the application not be approved, the current applicant will be required to adhere to the current licensing and leasing provisions in place and ensure the removal of any unauthorized gear.

Please note that this application is being reviewed pursuant to the *Canadian Navigable Waters Act* by Transport Canada. Written comments regarding the effect of this work on marine navigation may be submitted as follows:

1. Online at: http://cps.canada.ca/ under the Registry/NPP#2012-200389

2. By Mail at: Manager

Transport Canada - Navigation Protection Program

P.O. Box 42

Moncton NB E1C 8K6

Written submissions will be accepted from 12:00 AM to 11:59 PM during the submission period noted above. To be considered by the Administrator, a written submission from a member of the public must:

- Include the name, mailing address, civic address, phone number, and email of the person making the submission;
- Meet the criteria set out in Section 41 of the Aquaculture License and Lease Regulations; and
- Meet the guidelines outlined at http://novascotia.ca/fish/aquaculture/public-information/

Written comments may be submitted as follows:

1. By Mail at: Aquaculture Administrator

Department of Fisheries and Aquaculture

1575 Lake Road

Shelburne NS B0T 1W0

2. By Fax at: 902-875-7429

3. By Email at: Aqua.Admin@novascotia.ca

Comments received from the public that meet the submission requirements will be posted on the Department's website when the Administrator's decision concerning that application is made and communicated.

Comments as posted will include the name of the person submitting the comment as well as their community and province/state. The street address, email address and phone number of the person submitting the comment will be removed.

By submitting your comments with respect to an application you are consenting to the posting at the Department's website of the above-noted information. You also agree that your comments will comply with the terms applicable to the use of Government of Nova Scotia websites (a copy of those terms is available at https://novascotia.ca/terms).

The Department is not responsible for the content of your posted comments, including, but not limited to, the accuracy, reliability or currency of the information contained therein. Written information provided to the Department as submissions are subject to the protections and disclosures required by the Nova Scotia *Freedom of Information and Protection of Privacy Act*.

Commissioner Appointments

Nova Scotia Department of Justice Notaries and Commissioners Act, R.S.N.S. 1989, c. 312

The Minister of Justice and Attorney General, Randy Delorey, under the authority vested in him by clause 2(b) of Chapter 23 of the Acts of 1996, the *Court and Administrative Reform Act*, Order in Council 2004-84, the *Assignment of Authority Regulations*, N.S. Reg. 21/2004, and Sections 6 and 7 of Chapter 312 of the Revised Statutes of Nova Scotia, 1989, the *Notaries and Commissioners Act*, is hereby pleased to advise of the following changes to commissioner appointments:

Name	Status	Details
Askew, Brayden	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police
Bell, Nicholas A. P.	Appointed	North River, in the County of Colchester While employed with the Province of Nova Scotia (Agriculture, Crown Lending Agencies)
Bezaire, Catherine	Appointed	West River, in the County of Antigonish While employed with the Royal Canadian Mounted Police
Braund, Robert J.	Reappointed	Sweets Corner, in the County of Hants For a term commencing September 19, 2021 and to expire September 18, 2026 (private)
Brown, Quentin C.	Appointed	Dartmouth, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (private)
Browne, Helen	Reappointed	Dartmouth, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (Land Connect, Land Broker and Services)
Bueckert, Michael Nelson	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police
Burgess, Trevor Burton	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police
Carroll, Kimberly	Appointed	Valley, in the County of Colchester While employed with the Royal Canadian Mounted Police
Cormier, Lisa	Appointed	Dartmouth, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Service Nova Scotia and Internal Services – Access Nova Scotia)
Croft, Janice A.	Reappointed	Hebb's Cross, in the County of Lunenburg For a term commencing June 24, 2021 and to expire June 23, 2026 (MacLeod Webber, law firm)
Dempsey, Melanie	Appointed	Hantsport, in the County of Hants For a term commencing June 24, 2021 and to expire June 23, 2026 (Scotiabank, financial institution)
Doucet, Calixta J.	Reappointed	Bedford, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (Waterfront Law, law firm)

Name	Status	Details
Etter, E. Anne	Reappointed	Halifax, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (PricewaterhouseCoopers Inc., accounting firm)
Flemming, Tara	Appointed	Cambridge, in the County of Kings For a term commencing June 24, 2021 and to expire June 23, 2026 (Valley Credit Union, financial institution)
Frazee, Andrea H.	Appointed	Heatherton, in the County of Antigonish While employed with the Province of Nova Scotia (Justice – Court Services – Antigonish Justice Centre)
Fry, Lisa	Appointed	Halifax, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (Exhibit A Bailiff Enforcement, bailiff services)
Goodwin, Kelly A.	Reappointed	Greenville, in the County of Yarmouth For a term commencing August 4, 2021 and to expire August 3, 2026 (Muise Law Inc.)
Goudey, Brandon	Appointed	Howie Centre, in the County of Cape Breton While employed with the Royal Canadian Mounted Police
Gyurcsanyi, Agnes	Appointed	Dartmouth, in the Halifax Regional Municipality While employed with the Halifax Regional Municipality
Head, Aaron	Appointed	Bedford, in the Halifax Regional Municipality While employed with the Halifax Regional Police
Jeddore, Rory	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police
Kelly, Catherine	Reappointed	Dartmouth, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (Drake Recording Services, court transcription)
Kelly, Stephanie	Appointed	Centreville, in the County of Kings For a term commencing June 24, 2021 and to expire June 23, 2026 (Anderson Sinclair, law firm)
Langille, Gillian Marie	Appointed	Hammonds Plains, in the Halifax Regional Municipality While employed with the Royal Canadian Mounted Police
Leblanc, Marion Hannah	Appointed	Halifax, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Service Nova Scotia and Internal Services – Access Nova Scotia)
Lee, Nicole	Appointed	Dartmouth, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Service Nova Scotia and Internal Services – Access Nova Scotia)
Levesque, Nicolas W.	Appointed	Halifax, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (CIR Law, law firm)
Lukeman, Frank	Revoked	Antigonish, in the County of Antigonish No longer employed with the Province of Nova Scotia
MacCallum, Steven	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police

Name	Status	Details
MacDonald, Sheena	Appointed	Trenton, in the County of Pictou While employed with the Province of Nova Scotia (Service Nova Scotia and Internal Services – Access Nova Scotia)
MacKinnon, Katrina	Appointed	Dartmouth, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (CP Distributors Ltd., construction)
McComb, Michael	Appointed	Portuguese Cove, in the Halifax Regional Municipality, For a term commencing June 24, 2021 and to expire June 23, 2026 (private)
McKay, Gillian	Appointed	Fall River, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Public Prosecution Service)
McKee, Derrick	Appointed	Sydney, in the County of Cape Breton While employed with the Royal Canadian Mounted Police
McNair, Rhiannon K.	Appointed	Pictou, in the County of Pictou While employed with the Municipality of Pictou County
McNeil, Joanne	Revoked	Halifax, in the Halifax Regional Municipality Name change to Joanne Sweeney
McRae, Paisley	Appointed	Afton Station, in the County of Antigonish While employed with the Royal Canadian Mounted Police
Mailman, Andrea	Reappointed	Sydney, in the County of Cape Breton For a term commencing June 24, 2021 and to expire June 23, 2026 (Rita Anderson & Associates Inc., bankruptcy trustee)
Marchand, Glen W.	Appointed	Eastern Passage, in the Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (O'Regans National Leasing, vehicle leasing)
Matthews Geoff	Appointed	Whycocomagh, in the County of Inverness While employed with the Royal Canadian Mounted Police
Morris, Christine	Appointed	Halifax, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Public Prosecution Service)
Murphy, Lesley A.	Reappointed	Ellershouse, in the County of Hants For a term commencing November 10, 2021 and to expire November 9, 2026 (Municipal Group of Companies, construction)
Neate, Leslie	Appointed	Dartmouth, in the Halifax Regional Municipality While employed with the Halifax Regional Municipality
Peck, David	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police
Porter, Tania L.	Appointed	Glace Bay, in the County of Cape Breton For a term commencing June 24, 2021 and to expire June 23, 2026 (Crosby Burke Inc., law firm)
Power, Angela R.	Appointed	Halifax, Halifax Regional Municipality For a term commencing June 24, 2021 and to expire June 23, 2026 (IY Law Firm, law firm)

Name	Status	Details
Saif, Noor	Appointed	Bedford, in the Halifax Regional Municipality For a term commending June 24, 2021 and to expire June 23, 2026 (Scotiabank, financial institution)
Sangster, Katheryn	Appointed	Dartmouth, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Service Nova Scotia and Internal Services – Access Nova Scotia)
Smith, Samuel J.	Appointed	Port Hawkesbury, in the County of Inverness While employed with the Royal Canadian Mounted Police
Stairs, Trevor	Reappointed	Dartmouth, in the Halifax Regional Municipality For a term commencing November 10, 2021 and to expire November 9, 2026 (MNP, accounting firm)
Stewart, April	Appointed	Prospect Village, in the Halifax Regional Municipality While employed with the Halifax Regional Municipality
Sweeney, Joanne	Reappointed	Halifax, in the Halifax Regional Municipality While employed with the Halifax Regional Police
Walker, Ian	Appointed	Sydney, in the County of Cape Breton While employed with the Province of Nova Scotia (Correctional Services – Cape Breton Correctional Facility)
Waller, Donna	Appointed	Greenfield, in the County of Queens While employed with the Province of Nova Scotia (Agriculture, Crown Lending Agencies)
Weagle, Kelsey L.	Appointed	Laconia, in the County of Lunenburg For a term commencing June 24, 2021 and to expire June 23, 2026 (Power, Leefe, Reddy & Rafuse, law firm)
Young, Denise A.	Appointed	New Minas, in the County of Kings For a term commencing June 24, 2021 and to expire June 23, 2026 (Annapolis Valley Regional Centre for Education)

Dated at Halifax, Nova Scotia, this 24th day of June, 2021.

Honourable Randy Delorey Attorney General and Minister of Justice

The Minister of Justice and Attorney General, Randy Delorey, under the authority vested in him by clause 2(b) of Chapter 23 of the Acts of 1996, the *Court and Administrative Reform Act*, Order in Council 2004-84, the *Assignment of Authority Regulations*, N.S. Reg. 21/2004, and Sections 6 and 7 of Chapter 312 of the Revised Statutes of Nova Scotia, 1989, the *Notaries and Commissioners Act*, is hereby pleased to advise of the following changes to commissioner appointments:

Name	Status	Details
Atwood, Ashley	Appointed	Clarks Harbour, in the County of Shelburne For a term commencing July 15, 2021 and to expire July 14, 2026 (Donald G. Harding Barrister & Solicitor Inc.)
Bernier, Tamara	Appointed	Conquerall Bank, in the County of Lunenburg For a term commencing July 15, 2021 and to expire July 14, 2026 (Scotia Business Centre Limited, court transcription)

Name	Status	Details
Betz, Adam	Appointed	Sydney Forks, in the County of Cape Breton While employed with the Royal Canadian Mounted Police
Bruce, Stacy T. R.	Appointed	Truro, in the County of Colchester While employed with the Province of Nova Scotia (Fisheries and Aquaculture)
Calder, Linda M.	Revoked	Oxford, in the County of Cumberland No longer employed with the Royal Canadian Mounted Police
Calder, Linda M.	Reappointed	River Philip, in the County of Cumberland For a term commencing July 15, 2021 and to expire July 14, 2026 (private)
Doucet, Josette	Appointed	Danvers, in the County of Digby While employed with the Municipality of the District of Clare
Forgeron, Charlotte G.	Reappointed	Louisbourg, in the County of Cape Breton For a term commencing August 4, 2021 and to expire August 3, 2026 (BDO Canada Limited, bankruptcy)
Frairs, Nellie	Appointed	Brookfield, in the County of Colchester For a term commencing July 15, 2021 and to expire July 14, 2026 (Scotiabank)
Hoeg, Shelley G.	Appointed	River Hebert East, in the County of Cumberland While employed with the Municipality of the County of Cumberland
Laffin, Tammy L. L.	Revoked	Dartmouth, in the Halifax Regional Municipality No longer employed with the Province of Nova Scotia
Laffin, Tammy L. L.	Reappointed	Lake Loon, in the Halifax Regional Municipality For a term commencing July 15, 2021 and to expire July 14, 2026 (Construction Association of Nova Scotia)
MacDonald, Deborah M.	Reappointed	Lingan, in the County of Cape Breton For a term commencing August 4, 2021 and to expire August 3, 2026 (Rita Anderson & Associates Inc., insolvency)
MacDonald, Sara J.	Appointed	Dartmouth, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Service Nova Scotia and Internal Services – Access Nova Scotia)
MacKenzie, Micaela	Appointed	Reserve Mines, in the County of Cape Breton For a term commencing July 15, 2021 and to expire July 14, 2026 (Scotiabank)
MacLeod, Stephanie	Appointed	Baddeck, in the County of Victoria While employed with the Municipality of the County of Victoria
MacRae, Margaret	Reappointed	Lantz, in the County of Hants For a term commencing July 21, 2021 and to expire July 20, 2026 (The Yuill Law Firm)
Marinelli, Brian	Appointed	Sydney, in the County of Cape Breton While employed with the Royal Canadian Mounted Police
Orr, Rebecca	Appointed	Glace Bay, in the County of Cape Breton For a term commencing July 15, 2021 and to expire July 14, 2026 (Scotiabank)

Name	Status	Details
Sajko, Wayne	Appointed	Halifax, in the Halifax Regional Municipality While employed with the Province of Nova Scotia (Infrastructure and Housing)
Spicer, Jessica L.	Appointed	New Germany, in the County of Lunenburg While employed with the Bridgewater Police Service
Stevens, Gaetanne	Appointed	Eskasoni, in the County of Cape Breton While employed with the Royal Canadian Mounted Police
Teran, Sandra	Appointed	Windsor Junction, in the Halifax Regional Municipality For a term commencing July 15, 2021 and to expire July 14, 2026, and only while employed with Canada Border Services Agency
Topping, Eric	Appointed	Stillwater Lake, in the Halifax Regional Municipality For a term commencing July 15, 2021 and to expire July 14, 2026, and only while employed with Canada Border Services Agency
Trider, Christy	Appointed	Fort Ellis, in the County of Colchester For a term commencing July 15, 2021 and to expire July 14, 2026 (Marid Industries Ltd., construction)
White, Paul D.	Reappointed	Dartmouth, in the Halifax Regional Municipality For a term commencing September 22, 2021 and to expire September 21, 2026 (Ryson Interior Construction Limited)
Woodsworth, Kathy	Appointed	Newcombville, in the County of Lunenburg For a term commencing July 15, 2021 and to expire July 14, 2026 (Scotia Business Centre Limited, court transcription)

Dated at Halifax, Nova Scotia, this 15th day of July, 2021.

Honourable Randy Delorey Attorney General and Minister of Justice

Estate Notices

Probate Act, S.N.S. 2000, c. 31

All persons having legal demands against any of the estates listed in this issue must file a notice of claim in Form 32 no later than 6 months from the date of the first advertisement. All persons indebted to an estate must make immediate payment to the personal representative noted.

ESTATE NOTICES BEING PUBLISHED FOR THE FIRST TIME

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
ABBOTT, Deborah Ann Halifax, Halifax Regional Municipality July 5, 2021	Public Trustee PO Box 685 Halifax NS B3J 2T3	Adrienne Bowers Public Trustee PO Box 685 Halifax NS B3J 2T3 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
ALLEN, Cynthia Marie Halifax, Halifax Regional Municipality January 27, 2021	Howard Benson James 39 Briarwood Crescent Halifax NS B3M 1P2	Michael F. Feindel Feindel Law Office 401-7020 Mumford Road Halifax NS B3L 4S9 July 21, 2021 (6m)
BARRY, John Robert Halifax, Halifax Regional Municipality July 14, 2021	Christine Mary Seidl 70 Boul St. Jean Pointe-Claire QC H9S 4Z1	Jacqueline R. Farrow Farrow Hicks Law Inc. 5445 St. Margaret's Bay Road Upper Tantallon NS B3Z 2H9 July 21, 2021 (6m)
BOUDREAU, Edith Cavell MacGillivray Guest Home Sydney, Cape Breton Regional Municipality June 17, 2021	Public Trustee PO Box 685 Halifax NS B3J 2T3	Adrienne Bowers Public Trustee PO Box 685 Halifax NS B3K 2T3 July 21, 2021 (6m)
BOUDREAU, Mary Louise Petit de Grat, Richmond County July 7, 2021	Senley Tyson Boudreau 174 Cape Rouge Road Petit de Grat, NS B0E 2L0	Maurice Boudreau, JD EMM Law Incorporated 409 Granville Street Port Hawkesbury NS B9A 2M5 July 21, 2021 (6m)
BRUSHETT, Lucy Katherine Kentville, Kings County February 25, 2021	Roxanne Tina Riles 16 Talon Drive North Kentville NS B4N 0A9	Kent W. Rodgers Casey Rodgers Chisholm Penny Duggan LLP 201-219 Waverley Road Dartmouth NS B2X 2C3 July 21, 2021 (6m)
BUTLER, Kathleen Marion Dartmouth, Halifax Regional Municipality July 13, 2021	Teresa Ann Hurley 39 Burnaby Lane Dartmouth NS B2W 0H4	Lauren M. Randall BOYNECLARKE LLP 600-99 Wyse Road Dartmouth NS B3A 4S5 July 21, 2021 (6m)
CHASSE, Leopold Robert Lower Sackville, Halifax Regional Municipality March 16, 2021	Jean Chasse 1340 Myra Road Porters Lake NS B3E 1G9	G. Bernard Conway 448 Main Street Kentville NS B4N 1K8 July 21, 2021 (6m)
CHING, Mary Cecilia Dartmouth, Halifax Regional Municipality July 5, 2021	Shelagh Bernadette Wolfe 185 Donegal Drive Dartmouth NS B2V 2N3	Blair MacKinnon Heritage House Law Office 92 Ochterloney Street Dartmouth NS B2Y 1C5 July 21, 2021 (6m)
CLARKE, Shaun Richard First South, Lunenburg County April 26, 2021	Jessica Danielle Chapman 11 Richards Drive Dartmouth NS B3A 2P1	Lauren M. Randall BOYNECLARKE LLP 600-99 Wyse Road Dartmouth NS B3A 4S5 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
CRAIG, Frank J. Truro, Colchester County July 14, 2021	M. Beverley Craig 488 Cobequid Road Lower Sackville NS B4C 4E8	J. Ronald Creighton, QC Patterson Law 10 Church Street Truro NS B2N 3Z6 July 21, 2021 (6m)
CRAWFORD, Ronald Sydney Mines, Cape Breton Regional Municipality June 16, 2021	Kimberley Crawford 49 Rosemount Drive Bible Hill NS B2N 6J2 Rhonda Crawford 16-319 Young Street Truro NS B2N 3W4	Bradford G. Yuill The Yuill Law Firm 541 Prince Street Truro NS B2N 1E8 July 21, 2021 (6m)
CURRIE, Gerald Roy Dartmouth, Halifax Regional Municipality July 8, 2021	Gerald Roy Currie 1968 Lawrencetown Road Lawrencetown NS B2Z 1C5 Lynn Michele Currie 171 Broom Road Westphal NS B2W 6J6	W. Brian Smith, QC 604-45 Alderney Drive Dartmouth NS B2Y 2N6 July 21, 2021 (6m)
DAVIS, Ralph Manning Halifax, Halifax Regional Municipality June 7, 2021	Kelvin Peter Davis 34 Oakvale Court Timberlea NS B3T 1K7	Bradford G. Yuill The Yuill Law Firm 541 Prince Street Truro NS B2N 1E8 July 21, 2021 (6m)
DAWSON, John Walter Lower Sackville, Halifax Regional Municipality July 12, 2021	Mark Gerald Dawson 58 Sunnyvale Court Lower Sackville NS B4E 2T1 Duane Thomas Dawson 92 Sami Drive Lower Sackville NS B4C 3S5	Carrie E. Rice 5-400 Sackville Drive Lower Sackville NS B4C 2R8 July 21, 2021 (6m)
DOWNEY, Bruce B. North Preston, Halifax Regional Municipality July 2, 2021	Anastasia Milton 36 Maurice Lane North Preston NS B2Z 1A7 Cory Brown 10 Williams Street East Preston NS B2Z 1H4	Blair MacKinnon Heritage House Law Office 92 Ochterloney Street Dartmouth NS B2Y 1C5 July 21, 2021 (6m)
DROVER, Josiah Halifax, Halifax Regional Municipality June 16, 2021	Joseph Patrick Babin 405-3600 John Parr Drive Halifax NS B3K 5V3	Joseph A. MacDonell Carruthers MacDonell & Robson 5 Mill Village Road PO Box 280 Shubenacadie NS B0N 2H0 July 21, 2021 (6m)
FADER, Linda Pearl Halifax, Halifax Regional Municipality July 6, 2021	Allison Fader 22 Aster Court Halifax NS B3S 1G5 Laura Fader (aka Laura Evans- Fader) 59 Merchant Court	Eric Thomson Quackenbush Thomson Law 2571 Windsor Street Halifax NS B3K 5C4 July 21, 2021 (6m)

Halifax NS B3M 3E9

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
FERGUSON, George Evangeline Victoria, British Columbia July 9, 2021	David Delbert Ferguson 28 Brian Drive Westphal NS B2W 3N3	Jonathan Hooper Cox & Palmer 1100-1959 Upper Water Street Halifax NS B3J 3N2 July 21, 2021 (6m)
FRANCIS, Linda Margaret Lower Sackville, Halifax Regional Municipality July 2, 2021	Marleen H. Murray 113 Brookside Road Hatchet Lake NS B3T 1S3	Clyde A. Paul 103-531 Herring Cove Road Halifax NS B3R 1X3 July 21, 2021 (6m)
GARBER, Audrey Pauline Dartmouth, Halifax Regional Municipality July 5, 2021	Geoff Scott Garber 4 Evergreen Crescent West Porters Lake NS B3E 1L7	Blair MacKinnon Heritage House Law Office 92 Ochterloney Street Dartmouth NS B2Y 1C5 July 21, 2021 (6m)
HAVERSTOCK, Edith Mae Halifax, Halifax Regional Municipality June 28, 2021	Scott Elliott Haverstock 270 Oakfield Park Road Oakfield NS B2T 1V3 Steven Donald Haverstock 263 Boutiliers Cove Road Hackett's Cove NS B3Z 3J6	Page McInnis McInnes Cooper 1300-1969 Upper Water Street PO Box 730 Halifax NS B3J 2V1 July 21, 2021 (6m)
HOLMVALL, Vilma Karin Louise Marie (aka Marie W.C.L. Holmvall) Halifax, Halifax Regional Municipality July 8, 2021	Camilla Marita Holmvall 1808 Preston Street Halifax NS B3H 3V8	Ryan P. Brennan Patterson Law 2016 Quinn Street Halifax NS B3L 4E6 July 21, 2021 (6m)
HYNES, Raymond E. Halifax, Halifax Regional Municipality June 25, 2021	Gerald E. Hynes 146 Hillside Drive Boutilier's Point NS B3Z 1W8	Samuel R. Lamey, QC Wells, Lamey, Bryson, Schnare & Mailman 24 Pleasant Street PO Box 310 Chester NS B0J 1J0 July 21, 2021 (6m)
ISENOR, Morris Kendall Wallace, Cumberland County July 12, 2021	Jerome T. Langille 1007 Gulf Shore Road Pugwash NS B0K 1L0	Charles A. Ellis Hicks LeMoine Law PO Box 279 15 Princess Street Amherst NS B4H 3Z2 July 21, 2021 (6m)
JENNINGS, David Halifax, Halifax Regional Municipality November 30, 2020	Austin Jennings 6150B Chebucto Road Halifax NS B3L 1K5	Timothy J.C. Hall Landry, McGillivray 300-33 Ochterloney Street PO Box 1200 Dartmouth NS B2Y 4B8 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
JOHNSON, James Robert East Preston, Halifax Regional Municipality July 13, 2021	Robert "Robbie" Cain 48 Flying Cloud Drive Dartmouth NS B2W 4S9	Jessica L. Lyle, TEP Touchstone Legal Inc. 200-56 Portland Street Dartmouth NS B2Y 1H2 July 21, 2021 (6m)
JOHNSTON, H. Elizabeth Enfield, Hants County April 28, 2021	J.W. Stephen Johnston 63 Cochrane Point Road Enfield NS B2T 1H1	Joseph A. MacDonell Carruthers MacDonell & Robson 5 Mill Village Road PO Box 280 Shubenacadie NS BON 2H0 July 21, 2021 (6m)
KNIGHT, Dennis Patrick Sheet Harbour, Halifax Regional Municipality June 30, 2021	Dennis Purdy 8 Provost Street Chrysler ON K0A 1R0 Janice Leslie 60 Leslie's Hill Road Tangier NS B0J 3H0	Michael Brine 8339 Highway 7 Musquodoboit Harbour NS B0J 2L0 July 21, 2021 (6m)
LANCELEVE, Angus George West Lake Ainslie, Inverness County July 7, 2021	Lee Lanceleve 209 Charles Street Stewart Valley SK S0N 2P0	Wayne J. MacMillan, LLB EMM Law Incorporated 409 Granville Street Port Hawkesbury NS B9A 2M5 July 21, 2021 (6m)
LANDRY, Murray David Halifax, Halifax Regional Municipality June 30, 2021	Paul Gerrior PO Box 852 113 Old Foxbrook Road Stellarton NS B0K 1S0	July 21, 2021 (6m)
LANTZ, Shirley Lorraine Coldbrook, Kings County July 8, 2021	Darren Wayne Crowell 30 Craig Chandler Drive Pine Grove NS B4V 7Z9	Trinda L. Ernst, QC 469 Main Street PO Box 98 Kentville NS B4N 3V9 July 21, 2021 (6m)
LARKINS, John Kinnell Biggers Dartmouth, Halifax Regional Municipality July 2, 2021	Elisa Ann Bourdages 42 Newton Road St. John's NL A1C 4E1	Blair MacKinnon Heritage House Law Office 92 Ochterloney Street Dartmouth NS B2Y 1C5 July 21, 2021 (6m)
MacBRIDE, Wesley Gordon Canning, Kings County July 6, 2021	Janet Louise Devidal 1848 Sherman Street Coldbrook NS B4R 1A3 James Robert MacBride 309-33 Weldrick Road East Richmond Hill ON L4C 8W4	Trinda L. Ernst, QC 469 Main Street PO Box 98 Kentville NS B4N 3V9 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
MacDONALD, Charles Cyril Whycocomagh, Victoria County June 28, 2021	David MacDonald 109 Highland Street Ingonish NS B0C 1K0	Elliot K. Fraser Elliot K. Fraser Barrister & Solicitor Inc. 103-521 Chebucto Street PO Box 732 Baddeck NS B0E 1B0 July 21, 2021 (6m)
MacDONALD, Daniel Parnell Irish Vale, Cape Breton Regional Municipality March 16, 2016	Colleen (Donovan) Lemoine 1976 Morrison Road Sydney NS B1M 1A8	Demetri Kachafanas, QC Kachafanas Law Office 90 Beechwood Crescent Sydney NS B1S 1V8 July 21, 2021 (6m)
MacDONALD, John Murdock (aka John Jack Murdock MacDonald) Inlet Baddeck, Victoria County May 17, 2021	Kimberly Anne MacDonald 166 Yankee Line Nyanza NS B0E 1B0	Elliot K. Fraser Elliot K. Fraser Barrister & Solicitor Inc. 103-521 Chebucto Street PO Box 732 Baddeck NS B0E 1B0 July 21, 2021 (6m)
MacDONALD, Martina Marie Halifax, Halifax Regional Municipality June 7, 2021	Kevin C. MacDonald 89 Roxbury Crescent Halifax NS B3M 4S9	July 21, 2021 (6m)
MacKINNON, Lydia Ann Inverness, Inverness County July 8, 2021	Kenzie MacKinnon 8 St. Catherine Street St. Catherine's ON L2R 5E5	Wayne MacMillan, LLB EMM Law Incorporated 409 Granville Street Port Hawkesbury NS B9A 2M5 July 21, 2021 (6m)
MacLAUGHLIN, Rose Ann Delores (aka Ronnie Hayden) Northwood Care Halifax, Halifax Regional Municipality June 25, 2021	Public Trustee PO Box 685 Halifax NS B3J 2T3	Andrew P. Nicol Public Trustee PO Box 685 Halifax NS B3J 2T3 July 21, 2021 (6m)
MacLEOD, Patricia Irene Dartmouth, Halifax Regional Municipality July 12, 2021	Colin Gilbert MacLeod 302-235 Baker Drive Dartmouth NS B2W 0K4	George M. Clarke BOYNECLAKRE LLP 600-99 Wyse Road Dartmouth NS B3A 4S5 July 21, 2021 (6m)
MILLER, James William Camp Hill Veterans Memorial Building Halifax, Halifax Regional Municipality June 15, 2021	Shirley Neville (aka Shirley Jeanne Neville) 405-180 Basswood Run Dartmouth NS B2W 0L5	W. Brian Smith, QC 604-45 Alderney Drive Dartmouth NS B2Y 2N6 July 21, 2021 (6m)
MILLS-MARTIN, Evelyn Rose Lequille, Annapolis County July 7, 2021	Shirley Elizabeth Mills 50 Seaman Street Margaretsville NS B0S 1N0	John H. Armstrong Armstrong Law Office 240 St. George Street PO Box 575 Annapolis Royal NS B0S 1A0 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
MORRISON, David Francis Timberlea, Halifax Regional Municipality June 23, 2021	Douglas Paul Morrison 16 Nicholson Drive Lakeside NS B3T 1B3 Greta Ann Fillier 31 Oak Drive Truro NS B2N 5M3 Eric Lorne Morrison 2953 Winston Place Halifax NS B3L 3V5 Ethel Patricia Morrison 46 Eisener Street Timberlea NS B3T 1E3 Reginald James Morrison 98 Sugar Maple Drive Timberlea NS B3T 1G5 Lynne Marie Brown 51 Maplewood Drive Timberlea NS B3T 1G1 Albert Charles Morrison 44 Valleyview Drive Grand Bay-Westfield NB E5K 1X2 Cara Campbell 114 Mayflower Drive New Glasgow NS B2H 5S2 Ashley Morrison 21 Isaac Jennings Drive Hubbards NS B0J 1T0	Gary R. Armsworthy Armsworthy Lynch 215-6021 Young Street Halifax NS B3K 2A1 July 21, 2021 (6m)
MOSEY, Irene Marie Dartmouth, Halifax Regional Municipality July 12, 2021	Michelle Marie Mosey 509 Main Street Elsipogtog NB E4W 2X5	Micaela A. Sheppard Weldon McInnis 118 Ochterloney Street Dartmouth NS B2Y 1C7 July 21, 2021 (6m)
MULLINS, Benjamin T. (aka Thomas Benjamin Mullins) Main-a-Dieu, Cape Breton Regional Municipality June 16, 2021	Eileen Mullins (aka Eileen O. Mullins) 2435 Main-a-Dieu Road Main-a-Dieu NS B1C 1X2	Murray F. Hannem Sampson McPhee 200-66 Wentworth Street PO Box 1120 Sydney NS B1P 6T4 July 21, 2021 (6m)
PAYZANT, John William Bedford, Halifax Regional Municipality June 7, 2021	Andrew Payzant 1466 Bishopville Road Hantsport NS B0P 1P0 Jeffrey Payzant 273 Stonemount Drive Sackville NS B4C 4G9	Timothy C. Matthews, QC Stewart McKelvey PO Box 997 Halifax NS B3J 2X2 July 21, 2021 (6m)
PETERS, Leo Malcolm Westphal, Halifax Regional Municipality June 9, 2021	Kelly Joseph Peters 440 Caldwell Road Dartmouth NS B2V 1A6	Douglas J. Livingstone 201-219 Waverley Road Dartmouth NS B2X 2C3 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
PETTIPAS, Henrietta Howard Dean Plymouth, Pictou County July 13, 2021	Danny Fisher 87 Riverside Avenue Stewiacke NS B0N 2J0	Joseph A. MacDonell Carruthers MacDonell & Robson 5 Mill Village Road PO Box 280 Shubenacadie NS B0N 2H0 July 21, 2021 (6m)
PHILLIPS, Laurence Dartmouth, Halifax Regional Municipality July 13, 2021	Cheryl Howells 10 Plaza Drive Dartmouth NS B2W 3H6	Lisanne M. Jacklin Heritage House Law Office 92 Ochterloney Street Dartmouth NS B2Y 1C5 July 21, 2021 (6m)
POOLE, Wilson James (aka James Wilson Poole) Middle Musquodoboit, Halifax Regional Municipality July 2, 2021	Christopher Poole 41 Tower View Court Lantz NS B2S 1N7	Stephen J. Topshee Burchell MacDougall LLP 710 Prince Street PO Box 1128 Truro NS B2N 5H1 July 21, 2021 (6m)
ROGERS, Joseph Roy Sydney, Cape Breton Regional Municipality May 17, 2021	Katelynn Carter-Rogers 36 Beaver Crescent Dartmouth NS B2V 1C9 Tracey Carter Rogers 120-556 Bedford Highway Halifax NS B3M 4X7	July 21, 2021 (6m)
ROPER, Phoebe Winona Halifax, Halifax Regional Municipality June 11, 2021	Henry Roper c/o J. Walter Thompson, QC Quackenbush Thomson Law 2571 Windsor Street Halifax NS B3K 5C4	J. Walter Thompson, QC Quackenbush Thomson Law 2571 Windsor Street Halifax NS B3K 5C4 July 21, 2021 (6m)
ROSS, Gerald Richard Riverton, Pictou County February 22, 2021	William Gerald Ross PO Box 383 1 Cavell Street Trenton NS B0K 1X0	J. Gregory MacDonald, QC Patterson Law 47 Riverside Street PO Box 697 New Glasgow NS B2H 5G2 July 21, 2021 (6m)
ROSSETTI, Philip Ronald (aka Phillip Rossetti; aka Philip Rossetti) Lahr, Germany June 9, 2021	Hannelore Emma Rossetti 13 Leeds Street Westmount NS B1R 1G8	Kelly O'Brien Sampson McPhee PO Box 1120 200-66 Wentworth Street Sydney NS B1P 6T4 July 21, 2021 (6m)
SHORT, Gloria Alvina Truro, Colchester County July 15, 2021	William Bruce Short and Edmund John Short c/o 66 Osprey Court Bible Hill NS B2N 7B3	Cassandra Armsworthy Patterson Law 10 Church Street Truro NS B2N 3Z6 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
SMITH, Robert Dunlop Dartmouth, Halifax Regional Municipality June 3, 2021	Robert Dunlop Smith 15 Ridgewood Court Dartmouth NS B2W 6E5 Alexander Smith 14 Samuel Daniel Drive Eastern Passage NS B3G 1P7	Amanda Brousseau 203-94 Garland Avenue Dartmouth NS B3B 0A7 July 21, 2021 (6m)
SMITH, Stephen Leroy Cedar Lake, Yarmouth County June 23, 2021	Gary Leland Smith 227 Cedar Lake Road Cedar Lake NS B5A 7R5	Andrew S. Nickerson, QC Nickerson Jacquard Russell 77 Water Street PO Box 117 Yarmouth NS B5A 4B1 July 21, 2021 (6m)
SOUTHCOTT, Ronald Joseph (aka Ronald J. Southcott) Bedford, Halifax Regional Municipality June 7, 2021	Rita Marie Southcott (aka Rita M. Southcott) 120 Southgate Drive Bedford NS B4A 0B1	July 21, 2021 (6m)
SPIELDIENER, Gertrude Ida (aka Gertrude Ida Spieldiener-Fre) Schaan, Liechtenstein July 14, 2021	Brigitte Fash-Spieldiener 3231 Antigua Drive Punta Gorda FL 33950 USA	Maurice Boudreau, JD EMM Law Incorporated 409 Granville Street Port Hawkesbury NS B9A 2M5 July 21, 2021 (6m)
STEVENS, Cecil George LaBelle, Queens County March 31, 2021	Mary Jean Hilchey 15 Hampton Court Fall River NS B2T 1E7	Timothy J.C. Hall Landry, McGillivray 300-33 Ochterloney Street PO Box 1200 Dartmouth NS B2Y 4B8 July 21, 2021 (6m)
STUIFBERGEN, Marcia Dartmouth, Halifax Regional Municipality July 5, 2021	Public Trustee PO Box 685 Halifax NS B3J 2T3	Adrienne Bowers Public Trustee PO Box 685 Halifax NS B3J 2T3 July 21, 2021 (6m)
TARBOX, Wade Harry Gold River, Lunenburg County July 3, 2021	Christopher Wade Tarbox 5438 Highway 3 Chester Basin NS B0J 1K0	Patrick A. Burke, QC Burke, MacDonald & Luczak 28 King Street PO Box 549 Lunenburg NS B0J 2C0 July 21, 2021 (6m)
TITUS, Audrey Winnifred (aka Audrey W. Titus) Centreville, Digby County July 8, 2021	Sherry Longmire 59 Racquette Road PO Box 2232 Digby NS B0V 1A0	Oliver Janson Oliver Janson Law Inc. 93 Montague Row PO Box 129 Digby NS B0V 1A0 July 21, 2021 (6m)

ESTATE OF: Place of Residence of Deceased Date of Grant	Personal Representative(s)	Solicitor for Personal Representative Date of the First Insertion
TRIDER, Donald Murray Albert Dartmouth, Halifax Regional Municipality July 5, 2021	The Bank of Nova Scotia Trust Company 1709 Hollis Street, 5 th Floor Halifax NS B3J 2P8	Kent W. Rodgers Casey Rodgers Chisholm Penny Duggan LLP 201-219 Waverley Road Dartmouth NS B2X 2C3 July 21, 2021 (6m)
TRITES, Shirley Joan Halifax, Halifax Regional Municipality July 8, 2021	J. Fredrick Morley 2221 Newton Avenue Halifax NS B3L 3C1	Devon E. Cassidy Cox & Palmer 1100-1959 Upper Water Street Halifax NS B3J 3N2 July 21, 2021 (6m)
WAYE, Cindy Lee Halifax, Halifax Regional Municipality June 24, 2021	Allan Marshall Waye 1730 Dunvegan Drive Halifax NS B3H 4G1	July 21, 2021 (6m)
WHITE, Merlin Kennetcook, Hants County February 18, 2021	Robbie Burns 4829 Highway 215 Noel NS B0N 2C0	Timothy J.C. Hall Landry, McGillivray 300-33 Ochterloney Street PO Box 1200 Dartmouth NS B2Y 4B8 July 21, 2021 (6m)
WILLIAMS, Curtis Leigh Lockeport, Shelburne County June 25, 2021	Pamela Anne Williams 16 Paradise Street PO Box 531 Lockeport NS B0T 1L0	Donald G. Harding Donald G. Harding QC Barrister and Solicitor Inc. 3407 Highway 3 PO Box 580 Barrington Passage NS B0W 1G0 July 21, 2021 (6m)
WILSON, Elizabeth Ellen Halifax, Halifax Regional Municipality April 7, 2021	Brenda Smith 35 Kent Avenue Halifax NS B3M 2G2	Timothy J.C. Hall Landry, McGillivray 300-33 Ochterloney Street PO Box 1200 Dartmouth NS B2Y 4B8 July 21, 2021 (6m)
YOUNGBLUTH, Edith Agnes Gaetz Brook, Halifax Regional Municipality June 30, 2021	Carl Scott Youngbluth 58 Tower Road Head of Chezzetcook NS B0J 2L0	Doug R. Stevenson Quackenbush Thomson Law 2571 Windsor Street Halifax NS B3K 5C4 July 21, 2021 (6m)

ESTATE NOTICES BEING PUBLISHED FOR SECOND OR SUBSEQUENT TIME NOTE: Name of Personal Representative and Solicitor for the estate appears at the time of first insertion.

Index of Estate Notices currently being published for the required six-month period under subsection 63(1) of the *Probate Act*. The gazette issues shown below can be found at www.novascotia.ca/just/regulations/rg1/.

Estate Name	Date of First Insertion
ABRAHAM, Alan Rockwell	February 17, 2021
ACKERLEY, Georgena Doris	February 10, 2021
ADAMS, John (aka John Quincy Adams)	

ADAMS, Matthew Gary	
AITKEN, Annie Marie	
ALEXANDER, Blanche Mae	June 9, 2021
ALEXANDER, Eda	
ALEXANDER, James Armand	April 21, 2021
ALLAN, Jacqueline Amelia	June 9, 2021
ALLEN, Harold Spencer	
ALLEN, Judith Dian	
ALLEN, Ruth Frances (aka Ruth F. Allen)	
ALLEN, Shirley Ann	
ALLEN, Willard Robert.	
ALLISON, George S. (aka George Stephen Allison)	
ALLSOP, Doris Pauline	
AMERO, Joseph Alfred	
AMIRAULT, Anne Marie	•
AMIRO, Alban Anthony	
AMIT, Harris Royston	
ANDERSON, Linda Patricia	
ANDERSON, Robert Raymond James	
ANDERSSON, Lars Eric	
ANDREWS, Meredith Mabel	March 31, 2021
ANDREWS, Myrtle Helen June	June 16, 2021
ANDREWS, Sharon Diane	July 14, 2021
ANNAND, Betty Gordon	May 26, 2021
ANTHONY, Kenneth Alonzo	
ANTHONY, Myrtle	
APPLETON, Eileen Enis	
ARCHIBALD, Carl Spencer	
ARKLIE, Margaret Muir	
ARMSTRONG, Ronald David	
ARNBURG, Lloyd George	
ARNOLD, Alden Raymond	
ARNOLD, Michael Edward Maynard	
ARSENAULT, Wilfred Joseph	
ARTHURS, Douglas Walter	
AUCOIN, Edward Joseph	
AUCOIN, Marie Louise	
AUDAIN, Frances Janet (aka Janet Audain)	
AULENBACK, Robert St. Clair	February 17, 2021
AVERY, Evangeline Mary	May 5, 2021
BAIRD, John Russell	
BAIRD, Mary Belle	June 9, 2021
BAIRD, Teresa Reynelda	
BAISLY, Paul H.	
BALCOM, Bert Ingram	
BALDWIN, John Christopher George (aka Christopher Baldwin)	
BALTERMAN, Alisa M. (aka Alisa Marie Balterman)	
BANDURAK, Harry	
BARKHOUSE, Carol Ann	
BARKHOUSE, Lillis Cavell	
BARRETT, Florrie Margaret	
BARRETT, William John	
BARTLETT, Barbara Jean	
BASILE, Rita	
BASKER, Eugene Gerard	March 24, 2021
BEALS, Louis	March 17, 2021
BEAVER, Valerie Mae	
BELONG, Geraldine Kathleen	
BENNETT, Robert Bruce	
BENOIT, Imelda (aka Imelda Catherine Benoit)	
. ,	, -,

BENSON, Colleen Tresa (aka Colleen Tresa Benson-Carroll)	
BENT, Carol Marie	March 10, 2021
BENT, Dorothy May	February 24, 2021
BENT, Ralph Walter	July 14, 2021
BEREZUK, Cynthia	March 31, 2021
BERNARD, Rita Lucille	May 26, 2021
BEST, James William	
BEZANSON, Douglas Morrison	February 24, 2021
BINGLEY, Ronald David	
BISSON, Daphne Joan	
BLACK, John Gerard (aka Gerard Black; aka Gerard John Black)	March 31, 2021
BLACK, Shaun Campbell	
BLACKMAN, Thelma Leone	
BLAIN, Daniel, Jr.	March 31, 2021
BLAKENEY, Barry Arthur	
BOGGS, George David	
BOLGER, John Patrick	
BOLHUIS, Aafke	
BOLIVAR, Audrey Marie	
BOLLIVAR, Olive Victoria	
BONANG, Frederick Joseph	
BONNYMAN, Alice C.	
BOOMGAARS, Leonardus (Leo) Antonius	
BOONE, Margaret Matilda (aka Margaret Boone)	
BOUDREAU, Clarence Joseph	
BOUDREAU, Joseph Melvin	
BOUDREAU, Kenneth Gerard	
BOUDREAU, Kyle William	
BOURGEOIS, Joseph Daniel	
BOUTILIER, Anthony William	
BOUTILIER, Granville Leroy	
BOUTILIER, Lester Elroy	
BOUTILIER, Margaret	
BOUTILIER, Terry William	
BRADSHAW, Daniel Augustine	
BRAGG, Walter (aka Walter Freeman Bragg)	
BRAUNERT, Edith May	
BRIGGS, John Carmel (aka Carmel Briggs)	June 16, 2021
BRIGGS, Marceline Marie	June 16, 2021
BRITTAIN, Shirley Sherman	
BROWN, Alice Mildred	February 10, 2021
BROWN, Allison Webster	July 7, 2021
BROWN, Eleanor Christena	March 10, 2021
BROWN, Flora Minnie	
BROWN, Mary Irene (aka M. Irene Brown)	March 24, 2021
BROWN, Robert William	March 24, 2021
BROWN, Shirley William (Bill)	
BRUNELLE, Mabel Allan	
BRUNO, Biagio (aka Bruno Biagio)	
BRUSHETT, Ethel Beatrice	
BUCHAN, Robert MacPherson	
BUCHANAN, Catherine Bernardita	
BUGLAR, Alexander Isaac (aka Isaac Alexander Buglar)	
BULLERWELL, Robert Everett (aka Robert Bullerwell; aka Robert "Bob" Everett Bullerwell)	
BUNIN, Eric (aka Eric Paul Bunin; aka Eric P. Bunin)	
BUOTT, Carol Ruth	
BURDOCK, Barbara Marion	March 3, 2021
BURGESS, Helen Virginia	
BURKE, Harriet A.	
BURKE John James (Jackie)	June 16, 2021

BURKE, Larry William	March 17, 2021
BURRELL, Shirley Ann [corrected June 30, 2021]	June 9, 2021
BURTON, Jason Andrew	
BURTON, Melvin Alexander	
BUSHELL, Norris Leonard	January 27, 2021
BUTLER, Robert James	March 24, 2021
BYRNE, Helen Sylvia	
BYRNE, Robert Kim	
CAHILL, Thomas Joseph	
CAMERON, Clifford Murray	
CAMERON, Donald William	
CAMERON, Susan Avis	
CAMERON, William (aka William James Cameron)	
CAMERON, William Archibald	
CAMPBELL, Agnes Hughena	
CAMPBELL, Kevin Lionel	
CAMPBELL, Lorne Allister	
CAMPBELL, Robert Hugh	
CAMPBELL, Robin Bruce	
CAMPBELL, Susan Marie (aka Susan Marie Penney)	
CANNING, Jocelyn Yvonne	
CAREY, Philip Ross	
CARMAN, Raymond Thompson	
CARNES, Denise Bernadette	
CARSCADDEN, Joyce May	
CARTER, Andrew McCart	
CARTER, Barbara Joan	
CASHIN, Walthen	
CASTLE, Claire Corinne	* '
CENTA, Albert (aka Albert Joseph Centa)	
CHAMBERS, Neil	
CHARLTON, Arthur Dean	
CHARLTON, Ronald James	June 9, 2021
CHATELOIS, Roger Irvin (aka Irvin Roger Chatelois)	July 7, 2021
CHEHIL, Dalip S.	April 14, 2021
CHIASSON, Diane Marie	June 9, 2021
CHIASSON, Joseph Alfred	February 17, 2021
CHISHOLM, Thomas Joseph	June 23, 2021
CHRISTIE, Shirley Irene	March 31, 2021
CHRISTOPOULOS, Haris Constantine (aka Haris C. Hristopoulos)	
CHRONIS, Calliope	
CHURCHILL, Maureen Elizabeth	
CINNAMON, Lisa Sarah	
CLARK, Robert Trent	
CLARK, Shirley Bernice	
CLARKE, Leona Mary	
CLARKE, Marion Winnifred	
CLARKE, Nancy Marguerite	
CLEARY, Mary	
CLEM, Lorimer Austin	
CLEVELAND, Roselyn Cecile (aka Roselyn Cleveland)	
COADY, Flora Christena	
COADY, Judith Marie	
COGSWELL, Aaron Richard Lloyd	
COHEN, Harold	
COLEMAN, Gail Patrick	•
COLFORD, Cyril A.	
COMFALL Devold Levis (clas Devold Levis Company)	
COMEAU, Donald Louis (aka Donald L. Comeau; aka Donald Comeau)	
у у ругожет гроше така гроше и у отнежи: ака глоше тоsenn и отнежи	rencuary to 7071

COMEAU, Edward Joseph	
COMEAU, Irene Marie (aka Irene Comeau; aka Marie Irene Cecile Comeau)	April 28, 2021
COMER, Kathleen Agnes	May 5, 2021
CONRAD, Kathleen Elizabeth	March 3, 2021
CONRAD, Linda	The state of the s
CONRAD, Sylvia Eileen	
CONRAD, William Percy	
CONROD, Lloyd Merlin	
COOKE, Gerald Thomas	
COOLEN, Christine Helen	
COOLEN, Clark Leon	
COOLEN, Joan Marie	
COOPER, Stella Catherine (aka Stella Cooper)	
CORBIN, Elaine Janette	
COREY, John D.	
CORKUM, Harland Earle	
CORMIER, Leo Wilfred	
CORRA, Roberto Primo.	
COSTANDI, Monica	
COTTER, Erland Wilbert	April 21, 2021
COUSINS, Carol Ann	March 17, 2021
COUSINS, Dean Richard	May 5, 2021
COWBROUGH, Abbigail	
COX, Daniel William	
COYLE, Winifred Helen	
CRAFT, David Malcolm	
CREASER, Allan Forman	
CRITCHLEY, William Claude	
CROFT, Jacob Donald	
·	The state of the s
CROFT, Nellie May	
CROMWELL, Audrey Evangeline	
CROOK, Carrie Edna	
CROSBY, Delbert St. Clair	
CROSLAND, Paul Hydes	
CROWELL, Bryson Elliott	
CUNNINGHAM, Michael Ranald	The state of the s
CUNNINGHAM, Stephan Edwin	
CURRIE, Royden Duane	February 17, 2021
CURRIE, Stephen R.	May 19, 2021
CURRY, Leo J. (aka Leo Jerome Curry)	
CURRY, Patricia Ellen	
CUVELIER, Louis Barnard	
d'ENTREMONT, Inabelle M. (aka Inabelle Marguerite d'Entremont)	
D'ENTREMONT, Marjorie Thérèse	
D'EON, Felicita Thérèse	
DALY Musical S	
DALY, Muriel S.	
DANIELS, Lee Ross	
DANIELS, Leota Maie	
DARRAGH, Sheila	
DAUPHINEE, Harry Maurice	
DAUPHINEE, Lillian Frances	
DAVID, Kenneth Nelson	
DAVIDSON, Andrea Nancy	July 7, 2021
DAVIES, Percival James	
DAVIES, Susan Elizabeth	
DAVIS, Katherine Mary	<u> </u>
DAVIS, Matthew Christopher	
DAVIS, Ralph Manning	
DAVISON Paul Alfred	

DAWE, Eileen Marjorie	
DAWSON, James	
DAY, Charles Maxwell	
DEALE, Sheryl Anne	June 16, 2021
DEAN, Clyde Otto	February 17, 2021
DEAN, Johanna Helene Rosalie (aka Johanna Dean)	
DEAN, William Scott	April 21, 2021
DeCHAMP, Roger Howard (aka Roger H. W. DeChamp)	May 5, 2021
DeCOSTE, Carl Allister	January 27, 2021
DELANEY, Phyllis Marie	
DEMPSEY, Daniel Anthony Gerard	
DENIKE, Dorine Patricia	
DENNIS, Ruby Lucille	
DENSMORE, Garry Emmerson	
DEVAU, Charles Roy [cancelled, republished May 5, 2021 (Deveau, Charles Roy)]	
DEVEAU, Antoine Victor	
DEVEAU, Antoine Victor	
DEVEAU, Earl David	
DEVEAU, Paul V. (aka Paul Victor Deveau)	March 31, 2021
DeVOOGT, Klazina	
DEXTER, Annette Louise (aka Annette L. Rijnbeek; aka Annette L. Ritchie)	February 3, 2021
DeYOUNG, James Joseph	June 9, 2021
DEYOUNG, Rachel Marie	
DICKIE, Garnet Ross	
DICKIE, Robert Thomas	
DiLIBERATORE, Egea	•
DINGLEY, Margaret Kathleen	
DOANE, Karl Douglas	
DOIG, Peter Ramsay	
DOIRON, Rodney Alphonse	
DONAT, Peter C. [corrected April 21, 2021]	March 17, 2021
DORT, Ruby Ella Lillian (aka Ruby Lillian Dort; aka Ruby Dort)	July 14, 2021
DOUCET, Louise (aka Louise Marie Doucet)	
DOUCETTE, Emelie Claudette (aka Claudette Doucette)	April 14, 2021
DOUCETTE, Lorraine Joyce	March 10, 2021
DOUCETTE, Roy Thoms	
DRAPEAU, Harriet Elizabeth (aka Harriet Elizabeth Mann)	April 14, 2021
DRELLER, Josef	June 16, 2021
DREW, William Henry	March 17, 2021
DUFFY, Gloria Elizabeth	June 23, 2021
DUGAS, Francis (aka Joseph Francis Dugas)	
DUGGAN, Mary Kathleen	
DUNN, Terry (aka Terry Douglas Dunn)	
DURANT, Natalie W. (aka Natalie Winnifred Durant)	
EAGER, David (aka David Roy Eager)	
EARLEY, Shirley Edith Baltzer	
EDDINGTON, David Harcus	
EDDY, Maurice Sinclair	
EDGAR, Winnifred (aka Winnie Edgar)	
EDWARDS, Amelia Jane	
EGAN, Melvin (aka Melvin Wilfred Egan)	
EISENHAUER, Alice Kathryn	
ELDERKIN, Arthur Royce	
ELLIOTT, Lawrence Kevin	May 12, 2021
ELLIS, Joan Gwendolyn	May 19, 2021
ENGLAND, Margaret Eva	
EPIFANO, Edith Vera	
ERNEST, Faye Lorena	
ERSKINE, Brenda Mary Therese	
	, - -

ERVINE, Warren Basil	January 27, 2021
ETTINGER, Doris Adeline	May 19, 2021
ETTINGER, Jay Warren	April 28, 2021
EVANS, Herbert Charles	
EVANS, Mary Alice	
EVANS, Patricia Ann	
EVANS, Paul Leo	
FADER, Wayne Tennyson MacDougall	
FAHIE, Norma Laverne	
FARLEY, Letha	
FARMER, Irma Diana	
FARR, Jessie	
FARROW, Harold Joseph	
FAULKNER, Thomas G.	
FAWSON, Florence Agnes.	•
FEENER, Ronald Lemont	
FELMMING, Michael Dennis	•
FELTMATE, Emma Elseba	
FERGUSON, Donald John	
FERGUSON, Effie (aka Effie Jane Ferguson)	
FERGUSON, Margaret Ann	
FERGUSON, Ruth Elizabeth	
FETTERLY, Carmita Bernice	
FEVENS, Ralph William	
FILLMORE, Marion Eileen	
FINDLAY, Ann Ethelwyn Marion	
FINES, Herberta Ruth	,
FINLAY, Josephine Della (aka Josephine Bella Finlay)	
FIORE, Karen Lynn	
FITCH, Paul R.	
FITZMORRIS, Gail Marie	
FLEMING, Mary Colette	
FLYNN, Alice Mae	
FLYNN, Marie Corinne (aka Corinne Marie Flynn; aka Corinne Flynn)	
FLYNN, Wade Leslie	
FOGARTY, Andrew Joseph	March 31, 2021
FOLEY, Ruth Delores	March 3, 2021
FOLKINS, Ilze	June 23, 2021
FONTAINE, Joseph Marcel Richard Real	April 21, 2021
FORBES, Ralph Phillip	July 14, 2021
FORD, John Willard (aka John W. Ford)	
FOSTER, Bertha Karen	June 16, 2021
FOSTER, Eva Pauline	
FOSTER, Ruth Charlene	
FOUGERE, Agnes Gertrude	
FOUGERE, Doris Artimese (aka Doris Fougere)	
FOX, Shirley Marguerite	
FRAELIC, Mary Lou	
FRAME, Edward	
FRANCIS, Mary Lilian	
FRANKLIN, Eva Marilyn	
FRASER, Alexander Kenneth (aka Kenneth Fraser, Jr.)	
FRASER, Joan Edith (aka Joan E. Fraser)	
FRASER, Margaret Dolena	
FRASER, Richard Bruce	
FRASER, Verna Evelyn	
FREDERICKS, David Edgar Albert FREDERICKS, Ruth Alice	
FREEMAN, Roger Ralph	
FRISON Joan Marie	2021 April 28
TEN DAMAIN AUGULIVIALIE	ADD (0 /U/)

FROSST, Sigrid Estelle	April 7, 2021
FROTTEN, Barbara A.	
FULLER, Wilma Irene	March 3, 2021
FURLONG, Jacqueline Louise	April 21, 2021
GAETZ, Ross Alexander	
GALE, Edna Agnes	
GALE, Frank Harrison	
GALLAWAY, Thomas Joseph	
GARCIA, Maria Del Carmen (aka Maricarmen Garcia)	
GARDIN, Giacomo James	
GARDINER, Catherine	
GARDNER, Thomas Richard	
GARDNEROSE, Alice Patricia	
GARLAND, Malcolm Frederick	1 ,
GARRON, Bernard Morton	
GAUDET, Jeffrey Paul	
GAUTHIER, Dorothy Ann (aka Dorothy Gauthier)	
GENDRON, Donald Roger	
GENNOE, William Joseph (aka Joseph Gennoe; aka William J. Gennoe)	April 7, 2021
GEORGAKAKOS, Maria	
GEORGE, April Dawn	
GEORGE, Marjorie Dyanna	
GERMAINE, Earl Chester	
GHOSE, Dr. Tarunendu	
GILBERT, David Hugh	
GILES, Bernice Margery	
GILLIS, Donald Angus	
GILLIS, Norman Joseph	
GLASS, Elizabeth Agnes	
GOEL, Suraj	
GOLUCH, Chester [corrected June 16, 2021 (originally printed as GOLUGH, Chester)]	
GOODWIN, Florence Shirley	
GOSBEE, John James	
GOUBAULT, Eileen Marie	
GOUCHER, Chester William (aka Chester Goucher)	
GOUCHER, William Robert	
GOUCHIE, Kenneth Allan	
GOUDEY, Dorothy E.	
GOUGH, Norma Gwendolyn	
GOURLEY, Margaret Ellen	
GOURLEY, Mary Maxine	
GOUTHRO, John Gerard (aka Gerard John Gouthro)	
GRACIE, John Joseph	
GRAHAM, Kenneth Joseph	
GRAHAM, Perry Wade	
GRANDY, Bette (aka Betty Grandy)	
GRANT, David Foster	
GRAY, Doris Betty	
GRAY, Nora (aka Nora Mary Gray)	
GREENCORN, Elizabeth Margaret	
GREENLAW, Robert King	
GREENO, Emma Patricia	•
GREENOUGH, Irene C.	
GRIFFITHS, Elizabeth Safford	
GUIDRY, Stephen Cullman	
GUMMER, Gladys Irene (Daniels)	
GUY, Helen Leona	
HAGAN, Andrew D.	
HAINES, Carol Marie	
HAINES Clinton Curtis	July 14 2021

HAINES, Lloyd William	March 3, 2021
HAINES, Rita Shirley	March 3, 2021
HALL, James Stanley	February 3, 2021
HALLIDAY, Inez Mildred	
HAMILTON, William Ray	April 14, 2021
HANELY, Hugh Francis [cancelled, republished May 19, 2021 (Hanley, Hugh Francis)]	May 5, 2021
HANKINSON, Kenneth M. (aka Kenneth Munroe Hankinson)	
HANLEY, Hugh Francis	
HANLEY, Margaret Joan	
HARDING, Eleanor Noreen	
HARRIETHA, Elizabeth Hilda	
HARRISON, Alice Muriel	
HARRISON, Phillip (aka Phillip Allan Harrison)	
HART, James Leonard	
HARTLIN, Bruce Joseph	
HARTLING, Alice Rose Marie	
HARTT, Justin Avery	* .
HARVIE, Donald Garth	
HASHEM, Adele Mary	•
HASHEM, Vera Christina (aka Vera Kristina Hashem)	
HATT, Andrew John	
HATT, Frances Louise	
HATT, Robert MacLean	,
HAUGHN, Mary Anne	
HEBB, Cecil St. Clair	
HEFFLER, Joyce Olive	
HEFFLER, Una Frances	
HEIGHTON, Mary Alice	
HEIGHTON, Roderick Wayne	
HEINEKAMP, Bernard C.	
HEMEON, Freda	
HENDSBEE, Margarita "Alice"	
HENNESSEY, Margaret A. (aka Margaret Anne Hennessey)	
HENNESSY, Devan Jennifer Dawn	
HENNIGAR, Thelma Lee	
HENRIKSEN, Doris Mary	
HERBERT, Ruth Margaret	
HERSEY, Ernest Vincent	June 9, 2021
HEWITT, Lilla Marilyn	January 27, 2021
HILL, Muriel Grace (Goodwin)	
HILL, Olga Marie	
HILLIER, Frederick William	April 21, 2021
HILTZ, Margaret Joy	March 3, 2021
HINES, Allene Mary	
HINES, Joseph Angus	February 3, 2021
HOARE, David Oliver	June 9, 2021
HOARE, Leona Gertrude	July 14, 2021
HOBSON, Gordon Leroy	February 17, 2021
HODGES, Elizabeth Hennigar	June 16, 2021
HOLLETT, Ethel Violet	July 14, 2021
HOLLETT, Ruth Marceline	June 23, 2021
HOLT, Godlie Margaret (aka Goldie M. Holt; aka Goldie Holt)	
HOLTON, Margaret	-
HOOD, Gerald S.	June 2, 2021
HOOK, Zelina	
HOPE, Anthony (Tony) Charles	
HOPKINS, Donald George	
HOPKINS, Norma Jean	
HORNE, Martin James	
HORNE Murray Jerome	March 3 2021

HOUSTON, Heather Elizabeth	February 17, 2021
HOWATT, Roger Melborne	March 24, 2021
HOYLE-MOOD, Phyllis Hope (aka Phyllis Hoyle Mood; aka Phyllis Hoyle)	June 23, 2021
HUBLEY, Hattie Gertrude	
HUDSON, Gregory William Percy	
HUME, Sharon Jeanette [corrected July 14, 2021]	
HUMPHREYS, Rondell Putnam	
HUNTLEY, Violet Patricia	
HUSSEY, Robert Francis.	
HYLAND, Michael Shawn	
HYLAND, Vienna Lorraine Marie	
IZZARD, Mary Alice Jordan	
JACKSON, Verna Edith	
JAMES, Donald Graham	
JAMES, Theresa Lorraine	
JEDDRY, Roseann Nellie (aka Nellie Jeddry; aka Nellie Geddry)	
JEFFERSON, Joyce Elizabeth	
JEFFERSON, Thomas Bertram	
JENKIN, Samuel Burgess	
JENNEX, Otis Fortescue	
JENNINGS, Thomas Calvin, Jr	
JESS, Kathleen Margaret	
JESSOME, Eileen Elizabeth (aka Ruth Jessome)	May 19, 2021
JESSOP, Roy Douglas	April 21, 2021
JEWERS, Barry Leigh (aka Barry L. Jewers)	April 14, 2021
JEWERS, Doris Emily	April 28, 2021
JEWERS, Wallace Charles Owen	
JOHNSON, Arthur Keith (aka Arthur "Art" Keith Johnson)	
JOHNSON, Joy Marlene	
JOHNSON, Sarah Jane "Sally"	March 31, 2021
JOHNSTON, Alma Felicia (aka Alma Felicia Johnston-Tynes)	
JOHNSTON, Betty Lou (aka Betty-Lou Mae Johnston)	
JOLLIMORE, James Miles	
JOLLYMORE, Josephine Mary	
JONES, Brenda May Lillian	
JOSEY, Ruby Rhodena	
JOUDREY, Barry Scott	
JULIEN, Adrien Lawrence	,
·	
JURGENS, Mary	
KALAI, Nader Mohammad (aka Nader Kalai)	
KARREL, Samuel	•
KASSEM, Wafi A. R.	
KAZAMEL, Megan Patricia	
KEDDY, Bruce Milton	
KEDDY, Dianne Elizabeth	
KEEFE, Alice Gertrude	
KEIRSTEAD, Jean Pearl	
KEIZER, James Douglas	July 7, 2021
KELLY, Bonnie Sue (aka Bonnie Sue Hutchinson)	March 24, 2021
KELLY, Wilhemina Frances (aka Frances Wilhemina Kelly)	January 27, 2021
KENNEDY, Elizabeth Edna	February 17, 2021
KENNEY, Jennie Doris	May 5, 2021
KENNIE, June Lorraine	
KEYES, Velma	
KEYSTON, John Robert	
KING, Ernest Henry	
KING, Nancy Leigh	
KINSMAN, Marie Mattie	
KLAVER, Elaine Doreen	
KLFIN-NIRRFI INK Minnie	March 3, 2021

KNOWLES, Carroll Lee	
KONYK, William (aka Wasyl Konyk; aka Bill Konyk)	
KUHN, Carolyn Elisabeth	March 17, 2021
KUHN, Carolyn Elisabeth	
LAFRENIERE, Brian Michael	May 26, 2021
LAIDLAW, Ralph James	
LAIDLAW, Roseanne Marie	
LAKE, Ruth Marjorie	
LANDRY, George Michael	
LANE, Donald Arthur	•
LANE, Emily Odessa	
LANGILLE, Dennis Alan	
LANGILLE, Raymond John	
LANNON, Samuel F.	
LARSEN, Carla Mae	
LATIMER, Robert Stansfield	
LAVOIE, Marc Maurice	
LAWRENCE, Sheila Dianne	
LEBLANC, Maurice A.	
LEBRETON, Giselle Marie	
LEES, Lorraine Gertrude	
LEET, Jo Ann	
LEIGHTON, Jane Murphy	
LeLIEVRE, Alex John	
LEONARD, George Frederick, Jr.	
LESLIE, Lauchlin George	April 28, 2021
LeVATTE, Elizabeth Margaret	April 14, 2021
LEVESQUE, Marie Linda	
LEVY, Goldie Faye	
LEWIS, Edna Josephine	
LEWIS, Marjorie F.	
LEWIS, Susan B.	February 10, 2021
LEY, Daryll Patrick	
LIVINGSTON, Carl Bruce	March 31, 2021
LLEWELLYN, Robert Terry	March 24, 2021
LOCKHART, Ella M.	March 24, 2021
LOGAN, Mary Ellen Joyce	February 3, 2021
LOHNES, Paul Eldred	March 10, 2021
LOMBARD, Adrien Patrick	June 23, 2021
LONG, Kathleen Avis	
LONGMIRE, Joyce Lavaughan	
LORDLY, Joy Daphne	•
LOURENCO, Jose Jorge	•
LOURENCO, Natalia Pereira Eires	
LOWE, Ann Couseland	
LOWTHER, Marie Evelyn	
LOYST, Rhoda Theresa (aka Rhoda Teresa Loyst)	
LUNN, Glenda Marie (aka Glenda Lunn)	
LUNN, Joan Marilyn	
LUSBY, Harold Atkinson	
LYNCH, Maurice William	
LYONS, Anthony Edward	
MacAL OON Vovin Govern	
MacALOON, Kevin Gerard	
MacASYLL Poymond Eric	
MacASKILL, Raymond Eric	
MacAULAY, Barbara Olive	
MacBURNIE, Royden John	
MacDONALD, Clara Ann	
Machinia Al II Clara Ann	March 17 2021

MACDONALD, Elizabeth Jean	
MacDONALD, Gerald Joseph	
MacDONALD, James Duncan	
MacDONALD, John Bernard	
MacDONALD, Leo Samuel (aka Samuel MacDonald)	
MacDONALD, Leona (aka Leona Mary MacDonald)	
MacDONALD, Margaret Jean Neary	April 28, 2021
MacDONALD, Mary Christena	
MacDONALD, Mary Dawna	
MacDONALD, Mary Margaret	
MacDONALD, Myrna Virginia	
MacDONALD, Robert John	
MacDONALD, Roger Kent	
MacDONALD, Susan Ann	
MacDONELL, Theresa	
MacDONNELL, Filia May	
MacGEAN, John Robert (aka Robert MacGean)	
MacGILLIVRAY, Brian Scott	
MacGILLIVRAY, John Colin	
MacGILLIVRAY, John Daniel (aka John "Buddy" MacGillivray)	
MacINNIS, Marion Eloise	
MacISAAC, Johan Marie	•
MacISAAC, Mary Charlotte	•
MACKAY, Bernard Charles	
MacKAY, Colin Raymond	
MacKAY, Mary Allie	
MacKAY, Mildred Leverne (aka Mildred Laverne MacKay)	
MacKAY, Newton (aka Kenneth Newton MacKay)	
MacKAY, William Robert	
MacKEIGAN, Mary Elaine	
MacKEIGAN, Mildred	
MacKENZIE, Annabelle	
MacKENZIE, Catherine Sheila	
MacKENZIE, Catherine Sheha. MacKENZIE, Christina (aka Christena McKenzie)	1 '
MacKENZIE, Ogden Ross	
MACKIE, Darrin Wayne	
MacKINNON, Clarence Alexander	
MacKINNON, Ciarchice Alexander MacKINNON, Daniel Cyril	
MacKINNON, David Perrier	
MacKINNON, John Ronald	
MacKINNON, Patrick	•
MacKINNON, Sarah Agnes	
MacLEAN, Debra Ann	•
MacLEAN, Elizabeth (Betty)	
MacLEAN, John Howard	
MacLEAN, John Jerome (Jerry).	
MacLEAN, John Jeronie (Jerry)	
MacLEAN, Loyal Herbert	
MacLEAN, Mildred Marie	
MacLELLAN, Grace Pauline	
MacLELLAN, Orace Faume MacLELLAN, Patricia Margaret Catherine	
MacLEOD, George	
MacLEOD, George MacLEOD, Marilyn Ann (aka Marilyn Anne MacLeod)	
MACLEOD, Peggy Lynn (aka Peggy Lynn MacAskill)	
MacLEOD, Willard Allan	
THEOLOGY, IT HIGH A THEIR	cordary 3, 2021

MacLOED, Murdoch Gordon (aka Murdock Gordon MacLeod)	March 24, 2021
MacMILLAN, Donald Angus	
MacMILLAN, Philip Sawyer	January 27, 2021
MacMILLIAN, Rosalie Angela	May 12, 2021
MacMULLIN, Gregory Martin	March 3, 2021
MacMULLIN, John Bernard	
MACNAB, Roberta Sarty (aka Roberta Eulalie Macnab)	May 12, 2021
MacNAUGHTON, Lewis Cote	
MacNEIL, Loretta Mary	March 31, 2021
MacNEIL, Reverend Michael Hughie (aka Michael Hugh MacNeil)	
MacPHEE, Joyce Irene	
MacPHEE, William Neil	
MacPHERSON, Jessie Ann	
MacQUARRIE, Mary Ellen	
MacRAE, Angela Wanda Maria	
MacSWEEN, Bradley Scott	
MacWHA, Marilyn Joyce	
MADORE, Andrea Marie	
MAHANEY, Katherine Diana Rose	
MAILLET, Joseph Robert (aka J. Robert Maillet; aka Robert Maillet)	May 19, 2021
MAILMAN, Louisa Mae	June 16, 2021
MAILMAN, Louise Margaret	
MALIN, Ralph Frederick	
MALLOY, William Harlen	
MANNING, James Julian	
MANUEL, Thomas Roy	
MANUGE, Thomas Roy	
MARCH, Eleanor Ruby	
MARCH, Judith Marie	
MARCH, Madeline Gertrude	
MARCHAND, Elizabeth Ann (aka Elizabeth Marchand)	
MARGESON, Barbara Rose	
MARLOW, Joseph David	
MAROUN, Mary Josephine (aka Mary Josephine Merner)	
MARR, Alexander Louis	March 24, 2021
MARR, Barbara Barrymore	
MARR, Charlotte Patricia	
MARSH, Karl	
MARSHALL, Berma Millicent (aka Burma Millicent Marshall)	
MARTELL, Mary Alma	May 5, 2021
MARTELL, Nona Alayne	
MARTIN, Bonnie Patricia	
MARTINSON, Phyllis Lorna	
MASKELL, Susan Carol	
MASON, Calvin William	
MASON, Melvin Lorne	
MATKIN, Norman	•
MATTHEWS, Cynthia	June 9, 2021
MATTIE, Charlotte Elizabeth	May 26, 2021
MATTINSON, Kenneth Adair (aka Adair Mattinson; aka Adiar Mattinson)	June 2, 2021
MAYES, Florence Blanche	June 23, 2021
MAYO, Lloyd	
McBURNEY, Stephen G.	-
McBURNIE, Harold Gordon	
McCAFFREY, Anne Katherine	
McCAFFREY, Karen Lillian	
McCARTHY, Anna Mae	•
McCARTHY, Julia Laura	
McCAULEY, Normand Joseph (aka Normand McCauley)	
McCORMICK, Bernard Joseph	June 30, 2021

McCORMICK, Hilda Sarah	May 26, 2021
McCORMICK, John Ralph (aka Ralph McCormick)	March 10, 2021
McCRIMMON, Michael Robert	February 3, 2021
McCULLOCH, John Alisas (aka John Alvis McCulloch)	
McCULLOCH, Norma Alice	
McCULLOUGH, Joyce Virginia	
McCURDY, Linda Ann	
McCUTCHEON, Earle Ransford	
McDONALD, Barbara Jeanne	
McDONALD, Daniel Gregory	
McDOUGAL, Patti Lynn	
McEACHERN, Maureen	
McEWAN, Judith Kathleen	
McGINN, Rose Dorothy	
McGRATH, Ruth Alice	
McGUIGAN, Allan Charles	
McINNIS, Donna Winnifred Rose	
McINNIS, Gordon Allan	
McKENNA, Dorothy Vertilee	
McKENZIE, Carl Blendise (aka Carl Blennie MacKenzie)	
McLAUGHLIN, Emily Dawn Edna	March 10, 2021
McLEAN, John Dougald	
McLEAN, Karen A. (aka Karen Anne McLean)	
McMAHON, Mary Cleophus	
McMULLIN, Lawrence (aka Laurence McMullin; aka Larry McMullin)	February 17, 2021
McNEIL, Mary Josephine	June 2, 2021
McNEIL, Mary Lee (aka Mary Leonella McNeil)	June 9 2021
McRAE, David Mark	
MEAGHER, John Bernard	May 12 2021
MEISNER, Elva Gertie (aka Elva Meisner)	July 14 2021
MEISTER, Fronie	February 17, 2021
MEISTER, Fronie	February 17, 2021 May 5, 2021
MEISTER, Fronie	February 17, 2021 May 5, 2021 J.
MEISTER, Fronie	May 5, 2021 JJune 23, 2021
MEISTER, Fronie	
MEISTER, Fronie	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca)	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele	February 17, 2021May 5, 2021 JJune 23, 2021June 23, 2021April 28, 2021March 10, 2021April 21, 2021
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn)	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn)	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice. MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills)	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate]	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate] MILNE, Bonnie Marie	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate] MILNE, Bonnie Marie MILSOM, James Vernon	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate] MILNE, Bonnie Marie MILSOM, James Vernon MINOR, Elizabeth Marie	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate] MILNE, Bonnie Marie MILSOM, James Vernon MINOR, Elizabeth Marie MINOR, Elizabeth Marie MIRON-MORIN, Maxime	
MEISTER, Fronie	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate] MILNE, Bonnie Marie MILSOM, James Vernon MINOR, Elizabeth Marie MIRON-MORIN, Maxime MITCHELL, John B MITCHELL, William A	
MEISTER, Fronie	
MEISTER, Fronie	
MEISTER, Fronie	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLET, Marilyn Diane MILLET, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Rev. Dr. Robert Harry MILLS, Ronald Hartwell MILLS-MARTIN, Elizabeth Rose [cancelled, incorrect estate] MILNE, Bonnie Marie MILSOM, James Vernon MINOR, Elizabeth Marie MIRON-MORIN, Maxime MITCHELL, John B MITCHELL, William A MITSON, Paul Justin MITTON, Beulah Roberta MOFFATT, Anne Marie MOFFATT, Anne Marie MOFFATT, Anne Marie MOLLON, Reginald Lloyd	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Ronald Hartwell MILLS, Ronald Hartwell MILLS, Bonnie Marie MILNE, Bonnie Marie MILNE, Bonnie Marie MIRON-MORIN, Maxime MITOHELL, John B MITCHELL, John B MITCHELL, William A MITSON, Paul Justin MITTON, Beulah Roberta MOFFATT, Anne Marie MOLLON, Reginald Lloyd MOMBOURQUETTE, Gregory George	
MEISTER, Fronie	
MEISTER, Fronie MELANSON, Emma (née LeBlanc) (aka Emma Melanson; aka Emma Marie Melanson) MELANSON, Joseph Mederic (aka Medric Melanson; aka Medrick Melanson; aka Mederic Melanson) MELDRUM, Kenneth Anthony MELVIN, Beulah Agnes MENDONCA, Maria Fernanda (aka Frances Mendonca; aka Frances Maria Mendonca) MERRY, Dawn Adele MILBURN, Ellen Inga-Lill (aka Inga Milburn; aka Ellen Inga Lil Milburn) MILHAUSEN, Michael Grant MILLER, Joyce Bernice MILLETT, Marilyn Diane MILLETT, Marilyn Diane MILLS, Clarence Walter (aka Clarence W. Mills) MILLS, Ronald Hartwell MILLS, Ronald Hartwell MILLS, Bonnie Marie MILNE, Bonnie Marie MILNE, Bonnie Marie MIRON-MORIN, Maxime MITOHELL, John B MITCHELL, John B MITCHELL, William A MITSON, Paul Justin MITTON, Beulah Roberta MOFFATT, Anne Marie MOLLON, Reginald Lloyd MOMBOURQUETTE, Gregory George	

MONT, Madeleine Bernice (aka Madeline B. Mont)	
MOORE, Robert Allan	
MORASH, Anne Louise	
MORASH, Frederick D	
MORIARTY, Frances Elizabeth	
MORINE, Jean Gertrude	
MORRISON, Catherine Agnes	April 7, 2021
MORROW, Florence Isabel	
MORROW, Maurice William Gerard	
MOSER, Cyril MacDonald	
MOSHER, Louise Rhoda	
MOULTON, Exilda Myrtle	
MOWATT, Susan Jacklin Fielding	
MUELLER, Betty Lee	
MUISE, Lucille Marie	
MUNDEN, Genevieve Catherine (aka Jenny Catherine Munden; aka Genevieve Jenny Cather	
Munden)	
MUNROE, Ian Stanley (aka Ian Munroe)	
MUNROE, Wilda Bernice	April 28, 2021
MURPHY, Rhetta J. (aka Rhetta Jane Murphy)	
MURPHY, Sharon Anne	
MURRAY, Garth John	•
MURRAY, Joseph Wallace	
MYER, Ruth Shirley	
MYERS, John Douglas	
NADDAF, Mary (aka Mary Margaret Naddaf)	
NASH, John William	
NAUGLE, Robert Allen	
NAUSS, David Clifford	
NAWIJN, Judith Anne	March 24, 2021
NEGUS, Shannon Elizabeth	
NELSON, Catherine Pauline	
NELSON, Gordon Vincent	
NELSON, Sheevaun.	April 28, 2021
NELSON, Shirley Anne	
NEPEAN, Margaret Muriel	June 9, 2021
NEWELL, Murray Raymond	March 31, 2021
NEWMAN, John C.	March 31, 2021
NICHOLS, Alice Alberta	May 5, 2021
NICHOLS, Barry	March 31, 2021
NICHOLSON, Diane Joan	January 27, 2021
NICKERSON, Jack Murray	June 16, 2021
NICKERSON, James Anthony	July 7, 2021
NICKERSON, Layton Ephriam	February 3, 2021
NICKERSON, Margaret Elizabeth (aka Margaret E. Nickerson)	May 5, 2021
NICKERSON, Marion Gertrude	
NICKERSON, Nancy Lorraine	
NICKERSON, Patricia Mary	April 14, 2021
NICKERSON-BARTLETT, Shelly Bonnie (aka Shelly Bonnie Eileen Nickerson)	
NIGHTINGALE, Robert	April 14, 2021
NIPPARD, Brandon Robert	-
NOSEWORTHY, Bertram	•
NOWACZEK, Mary Anne	
NURSE, Thelma Frances	
NUTBEAN, Mary Catherine Beatrice	
O'CALLAGHAN, Emmett G. (aka Emmett George O'Callaghan)	
O'CONNOR, Patricia Marjorie	
O'HALLORAN, Clare	
O'LAUGHLIN, Ramona Constance	
O'LEARY, Owen Sanford	•
,	····· -, -, -, -, -, -, -, -, -, -, -, -, -,

O'NEIL, Earl Joseph	
O'NEIL, Stephen Patrick	June 2, 2021
O'NEILL, Karen Dianne	April 21, 2021
O'NEILL, Marion C	May 12, 2021
O'REILLY, Pamela Jean	February 17, 2021
O'TOOLE, Donna Doreen	June 23, 2021
OBEID, Obeid Saad	
ODO, Thomas Wilfred	
OICKLE, Murray Joseph, Jr	March 3, 2021
OLIVER, Phyllis Irene Gray	June 23, 2021
OSBORNE, Donna Evelyn	January 27, 2021
OUGH, Dorothy Betty	June 16, 2021
PACHOK, Irene	
PALMER, Charles Edward	April 14, 2021
PAPION, Claude Marcel	June 30, 2021
PARKER, Charles George	· · · · · · · · · · · · · · · · · · ·
PARKER, Kenneth Lawson	
PARKER, Kyle Isaac James	
PARKER, Ronald (aka Ronald Walton Parker)	
PATTERSON, Barbara Anne	
PATTON, Eric Joseph	· · · · · · · · · · · · · · · · · · ·
PEARCE, Derek John	
PEARCE, Karen Margaret	
PEARL, Beverley Sterling	
PEARL, Ramona Myrtle (aka Myrtle R. Pearl; aka Romona Myrtle Pearl)	
PEETOOM, Lola Marguerite	
PELHAM, Leonard Peter	
PENNER, Sheila Moira	
PENNY, Sheryl Marguerite	· · · · · · · · · · · · · · · · · · ·
PENNY, Thomas Winston	
PERO, Florence Doris	
PERO, Harold William	
PETRIE, Florence Angles "Faye" MacArthur	
PHELAN, Forrest Connell	
PHILLIPS, Albert Mervyn	
PHILLIPS, Malcolm Mebane	
PIERCE, Natalie Anina	
PIGGOTT, William Weston.	
PINEAU, Kenneth Paul Stephen	
PINEO, Myrtle Elanor	
PITMAN, Muriel Gaile	
PORTER, Lise M.	-
POTTIER, James A. (aka James Arthur Pottier)	
POWELL, Leslie Irving	
POWER, James Maurice	
POWER, Patricia	
POWERS, Shirley Elizabeth	
PREST, Merritt Roy	=
PRICE, Dorothy Elaine	•
PRINCE, Wanda	
PROCTER, Donald Arthur	
PULSIFER, Donald William	
PULSIFER, Reginald R. (aka Rex Pulsifer)	
PURCELL, Kenneth Joseph	
PYE, George Douglas	
PYKE, Karen Ann	
PYKE, Leslie Clossin	
PYNCH, Robert Alexander	
QUAINTANCE, David (Dave) Vincent	
	February 10, 2021

QUINN, Margaret	February 3, 2021
RAFUSE, Donald George	
RAFUSE, Harold (aka Harold Alfred Rafuse)	June 2, 2021
RAMSAY, Dianne Joy	
REAGH, Susan Elizabeth	
REARDON, Charles Francis	
REARDON, Venita	
RECHNITZER, Molly	March 3, 2021
REDMAN, David Joel	
REDMOND, Hugh Lloyd (aka Lloyd Redmond)	
REECE, Judith Elaine	•
REEVES, Elva Eileen (aka Elva Eileen Bezanson)	
REIBLING, Guenther	
REIBLING, Susan Jane	
REID, Clarence	
REID, Joseph Henry	
RENDELL, Elizabeth Helen (aka Elizabeth "Betty" Rendell)	
RENDELL, John Elric	
RHYMER, Steven Grant	
RICHARD, Thomas Clarence (aka Thomas C. Richard)	
RILEY, Clarence Edward	
RIPLEY, Joyce Marcella	
ROACH, Carolyn Sunshine (aka Carrie Sunshine Roach)	
ROACHE, Debbie Lynn	
ROBERT, Wallace Albert	
ROBERTS, Charles Burton	
ROBERTS, Lawrence Douglas, Jr.	
ROBERTS, Michael John	
ROBERTS, Wendy Olive Preston	
ROBERTSON, Mary Heather	
ROBINSON, Danette Patricia	
ROBINSON, France Elizabeth	
ROBINSON, Francis Adrian	
ROBINSON, George David	February 10, 2021
ROBINSON, Gerald Murray	April 21, 2021
ROBINSON, Ronald F	January 27, 2021
RODENHISER, Walter Clifford	June 16, 2021
ROGERS, Mabel Jean	April 7, 2021
ROGERS, Sarah Marcella	April 21, 2021
ROMARD, Yvon Joseph	June 2, 2021
ROOS, Wendy Janice	July 14, 2021
ROSATI, Vincent Sabato	February 10, 2021
ROSE, Theresa P.	
ROSS, David Charles	
ROSS, Lenley Warren (aka Lenely Warren Ross)	
ROSS, Maxwell Archibald	
ROSS, Phyllis (aka Phyllis Marie Ross)	June 16, 2021
ROSS, Terri Leslie	
ROY, Frank Wallace Kitchener	
ROY, Ian Wallace	
ROY, Olive Youlande	
RUMP, Paul Charles	
RUSHTON, Lorraine Mae	
RUSSELL, Maureen Marie	
RUTH, Josephine Mary	
RUTHERFORD, David Charles	
RYALL, David Valentine	
RYER, Meriam Roberta	
SALSMAN, Beverly Ann	
SALTER Dennis William	March 31 2021

SALTZMAN, Luvilla Mildred	
SAMPSON, James Gary	
SAMPSON, Marilyn Cecelia	
SAMPSON, Patricia Elaine	
SANDERSON, Donald Edward	May 5, 2021
SANDERSON, Hugh Keith	July 7, 2021
SARSON, Hazel Alberta	March 17, 2021
SARSON, Shirley E.	March 31, 2021
SAULNIER, Dorothy E.	
SAYADI, Hassan	
SCANLAN, Patricia Faye	
SCHOFIELD, Brenda Jane	May 5, 2021
SCHOFIELD, Pauline Bernice.	
SCHROTNER, Anneliese Elsa Kreszentia	
SCHROTTNER, Americae Elsa Rieszenda SCHROTTNER, Franz Josef	
SCOTT, Caroline Phelps	
SCOTT, Linda Anne	
SEDGEWICK, Marilyn Starr	
SEIFRIED, Judith Elaine	
SETTLE, Regina	
SHAUGHNESSY, Katherine Marguerite	April 21, 2021
SHAW, Elizabeth Beverley (aka Elizabeth Beverly Shaw)	April 7, 2021
SHAW, Sylvia Mae	February 3, 2021
SHAY, Darrell Whylie	April 7, 2021
SHEEHAN, Rebecca Isabelle (aka Rebecca Isabel Sheehan)	February 3, 2021
SHEPPARD, James Francis	
SHERMAN, Cora Mae Jean	
SHERMAN, Douglas Edward George	
SHERMAN, Lee Harrison	
SHORTELL, Evelyn Maude	
SIDDLE, Barry Lear	
SILLERS, Muriel Eilleen (aka Muriel Eileen Sillers)	
SILVER, Carolyn Esther	
SIMMONDS, Thomas	
SINGER, Barbara Jane	
SINGLETON, Corey Harlan	
SLATON, Pauline Lucy	
SLAUENWHITE, Donald Gerald	April 28, 2021
SLAUENWHITE, Sterling Leon	July 14, 2021
SMELTZER, Durling Reid	June 9, 2021
SMITH, Alan Graham	May 12, 2021
SMITH, Andrew Douglas	
SMITH, Edna Marguerite	
SMITH, Evelyn Mae	
SMITH, Harriett Mildred	•
SMITH, Helen Lorraine	
SMITH, Joan Bernice	
·	•
SMITH, Larry Allen	
SMITH, Margaret Mary	
SMITH, Mildred Melinda	
SMITH, Ruby Nora	
SMITH, Terrence William	
SMITH, Venita Laura	
SMITH, William Matthew (aka William Smith; aka William Mathew Smith)	April 28, 2021
SMITH, Willie Elijah	April 28, 2021
SNOW, Emily Diane (aka Emily Diana Snow)	
SNOW, Lily Jane Emma Maud	-
SNOWDON, Barbara Jane	
SOPER, Olive Vina.	
CDAVDC DDOVO Dovorty	May 5 2021

SPENCER, Donald Allister	February 3, 2021
SPURR, Henry Austin	March 17, 2021
STAILING, Robert Bruce	April 28, 2021
STEEVES, Isabel Mary	March 24, 2021
STEEVES, John Stephen	
STEPANIAK, Ernest Joseph	
STEWART, Josephine Adrianne	
STEWART, Kathleen Yvonne (née Smith)	
STEWART, Margaret Elizabeth	
STEWART, Phyllis Beatrice (aka Phyllis Stewart; aka Phyllis Mullen)	
STODDART, James Edward (aka Ted Stoddart)	
STOKES, Patricia Jane (aka Patricia Jane Streeter)	February 17, 2021
STONE, Catherine Anne	
STONE, Gordon James	
SURETTE, Carl Joseph	
SURETTE, Carl Joseph SURETTE, Doris Mae	
SURETTE, Bons Mae SURETTE, Elizabeth François (Frances)	
SURETTE, Joseph Franklin	
SURETTE, Kenneth Peter	
SUTHERLAND, Adele Louise	
SUTHERLAND, Alexandra Mary	
SWALLOW, Jean Elizabeth	
SWALLOW, Veronica	
SWAN, David Alexander	
SWIFT, Jacqueline Beryl	
SWIM, Langille Rupert	
SWIMM, Orley Nevera	
SWINAMER, Wayne Edward Clark	
SWINWOOD, Linda Alice	July 14, 2021
TALBOT, Ronald Douglas	March 31, 2021
TANNAHILL, Frances E.	May 26, 2021
TANNER, Amy Helena	April 28, 2021
TARDIF, Jerome Vianney	March 10, 2021
TATE, Dorothy Rose	March 10, 2021
TATTRIE, Carol G.	April 28, 2021
TAYLOR, Cynthia Jane	
TAYLOR, Everett Jess	
TAYLOR, Fred Marshall	
TAYLOR, Mary Etta Lynn	
TAYLOR, Michelle Martina Marie	
TAYLOR, Neil Edward	
TAYLOR, Ronald Carl	-
THISTLE, Mary Arlene	
THOMAS, Danny Lee	-
THOMAS, Franklyn Louis.	
THOMAS, Gordon Keith	
THOMPSON, Glen R	
THOMPSON, Kenneth Louis	
THOMPSON, Maria Pauline	
THOMPSON, Williamina Hamilton	
THORBURNE, Jean D.	
TIGHE, John Blaise	
TIMMINS, Phyllis Arlene	
TITUS, Peter H. (aka Peter Hantford Titus	
TOBIN, Marie Elizabeth	
TOMNEY, Stephen Peter	
TOOKE, Deborah Ann	
TOOLEY, Linda Marion	
TOOMEY, Cayla	April 28, 2021
TOWNSEND Viola Vivian	March 17, 2021

TOZER, Ada Eliza Mary	
TRACEY, Robert Fraser	
TRAVIS, C. George	
TREVOR, Wanda Elizabeth	
TULL, Vernal Alleyne	June 9, 2021
TURNER, Blanche Eleanor	June 23, 2021
TURNER, Jerome Kendall (Jerry)	January 27, 2021
VAN DEN EYNDEN, Antonia Marie	June 16, 2021
Van DONINCK, Bernard Francis	
VATER, Belinda Carol	
VEINOT, Carol Gladys	
VEINOTTE, Ferne Alberta	
VEINOTTE, Naaman	
VEINOTTE, Yvette	
VICKERS, Anne Marie (aka Ann Marie Vickers)	
VINCENT, Hugh Brian	
VINCENT, Olvia Caroline	
VINER, Sollie (aka Sol Viner)	
VOKEY, Olive (aka Olive Blanche Vokey)	
von KURNATOWSKI, Renate (aka Renate Maria Kurnatowski)	
WAGNER, Gerald Emery	
WALKER, John Edward	
WALSH, Alice Veronica	
WALSH, Edward Allan	
WALSH, Fay Annette	
WALSH, Gladys Muriel	
WALSH, Mary Enid	
WALSH, Russell E.	
WALTERS, Douglas Whitman	
WAMBOLT, Melvin Stanley	
WAMBOLT, Roy Melvin	May 5, 2021
WARNELL, Sharon Lee	June 16, 2021
WASSILASKUS, Marjorie Anastasia (aka Marjorie A. Wassilaskus)	February 17, 2021
WATERFIELD, Elaine Louise	
WATT, Douglas Alden	
WAYTE, Janet Heather	March 24, 2021
WEAGLE, Isabel Bertha	
WEAGLE, Willard Robinson	
WEATHERBEE, Ernest Everett (aka Ernest E. Weatherbee)	
WEATHERBY, Holly Ruth	
WEBB, John Todd	
WEBBER, Annie Laura	
WEBBER, Barbara Louise	
WEIDHAAS, Loretta Elizabeth	
WELD, Gordon Beecher	
WELDON, Richard Laurence	
WENTZEL, Kurt Anthony (aka Kurt Wentzel)	
WENTZELL, Daren Hubert	
WENTZELL, Phyllis Amelia (aka Phyllis A. Wentzell)	
WEST, Ruth Kathleen	
WHALEN, Michael Stephen	•
WHITE, Dwight Lancelot	
WHITE, Earle Sanford	
WHITEHEAD, John Paul	
WHITTLE, Albert Ellsworth	
WHYNOT, Hilda Florence	
WHYNOT, Kevin Wayne (aka Kevin Whynot)	
WIGGS, Catherine Anne	
WILE, Elaine Maxine	April 28, 2021
WILE Eye Saline (aka Eye Salane Wila)	February 10, 2021

WILKINS, Peter Carl	April 28, 2021
WILKINSON, Peter Fraser	
WILLIAMS, Bernard Eugene	April 21, 2021
WILLIAMS, Gerald Frederick	February 10, 2021
WILLIAMS, Joyce	May 12, 2021
WILLIAMS, Owen Patrick	January 27, 2021
WILLIS, Nema Loretta	March 10, 2021
WILSON, James Walter	March 24, 2021
WINKLER, Anneliese Margot	February 3, 2021
WINTER, Jean Marie	April 7, 2021
WINTERS, Danna Darlyn (aka Danna Winters)	
WINTERS, Marilyn Dawn	
WOLFE, Bruce Phillip	February 17, 2021
WOLFE, Florence Louise	March 31, 2021
WOLFE, Margaret Olive; Margaret O. Wolfe; aka Margaret Wolfe)	March 10, 2021
WONG, Philip Ying Fai (aka Philip Y.F. Wong)	June 9, 2021
WOOD, John James	May 19, 2021
WOOD, Nelson Nathaniel	February 17, 2021
WOOD, Ronald Norman	April 21, 2021
WOODHAMS, Harold Charles Lindsay	March 31, 2021
WOODINGTON, Claude Leighton	June 30, 2021
WORTH, Lawrence Arthur	
WORTHYLAKE, Anna May	April 21, 2021
WOURNELL, James Willoughby	May 5, 2021
YEADON, Pauline Joan	June 2, 2021
YOUNG, Donald Archibald (aka Donald A. Young; aka Donald Young)	February 17, 2021
YOUNG, John Wayne	
YOUNG, Marina Jessie	
YOUNG, Marion Elizabeth (aka Marian Elizabeth Young)	
YOUNG, Robert S.	January 27, 2021
YOUNG, Susan Ann	July 7, 2021
YOUNG, Thomas	March 17, 2021
YOUNG, Thomas Russell	March 24, 2021
ZINCK, Harold Lemual	April 14, 2021
ZINCK, Julia Dianne	
ZWICKER, Barbara Jean	February 3, 2021
ZWICKER, Bernard William	
ZWICKER, Deborah Frances Isobel (aka Deborah F.I. Zwicker)	February 24, 2021

Publishing Information

The *Royal Gazette* is published every Wednesday. Notices <u>must</u> be received by the Royal Gazette office **not later than 4:30 pm on Monday** in order to appear in that Wednesday's issue.

Prepayment is required for the publication of all notices. Cheques or money orders should be made payable to 'Minister of Finance'. All notices, subscription requests and correspondence should be sent to:

Office of the Royal Gazette Department of Justice 1690 Hollis Street, 10th Floor PO Box 7 Halifax NS B3J 1T0 Telephone: 902-424-8575

Email: RoyalGazette@novascotia.ca

Website: www.novascotia.ca/just/regulations/rg1/

The Royal Gazette Part I is available online beginning with the January 4, 2006 issue at the above website

Royal Gazette Part I Fees (15% HST included)

RESTATED ORDER #2 OF THE CHIEF MEDICAL OFFICER OF HEALTH UNDER SECTION 32 of the HEALTH PROTECTION ACT 2004, c. 4, s. 1.

July 14, 2021

Update Log:

July 14, 2021: Sections 2.2, 3.5A, 13A, 13.2, 15-15.3, 17, 18.1, 18.3, 19-19.2, 20, 20.3, 21A, 23.1, 24, 24.2, 24.4, 26.4, 26.5, 27, 28, 28.1, 30, 31.1, and 31.2 (amended) and Section 15.1 2 (new)

Section 13.1.2 (new) July 7, 2021: Sections 3.1, 3.5A, 3.5, 7 and 18.3 (amended) and Section 7.2 (new) July 5, 2021: Sections 2, 2.2, 3.2, 3.5, 4-12.1, 13A, 20, and 21A (amended) June 30, 2021: Sections 2-2.3, 3.1-3.2, 3.5A, 3.5, 4-4.2, 5, 12.3-12.4, 13A, 13.2, 15-15.3, 15.6, 16-16.2, 17-17.1, 18.3, 20-20.4, 21A, 24, 24.2, 24.4, 27, 28-28.3, 29, 30 and 31.1 (amended)

31.1 (amended) June 24, 2021: Sections 1, 3.5 and 3.6 (amended) and Section 3.5A (new) June 23, 2021: Sections 1, 2, 2.2, 2.3, 3.1, 3.5, 4, 4.2, 12.2, 12.3, 15.6, 24.4 and 27 (amended) and Section 3.6 (new)

June 16, 2021: Sections 2, 2.2, 2.3, 3.2, 13A, 13.2, 14, 15-15.6, 18.3, 20-20.9, 21A, 24, 24.2-24.5, 27, 28-28.2, 29, 30, 31.1 and 31.2 (amended) and Sections 17.1, 24.1 and 28.3 (new)

June 4, 2021: Sections 2.2, 13.2, 15.3, 15.3.1, 18.1 and 28.3 (amended) May 31, 2021; Sections 2, 2.2-2.3, 3.2, 13A, 13.2-13.5, 14, 15-15.3, 15.4.2, 18, 19-19.1, 20, 20.3-20.7, 21A, 23.1, 24.1-24.5, 26.4-26.5, 28, 28.2, 29, 31-31.2 (amended) and Sections 15.3.1-15.3.2 (new)

May 21, 2021: Sections 2-2.3, 10-10.1, 12.1, 12.3, 13A, 18.3, 18.7, 19, 20, 21A and 31-31.2.1 (amended) and Sections 15.3.1, 26.4-26.5 and 28.3 (new)

May 13, 2021: Sections 2-2.3, 3.2, 7-7.1, 13A, 13.2, 20 and 21A (amended) and Sections 13.5-13.6 (new)
May 8, 2021: Sections 2-2.3, 3.1.3, 2, 4.4, 2, 10, 11, 12, 1, 12, 3.12, 4, 13A, 19.19, 2

May 8, 2021: Sections 2-2.3, 3.1-3.2, 4-4.2, 10, 11, 12.1, 12.3-12.4, 13A, 19-19.2, 20, 21A, 22 and 31.1 (amened), and Sections 13.4 and 31.1.2 (new) May 4, 3021: Sections 2.2, 28.2 and 31.1 (amended) and Section 13.3 (new) April 29, 2021: Sections 24.3 and 31.2 (amended) and Section 31.2.1 (new)

- TO: 1.) All persons residing in or present in the Province of Nova Scotia;
 - 2.) All not-for-profit and for-profit businesses and organizations operating or carrying on business in Nova Scotia;
 - 3.) Such other persons or entities as may be identified by the Chief Medical Officer of Health or otherwise as set out in this Order.

ORDER made pursuant to Section 32 of the Health Protection Act (Nova Scotia)

.

- (a) "self-isolate" means the requirement of any person who has COVID-19 to remain separate from others in such places and under such conditions to prevent or limit the direct or indirect transmission of COVID-19;
- (b) "self-quarantine" means the requirement of any person who has been exposed or may have been exposed to COVID-19 during its period of communicability to restrict that person's activities in order to prevent disease transmission during the incubation period for this disease.
- 2. Effective 11:59p.m. July 5, 2021, all persons are prohibited from entering Nova Scotia, except as stated herein or set out elsewhere in this Order:
 - (a) permanent residents of Nova Scotia, Newfoundland and Labrador, Prince Edward Island, New Brunswick ("the Atlantic Provinces"), and their spouses, partners and dependant children when travelling together;
 - (b) persons travelling to Nova Scotia from a province or territory outside the Atlantic Provinces:
 - (c) persons travelling to Nova Scotia from outside Canada who are eligible to enter Canada in accordance with the Federal Order in Council 2021-0313, as amended.
- 2.1 Where any person travels into Nova Scotia in contravention of this Order, a peace officer is hereby authorized and directed to return that person to an interprovincial border and require the person to leave the Province immediately or at such a time as may be directed, and in doing so, may exercise authority under section 46 of the Health Protection Act
- 2.2 Effective 11:59p.m. July 5, 2021, all persons seeking entry into Nova Scotia must:
 - (a) apply through the Nova Scotia Safe Check-in form and complete daily digital check-ins, located at: https://travel-declaration.novascotia.ca/en, except:
 - (i) persons travelling to Nova Scotia from Newfoundland and Labrador, or Prince Edward Island and New Brunswick;
 - (ii) professional truck drivers;
 - (iii) persons who have received compassionate exceptions; and
 - (iv) essential healthcare workers who travel to and from Nova Scotia and another province or territory to carry out their work duties on an ongoing regular basis or on a locum basis.

WHEREAS Section 32 of the Health Protection Act states.

- 32 (1) Where a medical officer is of the opinion, upon reasonable and probable grounds, that:
 - (a) a communicable disease exists or may exist or that there is an immediate risk of an outbreak of a communicable disease;
 - (b) the communicable disease presents a risk to the public health; and
 - (c) the requirements specified in the order are necessary in order to decrease or eliminate the risk to the public health presented by the communicable disease.

the medical officer may by written order require a person to take or to refrain from taking any action that is specified in the order in respect of a communicable disease.

WHEREAS COVID-19 has been identified as a communicable disease that presents a risk to public health as defined under s.4(b) of the *Health Protection Act*, and;

WHEREAS I am the Chief Medical Officer of Health for the Province of Nova Scotia and am of the opinion, upon reasonable and probable grounds, that

- (a) a communicable disease (COVID-19) exists; and that there is an immediate risk of an outbreak of the communicable disease;
- (b) the communicable disease presents a risk to the public health; and
- (c) the requirements specified in the order are necessary to decrease or eliminate the risk to the public health presented by the communicable disease, and:

WHEREAS as the Chief Medical Officer of Health, I have determined it necessary to issue this Order to the Class of Persons to decrease the risk to public health presented by COVID-19.

Please be advised that:

I, Dr. Robert Strang, Chief Medical Officer of Health, order the following:

PART I ENTRY, ISOLATION AND QUARANTINE REQUIREMENTS

1. In this Order

2

- (b) disclose to the Chief Medical Officer of Health or his delegate information related to their COVID-19 vaccination, including:
 - (i) whether they received a COVID-19 vaccine;
 - (ii) the brand name or other information that identifies the vaccine that was administered;
 - (iii) the dose received, and
- (c) if they are fully vaccinated, provide the Chief Medical Officer of Health or his delegate evidence of COVID-19 vaccination.
- 2.3 For greater clarity, the evidence of COVID-19 vaccination must be a document issued by the government or non-governmental entity that administered the COVID-19 vaccine and must contain the following:
 - (a) the name of the person who received the vaccine;
 - (b) the name of the government or the name and civic address of the non-governmental entity that administered the vaccine;
 - (c) the brand name or any other information that identifies the vaccine that was administered; and
 - (d) the dates on which the vaccine was administered.
- $3.1\ {\rm The}$ requirements of section 3.2 apply to all persons residing in or present in the Province of Nova Scotia who:
 - (a) have travelled outside the Atlantic Provinces; or
 - (b) are identified as a close contact of a person who has or has been diagnosed with COVID-19;
 - (c) are identified as a person diagnosed with COVID-19; or
 - (d) have been tested for COVID-19 due to the presence of symptoms or as directed by public health and are awaiting the results of their test.
- 3.2 Effective 11:59p.m. July 5, 2021, those persons listed in section 3.1 must:
 - (a) self-quarantine or self-isolate, for:

3

- (i) the period commencing on the day of entry into Nova Scotia, and continuing thereafter for 14 consecutive days or as directed by a Medical Officer of Health, or
- (ii) the first day of close contact, or first day of symptoms, testing, diagnosis, and continuing thereafter for 14 consecutive days or as directed by a Medical Officer of Health.
- (b) During the 14-day period, conduct themselves in such a manner as not to in any way expose any other person to infection or potential infection from COVID-19, and follow all infection control instructions given to them on the Government of Nova Scotia's website, at: https://novascotia.ca/coronavirus/, or given to them by Telehealth 811 staff, public health staff or any other staff of a healthcare facility to which they may seek or receive treatment.
- (c) After the 14-day period in section 3.2(a.) has lapsed, they may cease self-isolation or self-quarantine if they do not exhibit symptoms of COVID-19.
- 3.3 All persons are encouraged to complete the online assessment tool should they exhibit symptoms of COVID-19.
- 3.4 For greater certainty, persons required to self-isolate or self-quarantine in accordance with section 3.2 must remain in their residence or residence grounds and otherwise remove themselves from the presence of others in public while they may be infectious during the 14-day period, so that all precautions necessary to protect others are in place. Specifically, such persons must not enter any buildings, public transportation, or other enclosed spaces (other than their residence) where other people are present.
- 3.5A For the purpose of sections 2.2, 2.3 and 3.5:
 - (a) "fully vaccinated" means receipt of 1 dose of a vaccine authorized as a 1 dose vaccine series such as Janssen plus 14 days, or 2 doses of a vaccine authorized as a 2 dose vaccine series such as Pfizer, Moderna or AstraZeneca plus 14 days;
 - (b) "partially vaccinated" means receipt of 1 dose of a vaccine authorized as a 2 dose vaccine series such as Pfizer, Moderna or AstraZeneca plus 14 days; and
 - (c) "vaccine" means a vaccine against COVID-19 that has been approved by either the Public Health Agency of Canada or the World Health Organization.
- 3.5 Notwithstanding section 3.2(a):
 - (a) persons who receive a negative COVID-19 test result after being referred for testing by the online assessment tool due to the presence of COVID-like symptoms may cease self-quarantine on the date of receipt of the negative test result;

- (b) persons who are required to self-isolate or self-quarantine may leave their
 - (i) to undergo COVID-19 testing as directed by a Medical Officer of Health;or
 - (ii) for 1 outing per day for outdoor exercise within walking/running distance of their home or isolation site for a maximum of 1 hour,
- (c) persons travelling from a province or territory outside the Atlantic Provinces to Nova Scotia, must adhere to the following:
 - (i) persons who are fully vaccinated before arriving in Nova Scotia are not required to self-quarantine on arrival in Nova Scotia;
 - (ii) persons who are partially vaccinated before arriving in Nova Scotia must self-quarantine for at least 7 days on arrival in Nova Scotia and must not cease self-quarantine until they receive 1 negative test result while in Nova Scotia, with testing on day 5 or 6;
 - (iii) persons who have had COVID-19, have recovered from it in the last 12 weeks before entering Nova Scotia, and are partially vaccinated must self-quarantine for 7 days on arrival into Nova Scotia but are not required to be tested; and
 - (iv) persons who have not received any vaccine before arriving in Nova Scotia must self-quarantine for 14 days in accordance with section 3.2(a).
- (d) persons travelling to Nova Scotia from outside Canada who eligible to enter Canada in accordance with the Federal Order in Council 2021-0313, as amended:
 - (i) who are fully vaccinated (the "fully vaccinated traveller" as set out in the Federal Order in Council 2021-0615) are not required to self-quarantine on arrival in Nova Scotia; and
 - (ii) who are not fully vaccinated are required to self-quarantine on arrival in Nova Scotia in accordance with Federal Quarantine requirements.
- 4. Notwithstanding sections 2, 3.1 and 3.5, persons who are well and showing no symptoms of COVID-19 may enter Nova Scotia for the purposes of facilitating child sharing between parents under an order or agreement providing for joint custody, and in such instances, both the persons facilitating custody and children showing no symptoms of COVID-19 are exempt from the requirement to self-quarantine.

- Further to section 4, parents and children entering or leaving Nova Scotia for the purposes of facilitating child sharing arrangements must adhere to the self-quarantine requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-child-custody.pdf.
- 6. Notwithstanding sections 2, 3.1 and 3.5, individuals who are well and showing no symptoms of COVID-19 and are engaged in a legal proceeding in Nova Scotia, whether the accused, victim, witness, party or lawyer in such proceeding, may enter Nova Scotia for participation in the legal proceeding and are exempt from self-quarantine if the person adheres to the COVID-19 Protocol for Exempt Travelers established by the Chief Medical Officer of Health. located at:

https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-exempt-travellers-en.pdf.

- 7. An employer or contractor of any Temporary Foreign Worker entitled to enter Nova Scotia pursuant to the Federal Orders in Council 2021-0614 and 2021-0615, as amended, must first, before the Temporary Foreign Worker enters Nova Scotia, satisfy me, as Chief Medical Officer of Health, that the employer or contractor has made adequate provision for compliance with the Federal Quarantine requirements applicable to the Temporary Foreign Workers in the Agriculture and Seafood Sector.
- 7.1 In addition, the employer or contractor and the Temporary Foreign Worker must, for the duration of the entire work period in Nova Scotia:
 - (a) adhere to all applicable terms and conditions of this Order; and
 - (b) adhere to the COVID-19 Protocol for Temporary Foreign Workers employed in Agriculture and Seafood Sectors established by the Chief Medical Officer of Health, located at: https://novascolia.ca/coronavirus/docs/COVID-19- https://novascolia.ca/coronavirus/docs/COVID-19- Protocolfortemporaryforeignworkers-agriculture-and-seafood-sectors-en.pdf.
 - (c) comply with any direction issued by me, as Chief Medical Officer of Health, or a Medical Officer of Health with respect to the Temporary Foreign Worker and their employment in Nova Scotia.
- 7.2 For greater certainty, a Temporary Foreign Worker in the Agriculture and Seafood Sector, on arrival in Nova Scotia, is required to quarantine in accordance with the Federal Quarantine requirements unless they are assessed as exempt from quarantine in accordance with Federal Order in Council 2021-0615.
- 8. Notwithstanding sections 2, 3.1 and 3.2, persons who are required to travel to Nova Scotia or outside Nova Scotia for essential health services, with accompanying support persons as permitted by health authority policy, are exempt from the requirement to self-quarantine but must adhere to the COVID-19 Protocol for Exempt Travelers established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-exempt-travellers-en.pdf

- 9. For the purpose of section 9.1, "rotational worker" means a person who lives in Nova Scotia and travels to work in another province or territory in Canada on a regular schedule or travels outside Canada on a regular schedule and is exempt from self-quarantine under the federal Quarantine Act.
- 9.1 Notwithstanding sections 2, 3.1 and 3.5, rotational workers must adhere to the self-quarantine and COVID-19 testing requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Directive-on-Exceptions-for-Rotational-Workers.pdf.
- 10. Notwithstanding sections 2, 3.1 and 3.5, specialized workers who are:
 - (a) needed for urgent work on critical infrastructure that cannot be done by individuals from within the Province and such work is crucial to the functioning of the Province; or
 - (b) needed for urgent work that cannot be done by individuals from within the Province and are necessary to preserve the viability of one or more Nova Scotia businesses,

are permitted to enter Nova Scotia from any geographic location outside Nova Scotia. A list of critical infrastructure included in this definition is available at: https://www.publicsafety.gc.ca/cnt/ntnl-scrt/crtcl-nfrstrctr/esf-sfe-en.aspx.

- 10.1 Specialized workers permitted to enter Nova Scotia in accordance with section 10 must adhere to self-quarantine and COVID-19 testing requirements established by the Chief Medical Officer of Health, located at:
- https://novascotia.ca/coronavirus/docs/COVID-19-Directive-on-Exceptions-for-Specialized-Workers.pdf.
- 11. Notwithstanding sections 2, 3.1 and 3.5, fish harvesters required to enter Nova Scotia from any geographic location outside Nova Scotia to perform the commercial or licensed activity of catching fish and other seafood for market or other approved activities, excluding recreational fishing, must adhere to self-quarantine requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-fish-harvesters-en.pdf.
- 12.1 Notwithstanding section 3.1 and 3.5, workers who are essential to the movement of people and goods, and who must enter Nova Scotia as part of their duty requirements, are exempt from the requirement to self-quarantine, particularly:
 - (a) healthy workers in the trade and transportation sector who are employed in the movement of goods and people across the Nova Scotia border by land, air, or water, including truck drivers, crew, maintenance and operational workers on any plane, train or ship;
 - (b) Canadian Military and Defence Team personnel, Coast Guard, RCMP, Canadian Border Services Agency, and Canadian Security Intelligence Service;

- (c) first responders, including police, fire, EHS paramedic workers; and
- (d) essential healthcare workers who travel to and from Nova Scotia and another province or territory to carry out their work duties on an ongoing regular basis or on a locum basis

but must adhere to the COVID-19 Protocol for Exempt Travelers established by the Chief Medical Officer of Health, located at:

https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-exempt-travellers-en.pdf.

- 12.2 For greater clarity, airline crew employed in the movement of people and goods and who are required to travel to Nova Scotia or from Nova Scotia to carry out their work duties are exempt from the self-quarantine requirements set out in sections 3.2 and 3.5.
- 12.3 Persons exempt under sections 12.1 to 12.2 must practice physical distancing of 2 metres (6 feet) to the best of their ability, follow all public health recommendations, closely self-monitor and must self-isolate or self-quarantine should they exhibit any COVID-19 symptoms as set out in the online assessment tool.

PART II PHYSICAL DISTANCING, GATHERING LIMITS, MASKS AND FACE COVERINGS

- 13A Effective 8:00a.m. July 14, 2021, except where otherwise stated in this Order the requirements for physical distancing, gathering limits, masks and face coverings apply to all persons present and residing in Nova Scotia.
- 13. All persons present and residing in Nova Scotia must maintain physical distancing of 2 metres (6 feet).
- 13.1 All persons present and residing in Nova Scotia must not participate in any gatherings, whether indoors or outdoors, unless subject to a specific exception set out in this Order.
- 13.2 Notwithstanding sections 13 and section 13.1, persons may gather together for informal gatherings with their immediate family members residing in the same household and close social contacts up to a maximum of 25 persons indoors and 50 persons outdoors, and they are not required to practice physical distancing or masking, unless they are in a public place where masks are required.
- $13.3\ Notwith standing\ sections\ 13.1\ and\ 13.2\ and\ for\ greater\ certainty,\ parties\ to\ a\ child\ sharing\ arrangement,\ or\ an\ order\ or\ agreement\ providing\ for\ joint\ custody:$
 - (a) may facilitate and participate in such child sharing or custody arrangement between households.

- (g) funerals, visitations and receptions;
- (h) faith gatherings and drive-in or parking lot faith service; and
- (i) drive-in movie theatre,

while adhering to the following:

- (a) must adhere to the informal gathering limits set out in section 13.2 when not hosted by a recognized business or organization (i.e. event planner, event venue):
- (b) practice physical distancing of 2 metres (6 feet), except persons gathered informally in accordance with section 13.2;
- (c) practice masking requirements set out in section 18.2 unless exempt under section 18.3; and
- (d) if attending a drive-in or parking lot faith service, or drive-in movie theatre, all persons in attendance must adhere to the following:
 - (i) the service or movie is conducted over speakers or by remote radio broadcast:
 - (ii) there is no contact between cars and no transfer of things between cars:
 - (iii) participants remain in their respective vehicle while "attending" the service, unless accessing a washroom facility or concession stand and while doing so, must adhering to protocol for gathering limits, physical distancing, and masking requirements set out in this Order;
 - (iv) participants practice physical distancing among vehicles, such that vehicles must be at least 2 metres (6 feet) apart from each other; and
 - (v) participants in the same car adhere to the gathering limits of section 13.2.
- 15.1 Notwithstanding sections 15, where any of the activities listed in section 15 are hosted by an existing business or organization legally operating in Nova Scotia, then the business or organization:
 - (a) may host up to 50% of legal capacity of the indoor establishment in which the event or activity is held, to an overall maximum of 150 persons, or up to a maximum of 250 persons if the event or activity is outdoors;

but

- (b) must adhere to the self-quarantine requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-child-custody.pdf if a parent or child develops symptoms or tests positive for COVID-19.
- 13.4 For the purpose of section 13.5, an "illegal public gathering" is defined as a gathering that does not comply with the requirements of this Order, including:
 - (a) the attendance limits applicable to gatherings, whether indoors or outdoors:
 - (b) physical distancing requirements; and
 - (c) masking requirements.
- 13.5 For greater certainty, persons are prohibited from:
 - (a) organizing an in-person gathering, including requesting, inciting, or inviting others to attend an illegal public gathering;
 - (b) promoting an illegal public gathering via social media or otherwise; or
 - (c) attending an illegal public gathering of any nature, whether indoors or outdoors.
- 14. Notwithstanding section 13.1, persons may engage in outdoor physical activity while adhering to the gathering limitations set out in section 13.2 or the gathering limitations while participating in an outdoor fitness class and other recreational or leisure activities as set out in sections 28 and 28.1, and in doing so, must adhere to physical distancing requirements of 2 metres (6 feet) as set out in section 13.
- 15. Persons may engage in the following events and activities, whether indoors or
 - (a) social events;
 - (b) arts and culture events;
 - (c) sports, recreation and physical activity events;
 - (d) special events;
 - (e) festivals;
 - (f) weddings and receptions;

- (b) must ensure all persons in attendance practice physical distancing and masking requirements set out in this Order;
- (c) must have a COVID-19 Prevention Plan;
- (d) may submit specific organizational plans to nseconomy@novascotia.ca for large venues such as Scotiabank Centre, Halifax Exhibition Centre, Halifax Convention Centre, Centre 200 and Speedworld. Plans will be reviewed by CCH and LAE staff and approved by the Chief Medical Officer of Health.
- 15.1.1 Business and organizations may host events listed in section 15 virtually while adhering to the following:
 - (a) the capacity limits set out in section 15.1(a) to support the delivery of the event; and
 - (b) their COVID-19 Workplace Prevention Plan to carry out virtual production in accordance with section 26.
- 15.1.2 For greater clarity, persons may participate in dancing when attending an event listed in section 15 but must adhere to the applicable public health requirements when the event is hosted by a person informally, or hosted by a recognized business or organization.
- 15.2 Notwithstanding section 15, players, participants, and officials:
 - (a) may engage in organized sport practice and training, games and regular league play recreational, amateur, and professional), up to a maximum of 25 persons indoors and 50 persons outdoors without physical distancing, or in multiple groups of 25 persons indoors or 50 persons outdoors with physical distancing between groups;
 - (b) where competition is not possible with the number restrictions stated in (a), Sport Nova Scotia, in consultation with CCH, may, through an application process, provide approval to the Provincial Sport Organization, for additional numbers in the field of play. In addition, registered organizations may host competitions as part of an event while adhering to the standard sport participation numbers on the field;
 - (c) may engage in organized arts and culture (amateur) rehearsals and performances, up to a maximum of 25 persons indoors and 50 persons outdoors without physical distancing, or in multiple groups of 25 persons indoors or 50 persons outdoors with physical distancing between groups;
 - (d) are recommended to wear a mask

but

- (e) may only engage in tournaments when affiliated with or hosted by a Provincial Sport Organization following their respective Return to Sport Plan; and
- (f) spectators are permitted with a limit of 50% of indoor capacity of the facility up to a maximum of 150 spectators or up 250 spectators outdoors, when the organized sport practice, training and competition (recreational, amateur, and professional), or the organized arts and culture rehearsals (amateur) is hosted by a recognized business or organization. Plans for hosting multiple groups, each group meeting the gathering limits, are permitted if following the guidance outlined by CCH.
- 15.3 Notwithstanding sections 15, 15.1 and 15.2, professional artists, musicians, dancers and actors may engage in rehearsal and performances up to a maximum of 25 persons indoors and 50 persons outdoors, without physical distancing while adhering to the following:
 - (a) the masking and spectator requirements set out in sections 15.2(d) and (f);
 - (b) an appropriate COVID-19 Workplace Prevention Plan in accordance with section 26: and
 - (c) the Covid-19 Guidance for Musicians (Vocalists and Instrumentalists) located at: https://novascotia.ca/coronavirus/docs/COVID-19-Guidance-for-Vocalists-and-Instrumentalists.pdf.
- 15.4 Any person, business or organization that hosts an event authorized by sections 15 to 15.3 is responsible for maintaining oversight of the gathering and for ensuring that all persons in attendance comply with the requirements of this Order.
- 15.5 For greater clarity, the maximum number of persons requirement set out in section 15.1 applies to meetings and training hosted by, including but not limited to:
 - (a) provincial and municipal governments;
 - (b) private businesses and organizations;
 - (c) first responder organizations;
 - (d) organized clubs; and
 - (e) mental health and addictions support group meetings,

all persons in attendance must adhere to the masking requirements set out in section 18.2 unless exempt under section 18.3.

- 15.6 Notwithstanding sections 13 and 15.5, any person, volunteer or employee who provides emergency first response are not required to adhere to the physical distancing requirement or gathering limits during an emergency, when carrying out duties or when necessary during training.
- 16. For greater clarity, physical distancing and the person limit rule set out in sections 13, 13.1 and 13.2 do not apply to the following organizations, activities, persons, or places:
 - (a) hospitals as defined in the Hospitals Act and a health authority as defined in the Health Authorities Act;
 - (b) profit, not-for-profit or government operated Department of Community Services funded organizations or representatives that are covered under the Homes for Special Care Act and the Children and Family Services Act including places of safety for children and youth, and customized placements for persons with disabilities:
 - (c) profit or not-for-profit Department of Health and Wellness funded long-term care facilities licensed under the *Homes* for Special Care Act or home care agencies funded under the *Homemaker Services Act*;
 - (d) persons providing care under the self-managed care program, supportive care program, caregiver benefit program funded by the province of Nova Scotia;
 - (e) persons providing support under the Independent Living Support, Supported Apartment and Supervised Apartment Programs funded by the Department of Community Services;
 - (f) homeless shelters receiving operational grants from the Department of Municipal Affairs and Housing, and those operated by religious and other voluntary organizations;
 - (g) unlicensed child-care facilities;
 - (h) a place designated or established under the authority of the *Correctional Services Act* or the *Youth Criminal Justice Act* (Canada) for the supervision or custody of offenders and includes community-based correctional services;
 - (i) any administrative tribunal, arbitration proceeding or court operating essential services in the Province under the authority of any provincial or federal enactment, including but not limited to, a justice centre or courthouse under the authority of the *Judicature Act* or a provincial court under the authority of the *Provincial Court Act* or the *Family Court Act*,

14

- (ia) notwithstanding clause (i), the Labour Board must still comply with section 13 when conducting a vote under the *Trade Union Act*.
- (j) Emergency Medical Care Incorporated;
- (k) persons providing, servicing or repairing medical equipment, such as wheelchairs, red cross beds/equipment, home oxygen equipment;
- (I) private not-for-profit community transportation providers;
- (m) food production plants; and
- (n) fishing vessels.
- 16.1 Physical distancing requirements and the person limit set out in sections 13, 13.1 and 13.2 do not apply to the following municipal entities and their contractors when carrying out their work duties:
 - (a) Police and Fire Services;
 - (b) Municipal Utilities such as water, wastewater and stormwater;
 - (c) Maintenance of utilities and municipal facilities;
 - (d) Transportation;
 - (e) Road maintenance/repair;
 - (f) Municipal ICT systems and services;
 - (g) Public Transit;
 - (h) Solid Waste, garbage and litter collection and disposal;
 - (i) Urban Forestry; and
 - (j) Municipal logistic, distribution, storage, inventory and repair services.
- 16.2 Physical distancing requirements and the person limit rule set out in sections 13, 13.1 and 13.2 do not apply to the following provincial entities and their contractors when carrying out their work duties:
 - (a) Transportation and Active Transit;
 - (b) Infrastructure and Housing;
 - (c) Road maintenance/repair;

- (d) Government building construction and/or repair; and
- (e) Enforcement or compliance officers authorized by their statutory appointments or delegated authority to inspect, investigate and/or enforce provincial legislation while carrying out their powers pursuant to the relevant statutory authority.
- 17. Recognized businesses and organizations:
 - (a) may host day camps, indoors or outdoors, while adhering to the following:
 - (i) up to a maximum of 30 persons per group excluding staff and volunteers without distancing, or in multiple groups of 30 persons with physical distancing between groups; and
 - (ii) must follow the COVID-19 Return to Day Camp Guidelines established by the Chief Medical Officer of health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Return-to-Day-Camp-Guidelines.pdf,
 - (b) may host overnight camps, while adhering to the following:
 - (i) up to a maximum of 15 persons per group excluding staff and volunteers without distancing, or in multiple groups of 15 persons with physical distancing between groups; and
 - (ii) must follow the COVID-19 Return to Overnight Camp Guidelines established by the Chief Medical Officer of Health, located at: covid-19-overnight-camping-guidelines-June14.pdf (novascotia.ca).
- 18. For the purpose of sections 18.2 -18.7, a "mask" means a commercial medical or non-medical mask or a home-made mask made as per the PHAC instructions located at: https://www.canada.ca/en/public-health/services/diseases/2019-novel-coronavirus-infection/prevention-risks/sew-no-sew-instructions-non-medical-masks-face-coverings.html, that covers the nose and mouth.
- 18.1 For the purpose of section 18.2, a "public place" means the part of the following places accessible to the public,

insofar as it is enclosed:

(a) a retail business, a shopping centre, or a building or room of a business where personal care services are provided;

- (b) a restaurant or a liquor licensed establishment, including the kitchen and preparatory space of a restaurant or a liquor licensed establishment;
- (c) a place of worship or faith gathering;
- (d) a place where activities or services of a cultural or entertainment nature are offered:
- (e) a place where sports are played, fitness, recreational or leisure activities are
- (f) a rental hall or other place used to hold events, including conventions and conferences, or to hold receptions;
- (g) a place where municipal or provincial government services are available to the public;
- (h) a common area, including an elevator, of a tourist accommodation establishment:
- (i) a lobby, reception area or elevator in an office building;
- (j) a common area or public space on a university or college campus;
- (k) a train or bus station, a ferry terminal, or an airport;
- (m) common areas of a multi-residential building;
- (n) all common areas of private indoor workplaces;
- (o) private indoor workplaces where there are:
 - (i) interactions with the public;
 - (ii) areas with poor ventilation; or
 - (iii) areas where physical distance as set out in section 13 cannot be maintained,

and insofar as it is outdoors

- (p) all serviced areas of a restaurant or a liquor licensed establishment, including their patios but excluding holes on the golf course that are licensed; and
- (q) organized outdoors gatherings such as special or temporary events where 2 metres (6 feet) physical distancing cannot be consistently maintained.

- 18.2 All persons must wear a mask that covers their nose and mouth while present in a
- 18.3 Notwithstanding section 18.2, a person is exempt from the requirement to wear a mask in a public place if the person:
 - (a) is less than 2 years of age or age 2 to 4 years and their caregiver cannot persuade them to wear a mask;
 - (b) for whom the wearing of a mask is not possible because of the person's medical condition;
 - (c) is reasonably accommodated by not wearing a mask in accordance with the Nova Scotia Human Rights Act;
 - (d) is in the public place receiving care or being provided a service or while participating in a physical or other activity requiring the mask be removed, in which case the person may remove the mask for the duration of the care, service or activity:
 - (e) removes the mask momentarily for identification or ceremonial purposes;
 - (f) is in a courtroom, jury room or secured area in a courthouse, or room where a proceeding or meeting of an administrative tribunal established by legislation is being held;
 - (g) is an officiant in the course of performing activities requiring vocalization such as talkling at the event or activity as described in sections 15, 15.1, 15.2 or 15.3;
 - (h) is 12 years of age or less, attending a day camp, overnight camp, or childcare centre
- 18.4 All persons must wear a mask that covers their nose and mouth as defined in section 18 while travelling on vehicles providing transportation to the public, including:
 - (a) any municipally operated public transit, including municipally operated buses and ferries;
 - (b) any public passenger vehicle licensed under the Motor Carrier Act, including community transit vehicles, commercial vehicles (shuttle vans), and vehicles providing charters and/or tours;
 - (c) any school buses licensed under the *Motor Carrier Act* and any vehicles of any capacity operated by private schools recognized by the Minister of Education and Early Childhood Development;

18

- (d) commuter vehicles and courtesy vehicles as defined under the *Motor Carrier Act*, vans, mini-buses, or buses of any passenger capacity providing services to the public;
- (e) any vehicles serving residents and staff of facilities listed in section 20.1; and
- (f) taxicabs regulated by municipalities under the authority of the *Motor Vehicle Act*.
- 18.5 Notwithstanding section 18.4, the following persons are exempt from the requirement to wear a mask while travelling on vehicles providing transportation to the public:
 - (a) a person is less than 2 years of age or age 2 to 4 years and their caregiver cannot persuade them to wear a mask;
 - (b) a person for whom the wearing of a mask is not possible because of the person's medical condition; and
 - (c) a person who is reasonably accommodated by not wearing a mask in accordance with the *Nova Scotia Human Rights Act*.
- 18.6 A person may remove the mask momentarily for identification purposes when boarding any public transit set out in section 18.4.
- 18.7 For greater certainty, the requirement to wear a mask, as set in sections 18.2 and 18.4, is the minimum standard that persons and businesses must adhere to, and where the business's plan approved in accordance with section 26.1 of this Order imposes a greater standard, then that standard applies.
- 19. Childcare facilities and family childcare homes regulated under the *Early Learning* and *Child Care Act*:
 - (a) may remain open; and
 - (b) directors, caregivers, staff and visitors must wear a mask as defined in section 18, unless subject to one of the exemptions as set out in section 18.3.

PART III LONG TERM CARE FACILITIES AND OTHER VULNERABLE POPULATIONS

- Effective 8:00a.m. July 14, 2021, except where otherwise stated in this Order, the
 restrictions on long term care facilities and other vulnerable populations apply to all
 persons present and residing in Nova Scotia.
- 20.1 Notwithstanding section 16(c) and subject to section 20.2, all for-profit or not-for-profit Department of Health and Wellness funded long-term care facilities licensed under the Homes for Special Care Act must comply with Schedule "A", "COVID-19 Management Long term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at:

 $\underline{https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf;}$

20.2 All long term care facilities licenced by the Department of Health and Wellness under the *Homes for Special Care Act*, and all residents of such facilities must comply with the process for isolating COVID-19 positive long term care residents set out in Schedule "A", "COVID-19 Management Long Term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at:

https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf.

- 20.3 All long term care facilities licensed by the Department of Health and Wellness under the *Homes for Special Care Act* are open to family for visits, contracted service providers performing non-essential long term care work, 2 designated caregivers and volunteers, in accordance with the terms and conditions set out in Schedule "A", "COVID-19 Management Long term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at: https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf.
- 20.4 All residents in homes licensed by the Minister of Community Services under the Homes for Special Care Act may have visitors so long as approved by the home in which they reside and comply with all general public health measures required under this Order.
- 20.5 All residents in homes licensed by the Minister of Health and Wellness are permitted to have community access in accordance with the terms and conditions set out in Schedule "A", "COVID-19 Management Long term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at:

https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf.

- 20.6 All residents in homes licensed by the Minister of Community Services under the Homes for Special Care Act are permitted to have community access as long as they comply with all general public health measures required under this Order.
- 20.7 For the purpose of section 20.8, "Adult Day Program" means a planned program of activities in a professional care setting designed for older adults who require supervised care during the day, or those who are isolated and lonely. Adult day care centers enable seniors to socialize and enjoy planned activities in a group setting, while still receiving needed health services. At the same time, they offer family caregivers respite from caregiving duties while knowing that their loved one is in a safe place.
- 20.8 All senior's Adult Day Programs are open.
- 20.9 For greater clarity, nothing in this Order prevents the:
 - (a) discharge of a COVID-19 patient from a hospital to a long-term care or residential care facility;
 - (b) transfer of a COVID-19 patient from community to a long-term care or residential care facility; or
 - (c) return of a COVID-19 patient who has left a long-term care or residential care facility for healthcare services back to that facility after receiving treatment at a hospital.

PART IV BUSINESSES. ORGANIZATIONS AND PROFESSIONS

- 21A Effective 8:00a.m. July 14, 2021, except where otherwise stated in this Order, the restrictions on businesses, organizations and professions apply to all established businesses, organizations and professions carrying on business and operating in Nova Scotia.
- 21.1The Occupational Health and Safety Act, 1996, c.7, s.1, is hereby incorporated by reference and must be followed by all employers, contractors, constructors, suppliers, owners and employees and each shall take every precaution that is reasonable in the circumstances to ensure the health and safety of all persons at or near a workplace.
- 22. All not-for-profit and for-profit businesses and organizations operating or carrying on business in Nova Scotia may not carry out COVID-19 Point of Care Screening Tests (PCTs) unless the business or organization:

- (a) has obtained prior approval from the Office of the Chief Medical Officer of Health: and
- (b) complies with any direction issued by me, as Chief Medical Officer of Health.
- 23. Subject only to specific closures or limitations directed elsewhere in this Order, all not-for-profit and for-profit businesses and organizations operating or carrying on business in Nova Scotia may continue to operate but must implement physical distancing of 2 metres (6 feet) within all workplaces and meeting spaces, except where otherwise provided in this Order.
- 23.1 Any not-for-profit or for-profit business or organization carrying on business in Nova Scotia that cannot maintain the physical distancing requirements set out in section 13 of this Order must limit the number of customers or clients on its premises to no more than 25 persons at a time.
- 24. Restaurants and liquor licensed establishments may provide in-person dining while adhering to the following:
 - (a) all restaurants may offer in-person dining to a maximum of full legal capacity of the restaurant provided the operator:
 - (i) ensures no group of patrons shall have more than 25 persons at a table:
 - (ii) maintains a minimum physical distance of 2 metres (6 feet) between seats or a physical barrier between tables, booths and single seats;
 - (iii) ensures food is delivered to the table by staff
 - (iv) ensures patrons remain seated at their table, except while obtaining bar service, waiting to be seated, going to the bathroom, and getting ready to leave;
 - (v) ensures patrons wear masks when not consuming food or alcohol; and
 - (vi) ensures staff (servers and kitchen) wear masks while carrying out their work duties,

and

- (b) all liquor licensed establishments, including liquor manufacturers (as defined in the *Liquor Control Act*) may offer in-person dining to a maximum of full legal capacity of the restaurant provided the operator provided the licensee:
 - (i) ensures no group of patrons have more than 25 persons per table.

21

- (ii) maintains a minimum physical distance of 2 metres (6 feet) between seats or a physical barrier between tables, booths and single seats;
- (iii) provides food and alcohol consumption only for in-seat service;
- (iv) ensures food is delivered to the table by staff;
- (v) ensures food and alcohol remains at the table;
- (vi) ensures patrons remain seated at their table, except while obtaining bar service, waiting to be seated, going to the bathroom, participating in a game/activity being held in the establishment and getting ready to leave;
- (vii) ensures patrons comply with the physical distancing requirements set out in section 13;
- (viii) ensures patrons wear masks when not consuming food or alcohol; and
- (ix) ensures staff (servers and kitchen) wear masks while carrying out their work duties
- 24.1 Bingos that hold a bingo license must adhere to the same provisions as set out in section 24(b) for liquor licensed establishments and liquor manufacturers, and the Communities, Culture and Heritage Re-opening Guide for Bingo Operations, located at: https://novascotia.ca/coronavirus/docs/Bingo-operations-reopening-quidelines.pdf.
- 24.2 Liquor boutiques, distilleries, breweries, and wineries:
 - (a) may operate at the maximum allowable occupancy of their retail space for purchase and sale for offsite consumption, while adhering to public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements;
 - (b) may offer in-person tastings and samplings in accordance with their approved Workplace COVID-19 Prevention Plan set out in section 26; and
 - (c) may offer in-person dining in accordance with section 24(a)
- $24.3\ Restaurants, liquor\ licensed\ establishments, liquor\ boutiques,\ distilleries,\ breweries,\ and\ wineries\ offering\ in-person\ dining\ must:$
 - (a) collect and maintain records of each patron contact information that includes the following minimum data:

- (i) name(s);
- (ii) contact telephone number;
- (iii) date; and
- (iv) time

that the patron was at the restaurant or liquor licensed establishment;

- (b) retain information collected under section 24.3 for a period of 30 calendar days and destroy it on the following day immediately thereafter;
- (c) provide the information collected under section 24.3 to the Chief Medical Officer of Health or his delegate in electronic form; and
- (d) only use the information collected under section 24.3 for the purpose of provision to the Chief Medical Officer of Health in compliance with this Order and may not use it for any other purpose.
- 24.4 Notwithstanding other provisions of this Order, restaurants, liquor licensed establishments, liquor boutiques, distilleries, breweries, and wineries:
 - (a) may have musicians (professional or amateur) perform and the musicians must adhere to the gathering limits for professional musicians set out in section 15.3, and the Covid-19 Guidance for Musicians (Vocalists and Instrumentalists) located at: https://novascotia.ca/coronavirus/docs/COVID-19-Guidance-for-Vocalists-and-Instrumentalists.pdf; and
 - (b) allow patrons to participate in dancing, while ensuring patrons' adherence to the public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements.
- 24.5 Owner/operators of liquor licensed establishments and non-liquor licensed establishments, including community centres, charities or organized clubs, may host activities such as darts, pool, cards or bowling but in doing so, must adhere to requirements established by the Chief Medical Officer of Health located at: https://novascotia.ca/coronavirus/docs/COVID-19-Games-and-activities-licensed-establishments.pdf.
- 25. Regulated and unregulated health professions practicing may continue to practice and provide services within their scope of practice if they have a Workplace COVID-19 Prevention Plan that has been approved by the Chief Medical Officer of Health.

23

- 26. All not for-profit, for-profit businesses, organizations and professions named in Part IV of this Order shall, as a requirement of their ongoing operations, develop and comply with a Workplace COVID-19 Prevention Plan, as amended from time to time.
- 26.1 In addition to the physical distancing requirements and person limit rule set out in sections 13 and 23.1, the Workplace COVID-19 Prevention Plan required by section 26 shall address the followino:
 - (a) how to work and interact with customers or clients;
 - (b) physical distancing in the workplace;
 - (c) cleaning;
 - (d) equipment;
 - (e) preparing employees to return to work;
 - (f) preparing for customers or clients; and
 - (g) monitoring and communicating of plan,

and may be substantially similar to those proposals listed in Schedule "B", located at: https://novascotia.ca/coronavirus/docs/Schedule-B-HPA-Order.pdf for similar businesses or health professionals and must be made available for review by the Chief Medical Officer of Health.

- 26.2 All businesses or regulated health professionals that are a member of an association listed in Schedule "B" must adopt as their Workplace COVID-19 Prevention Plan the proposal submitted by their association and must be made available for review by the Chief Medical Officer of Health.
- 26.3 For greater certainty, in the case of conflict between a Workplace COVID-19 Prevention Plan and this Order, the more stringent provision applies.
- 26.4 Effective 8:00a.m. July 14, 2021, all Provincial Park campgrounds may operate at 100% capacity while adhering to the public health requirements set out in this Order.
- 26.5 Effective 8:00a.m. July 14, 2021, all privately operated campgrounds may operate at 100% capacity while adhering to the public health requirements of this Order and must operate in accordance with the terms and conditions as approved by the Chief Medical Office of Health referenced in Schedule "B" located at: https://novascotia.ca/coronavirus/docs/Schedule-B-HPA-Order.pdf.
- 27. Casino Nova Scotia (Halifax and Sydney locations) and Video Lottery Terminals may operate at 100% capacity while adhering to their approved Workplace COVID-19

Prevention Plan in accordance with section 26, and those that hold a liquor license may operate at normal hours of business as per their license.

- 28. All not-for-profit and for-profit fitness facilities such as gyms and yoga studios, sport and recreational facilities such as arenas, tennis courts, and large multipurpose recreation facilities may open and:
 - (a) operate at maximum capacity that can be achieved while adhering to the public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements;
 - (b) offer outdoor fitness classes and other recreational activities up to a maximum capacity of 25 persons; and
 - (c) operate indoor and outdoor pools at maximum that can be achieved while adhering to the public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements.
- 28.1 Businesses and organizations that offer indoor recreation and leisure activities such as indoor play areas, arcades, climbing facilities, dance classes and music lessons may open and operate at 100% indoor capacity, while ensuring two metres (six feet) distance between participants and participants wear masks except during an activity that makes it difficult to wear a mask.
- 28.2 All not-for-profit and for-profit fitness facilities, recreation and leisure businesses, and organized clubs may host lessons for recreational activities or offer personal training, while adhering to the gathering limits set out in section 28.
- 28.3 Notwithstanding section 28 and 28.1, golf courses may operate at full capacity while ensuring players practice physical distancing between each group on the golf course.
- 29. All personal service establishments such as hair salons, barber shops, spas, nail salons and body art establishments may open for provision of personal services by appointment or drop-in, while operating within their approved Workplace COVID-19 Prevention Plan in accordance section 26 and adhering to the physical distancing requirements and person limit rule set out in this Order.
- 30. The Art Gallery of Nova Scotia and all museums and public libraries may open and operate at 100% capacity, while adhering to the physical distancing and other public health requirements set out in this Order.
- 31. For the purpose of sections 31.1 to 31.2, "retail business" means a business operating on an ongoing basis at a fixed location primarily selling goods or products for use or consumption by individual purchasers.
- 31.1 A retail business that engages in the retail sale, or rental, of items to the public

25

26

may operate at maximum capacity that can be achieved while adhering to the public health requirements set out in this Order, including but not limited to compliance with the following conditions:

- (a) in accordance with section 13, members of the public must be able to maintain a physical distance of at least 2 metres (6 feet) from every other person in the place of business, except those persons residing in the same household;
- (b) no patrons may be permitted to line up inside or line up or congregate outside of the establishment unless they are,
 - (i) maintaining a physical distance of at least 2 metres (6 feet) from other persons inside or outside the establishment; and
 - (ii) wearing a mask or face covering in a manner that covers their nose and mouth, unless they are entitled to any of the exceptions set out in section 18.3.

and

(c) the person responsible for the place of business must ensure that music is not played at the place of business at a decibel level that exceeds the level at which normal conversation is possible, and in any event no greater than 50 db.

PART V EXCEPTIONS

32. Under exceptional circumstances and under the authority granted to me as the Chief Medical Officer of Health under Part I of the *Health Protection Act*, I may exercise discretion to grant an exception to any term and condition of this Order.

PART VI PENALITIES

- 33. Any direction provided by a medical officer of health to a person, business, organization or other entity pertaining to COVID-19 and the terms and conditions of this Order must be followed.
- 34.Failure to comply with this health protection order may be considered a breach of this Order issued under the *Health Protection Act* and may result in penalties under the *Act*.

This Order remains in effect until notice is provided by myself, as Chief Medical Officer of Health, under the authority granted under Part I of the *Health Protection Act* and will be updated from time to time.

Signed

Dr. Robert Strang Chief Medical Officer of Health

War

Nova Scotia Department of Health and Wellness

cc The Honourable Iain Rankin, Premier of Nova Scotia
The Honourable Zach Churchill, Minister of Health and Wellness
Laura Lee Langley, Deputy to the Premier and Clerk of Executive Council
Dr. Kevin Orrell, Deputy Minister of Health and Wellness
Dr. Shelley Deeks, Deputy Chief Medical Officer of Health
Tina M. Hall, Legal Counsel, Nova Scotia Dept. of Justice

RESTATED ORDER #2 OF THE CHIEF MEDICAL OFFICER OF HEALTH UNDER SECTION 32 of the HEALTH PROTECTION ACT 2004, c. 4, s. 1.

July 16, 2021

Update Log:

July 16, 2021: Sections 18.1 and 24.2 (amended)
July 14, 2021: Sections 2.2, 3.5A, 13A, 13.2, 15-15.3, 17, 18.1, 18.3, 19-19.2, 20, 20.3, 21A, 23.1, 24, 24.2, 24.4, 26.4, 26.5, 27, 28, 28.1, 30, 31.1, and 31.2 (amended) and Section 15.10 (page)

July 7, 2021: Sections 3.1, 3.5A, 3.5, 7 and 18.3 (amended) and Section 7.2 (new) July 5, 2021: Sections 2, 2.2, 3.2, 3.5, 4-12.1, 13A, 20, and 21A (amended) June 30, 2021: Sections 2-2.3, 3.1-3.2, 3.5A, 3.5, 4-4.2, 5, 12.3-12.4, 13A, 13.2, 15-15.3, 15.6, 16-16.2, 17-17.1, 18.3, 20-20.4, 21A, 24, 24.2, 24.4, 27, 28-28.3, 29, 30 and 31.1 (amended)

June 24, 2021: Sections 1, 3.5 and 3.6 (amended) and Section 3.5A (new) June 23, 2021: Sections 1, 2, 2.2, 2.3, 3.1, 3.5, 4, 4.2, 12.2, 12.3, 15.6, 24.4 and 27 (amended) and Section 3.6 (new) June 16, 2021: Sections 2, 2.2, 2.3, 3.2, 13A, 13.2, 14, 15-15.6, 18.3, 20-20.9, 21A, 24,

June 16, 2021: Sections 2, 2.2, 2.3, 3.2, 13A, 13.2, 14, 15-15.6, 18.3, 20-20.9, 21A, 24 24.2-24.5, 27, 28-28.2, 29, 30, 31.1 and 31.2 (amended) and Sections 17.1, 24.1 and 28.3 (new)

20.3 (new) June 4, 2021: Sections 2.2, 13.2, 15.3, 15.3.1, 18.1 and 28.3 (amended) May 31, 2021; Sections 2, 2.2-2.3, 3.2, 13A, 13.2-13.5, 14, 15-15.3, 15.4.2, 18, 19-19.1, 20, 20.3-20.7, 21A, 23.1, 24.1-24.5, 26.4-26.5, 28, 28.2, 29, 31-31.2 (amended) and Sections 15, 31-15.3, (new)

Zot, 20.3-20.7, 24.7, 25.1, 24.7-24.3, 20.4-20.3, 20.2, 29, 31-31.2 (alinehede) and Sections 15.3.1-15.3.2 (new)

May 21, 2021: Sections 2-2.3, 10-10.1, 12.1, 12.3, 13A, 18.3, 18.7, 19, 20, 21A and 31-31.2.1 (amended) and Sections 15.3.1., 26.4-26.5 and 28.3 (new)

May 13, 2021: Sections 2-2.3, 3.2, 7-7.1, 13A, 13.2, 20 and 21A (amended) and Sections 13.5-13.6 (new)

May 8, 2021: Sections 2-2.3, 3.1-3.2, 4-4.2, 10, 11, 12.1, 12.3-12.4, 13A, 19-19.2, 20, 21A, 22 and 31.1 (amened), and Sections 13.4 and 31.1.2 (new) May 4, 3021: Sections 2.2, 28.2 and 31.1 (amended) and Section 13.3 (new) April 29, 2021: Sections 24.3 and 31.2 (amended) and Section 31.2.1 (new)

- TO: 1.) All persons residing in or present in the Province of Nova Scotia;
 - 2.) All not-for-profit and for-profit businesses and organizations operating or carrying on business in Nova Scotia;
 - Such other persons or entities as may be identified by the Chief Medical Officer of Health or otherwise as set out in this Order.

ORDER made pursuant to Section 32 of the Health Protection Act (Nova Scotia)

- (a) "self-isolate" means the requirement of any person who has COVID-19 to remain separate from others in such places and under such conditions to prevent or limit the direct or indirect transmission of COVID-19;
- (b) "self-quarantine" means the requirement of any person who has been exposed or may have been exposed to COVID-19 during its period of communicability to restrict that person's activities in order to prevent disease transmission during the incubation period for this disease.
- 2. Effective 11:59p.m. July 5, 2021, all persons are prohibited from entering Nova Scotia, except as stated herein or set out elsewhere in this Order:
 - (a) permanent residents of Nova Scotia, Newfoundland and Labrador, Prince Edward Island, New Brunswick ("the Atlantic Provinces"), and their spouses, partners and dependant children when travelling together;
 - (b) persons travelling to Nova Scotia from a province or territory outside the Atlantic Provinces:
 - (c) persons travelling to Nova Scotia from outside Canada who are eligible to enter Canada in accordance with the Federal Order in Council 2021-0313, as
- 2.1 Where any person travels into Nova Scotia in contravention of this Order, a peace officer is hereby authorized and directed to return that person to an interprovincial border and require the person to leave the Province immediately or at such a time as may be directed, and in doing so, may exercise authority under section 46 of the Health Protection 4ct
- 2.2 Effective 11:59p.m. July 5, 2021, all persons seeking entry into Nova Scotia must:
 - (a) apply through the Nova Scotia Safe Check-in form and complete daily digital check-ins, located at: https://travel-declaration.novascotia.ca/en, except:
 - (i) persons travelling to Nova Scotia from Newfoundland and Labrador, or Prince Edward Island and New Brunswick;
 - (ii) professional truck drivers;
 - (iii) persons who have received compassionate exceptions; and
 - (iv) essential healthcare workers who travel to and from Nova Scotia and another province or territory to carry out their work duties on an ongoing regular basis or on a locum basis.

WHEREAS Section 32 of the Health Protection Act states:

- **32 (1)** Where a medical officer is of the opinion, upon reasonable and probable grounds, that:
 - (a) a communicable disease exists or may exist or that there is an immediate risk of an outbreak of a communicable disease;
 - (b) the communicable disease presents a risk to the public health; and
 - (c) the requirements specified in the order are necessary in order to decrease or eliminate the risk to the public health presented by the communicable disease,

the medical officer may by written order require a person to take or to refrain from taking any action that is specified in the order in respect of a communicable disease.

WHEREAS COVID-19 has been identified as a communicable disease that presents a risk to public health as defined under s.4(b) of the Health Protection Act, and;

WHEREAS I am the Chief Medical Officer of Health for the Province of Nova Scotia and am of the opinion, upon reasonable and probable grounds, that

- (a) a communicable disease (COVID-19) exists; and that there is an immediate risk of an outbreak of the communicable disease;
- (b) the communicable disease presents a risk to the public health; and
- (c) the requirements specified in the order are necessary to decrease or eliminate the risk to the public health presented by the communicable disease, and:

WHEREAS as the Chief Medical Officer of Health, I have determined it necessary to issue this Order to the Class of Persons to decrease the risk to public health presented by COVID-19.

Please be advised that:

I, Dr. Robert Strang, Chief Medical Officer of Health, order the following:

PART I ENTRY, ISOLATION AND QUARANTINE REQUIREMENTS

In this Order

2

- (b) disclose to the Chief Medical Officer of Health or his delegate information related to their COVID-19 vaccination, including:
 - (i) whether they received a COVID-19 vaccine;
 - (ii) the brand name or other information that identifies the vaccine that was administered;
 - (iii) the dose received, and
- (c) if they are fully vaccinated, provide the Chief Medical Officer of Health or his delegate evidence of COVID-19 vaccination.
- 2.3 For greater clarity, the evidence of COVID-19 vaccination must be a document issued by the government or non-governmental entity that administered the COVID-19 vaccine and must contain the following:
 - (a) the name of the person who received the vaccine;
 - (b) the name of the government or the name and civic address of the non-governmental entity that administered the vaccine;
 - (c) the brand name or any other information that identifies the vaccine that was administered; and
 - (d) the dates on which the vaccine was administered.
- $3.1\ {\rm The}$ requirements of section 3.2 apply to all persons residing in or present in the Province of Nova Scotia who:
 - (a) have travelled outside the Atlantic Provinces; or
 - (b) are identified as a close contact of a person who has or has been diagnosed with COVID-19;
 - (c) are identified as a person diagnosed with COVID-19; or
 - (d) have been tested for COVID-19 due to the presence of symptoms or as directed by public health and are awaiting the results of their test.
- 3.2 Effective 11:59p.m. July 5, 2021, those persons listed in section 3.1 must:
 - (a) self-quarantine or self-isolate, for:

- (i) the period commencing on the day of entry into Nova Scotia, and continuing thereafter for 14 consecutive days or as directed by a Medical Officer of Health, or
- (ii) the first day of close contact, or first day of symptoms, testing, diagnosis, and continuing thereafter for 14 consecutive days or as directed by a Medical Officer of Health.
- (b) During the 14-day period, conduct themselves in such a manner as not to in any way expose any other person to infection or potential infection from COVID-19, and follow all infection control instructions given to them on the Government of Nova Scotia's website, at: https://novascotia.ca/coronavirus/, or given to them by Telehealth 811 staff, public health staff or any other staff of a healthcare facility to which they may seek or receive treatment.
- (c) After the 14-day period in section 3.2(a.) has lapsed, they may cease self-isolation or self-quarantine if they do not exhibit symptoms of COVID-19.
- 3.3 All persons are encouraged to complete the online assessment tool should they exhibit symptoms of COVID-19.
- 3.4 For greater certainty, persons required to self-isolate or self-quarantine in accordance with section 3.2 must remain in their residence or residence grounds and otherwise remove themselves from the presence of others in public while they may be infectious during the 14-day period, so that all precautions necessary to protect others are in place. Specifically, such persons must not enter any buildings, public transportation, or other enclosed spaces (other than their residence) where other people are present.
- 3.5A For the purpose of sections 2.2, 2.3 and 3.5:
 - (a) "fully vaccinated" means receipt of 1 dose of a vaccine authorized as a 1 dose vaccine series such as Janssen plus 14 days, or 2 doses of a vaccine authorized as a 2 dose vaccine series such as Pfizer, Moderna or AstraZeneca plus 14 days;
 - (b) "partially vaccinated" means receipt of 1 dose of a vaccine authorized as a 2 dose vaccine series such as Pfizer, Moderna or AstraZeneca plus 14 days; and
 - (c) "vaccine" means a vaccine against COVID-19 that has been approved by either the Public Health Agency of Canada or the World Health Organization.
- 3.5 Notwithstanding section 3.2(a):
 - (a) persons who receive a negative COVID-19 test result after being referred for testing by the online assessment tool due to the presence of COVID-like symptoms may cease self-quarantine on the date of receipt of the negative test result:

- (b) persons who are required to self-isolate or self-quarantine may leave their
 - (i) to undergo COVID-19 testing as directed by a Medical Officer of Health;
 - (ii) for 1 outing per day for outdoor exercise within walking/running distance of their home or isolation site for a maximum of 1 hour,
- (c) persons travelling from a province or territory outside the Atlantic Provinces to Nova Scotia, must adhere to the following:
 - (i) persons who are fully vaccinated before arriving in Nova Scotia are not required to self-quarantine on arrival in Nova Scotia;
 - (ii) persons who are partially vaccinated before arriving in Nova Scotia must self-quarantine for at least 7 days on arrival in Nova Scotia and must not cease self-quarantine until they receive 1 negative test result while in Nova Scotia, with testing on day 5 or 6;
 - (iii) persons who have had COVID-19, have recovered from it in the last 12 weeks before entering Nova Scotia, and are partially vaccinated must self-quarantine for 7 days on arrival into Nova Scotia but are not required to be tested: and
 - (iv) persons who have not received any vaccine before arriving in Nova Scotia must self-quarantine for 14 days in accordance with section 3.2(a),
- (d) persons travelling to Nova Scotia from outside Canada who eligible to enter Canada in accordance with the Federal Order in Council 2021-0313, as amended:
 - (i) who are fully vaccinated (the "fully vaccinated traveller" as set out in the Federal Order in Council 2021-0615) are not required to self-quarantine on arrival in Nova Scotia; and
 - (ii) who are not fully vaccinated are required to self-quarantine on arrival in Nova Scotia in accordance with Federal Quarantine requirements.
- 4. Notwithstanding sections 2, 3.1 and 3.5, persons who are well and showing no symptoms of COVID-19 may enter Nova Scotia for the purposes of facilitating child sharing between parents under an order or agreement providing for joint custody, and in such instances, both the persons facilitating custody and children showing no symptoms of COVID-19 are exempt from the requirement to self-quarantine.

6

- Further to section 4, parents and children entering or leaving Nova Scotia for the purposes of facilitating child sharing arrangements must adhere to the self-quarantine requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-child-custody.pdf.
- 6. Notwithstanding sections 2, 3.1 and 3.5, individuals who are well and showing no symptoms of COVID-19 and are engaged in a legal proceeding in Nova Scotia, whether the accused, victim, witness, party or lawyer in such proceeding, may enter Nova Scotia for participation in the legal proceeding and are exempt from self-quarantine if the person adheres to the COVID-19 Protocol for Exempt Travelers established by the Chief Medical Officer of Health, located at:

https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-exempt-travellers-en.pdf.

- 7. An employer or contractor of any Temporary Foreign Worker entitled to enter Nova Scotia pursuant to the Federal Orders in Council 2021-0614 and 2021-0615, as amended, must first, before the Temporary Foreign Worker enters Nova Scotia, satisfy me, as Chief Medical Officer of Health, that the employer or contractor has made adequate provision for compliance with the Federal Quarantine requirements applicable to the Temporary Foreign Workers in the Agriculture and Seafood Sector.
- 7.1 In addition, the employer or contractor and the Temporary Foreign Worker must, for the duration of the entire work period in Nova Scotia:
 - (a) adhere to all applicable terms and conditions of this Order; and
 - (b) adhere to the COVID-19 Protocol for Temporary Foreign Workers employed in Agriculture and Seafood Sectors established by the Chief Medical Officer of Health, located at: https://novascolia.ca/coronavirus/docs/COVID-19- https://novascolia.ca/coronavirus/docs/COVID-19- Protocolfortemporaryforeignworkers-agriculture-and-seafood-sectors-en.pdf.
 - (c) comply with any direction issued by me, as Chief Medical Officer of Health, or a Medical Officer of Health with respect to the Temporary Foreign Worker and their employment in Nova Scotia.
- 7.2 For greater certainty, a Temporary Foreign Worker in the Agriculture and Seafood Sector, on arrival in Nova Scotia, is required to quarantine in accordance with the Federal Quarantine requirements unless they are assessed as exempt from quarantine in accordance with Federal Order in Council 2021-0615.
- 8. Notwithstanding sections 2, 3.1 and 3.2, persons who are required to travel to Nova Scotia or outside Nova Scotia for essential health services, with accompanying support persons as permitted by health authority policy, are exempt from the requirement to self-quarantine but must adhere to the COVID-19 Protocol for Exempt Travelers established by the Chief Medical Officer of Health, located at:
- https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-exempt-travellers-en.pdf

- 9. For the purpose of section 9.1, "rotational worker" means a person who lives in Nova Scotia and travels to work in another province or territory in Canada on a regular schedule or travels outside Canada on a regular schedule and is exempt from self-quarantine under the federal Quarantine Act.
- 9.1 Notwithstanding sections 2, 3.1 and 3.5, rotational workers must adhere to the selfquarantine and COVID-19 testing requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Directive-on-Exceptions-for-Rotational-Workers.pdf.
- 10. Notwithstanding sections 2, 3.1 and 3.5, specialized workers who are:
 - (a) needed for urgent work on critical infrastructure that cannot be done by individuals from within the Province and such work is crucial to the functioning of the Province; or
 - (b) needed for urgent work that cannot be done by individuals from within the Province and are necessary to preserve the viability of one or more Nova Scotia businesses,

are permitted to enter Nova Scotia from any geographic location outside Nova Scotia. A list of critical infrastructure included in this definition is available at: https://www.publicsafety.gc.ca/cnt/ntnl-scrt/crtcl-nfrstrctr/esf-sfe-en.aspx.

- 10.1 Specialized workers permitted to enter Nova Scotia in accordance with section 10 must adhere to self-quarantine and COVID-19 testing requirements established by the Chief Medical Officer of Health, located at:
- https://novascotia.ca/coronavirus/docs/COVID-19-Directive-on-Exceptions-for-Specialized-Workers.pdf.
- 11. Notwithstanding sections 2, 3.1 and 3.5, fish harvesters required to enter Nova Scotia from any geographic location outside Nova Scotia to perform the commercial or licensed activity of catching fish and other seafood for market or other approved activities, excluding recreational fishing, must adhere to self-quarantine requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-fish-harvesters-en.pdf.
- 12.1 Notwithstanding section 3.1 and 3.5, workers who are essential to the movement of people and goods, and who must enter Nova Scotia as part of their duty requirements, are exempt from the requirement to self-quarantine, particularly:
 - (a) healthy workers in the trade and transportation sector who are employed in the movement of goods and people across the Nova Scotia border by land, air, or water, including truck drivers, crew, maintenance and operational workers on any plane, train or ship;
 - (b) Canadian Military and Defence Team personnel, Coast Guard, RCMP, Canadian Border Services Agency, and Canadian Security Intelligence Service;

- (c) first responders, including police, fire, EHS paramedic workers; and
- (d) essential healthcare workers who travel to and from Nova Scotia and another province or territory to carry out their work duties on an ongoing regular basis or on a locum basis

but must adhere to the COVID-19 Protocol for Exempt Travelers established by the Chief Medical Officer of Health, located at:

https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-exempt-travellers-en.pdf

- 12.2 For greater clarity, airline crew employed in the movement of people and goods and who are required to travel to Nova Scotia or from Nova Scotia to carry out their work duties are exempt from the self-quarantine requirements set out in sections 3.2 and 3.5.
- 12.3 Persons exempt under sections 12.1 to 12.2 must practice physical distancing of 2 metres (6 feet) to the best of their ability, follow all public health recommendations, closely self-monitor and must self-isolate or self-quarantine should they exhibit any COVID-19 symptoms as set out in the online assessment tool.

PART II PHYSICAL DISTANCING, GATHERING LIMITS, MASKS AND FACE COVERINGS

- 13A Effective 8:00a.m. July 14, 2021, except where otherwise stated in this Order the requirements for physical distancing, gathering limits, masks and face coverings apply to all persons present and residing in Nova Scotia.
- 13. All persons present and residing in Nova Scotia must maintain physical distancing of 2 metres (6 feet).
- 13.1 All persons present and residing in Nova Scotia must not participate in any gatherings, whether indoors or outdoors, unless subject to a specific exception set out in this Order.
- 13.2 Notwithstanding sections 13 and section 13.1, persons may gather together for informal gatherings with their immediate family members residing in the same household and close social contacts up to a maximum of 25 persons indoors and 50 persons outdoors, and they are not required to practice physical distancing or masking, unless they are in a public place where masks are required.
- $13.3\ Notwith standing\ sections\ 13.1\ and\ 13.2\ and\ for\ greater\ certainty,\ parties\ to\ a\ child\ sharing\ arrangement,\ or\ an\ order\ or\ agreement\ providing\ for\ joint\ custody:$
 - (a) may facilitate and participate in such child sharing or custody arrangement between households.

9

- (g) funerals, visitations and receptions;
- (h) faith gatherings and drive-in or parking lot faith service; and
- (i) drive-in movie theatre,

while adhering to the following:

- (a) must adhere to the informal gathering limits set out in section 13.2 when not hosted by a recognized business or organization (i.e. event planner, event venue):
- (b) practice physical distancing of 2 metres (6 feet), except persons gathered informally in accordance with section 13.2;
- (c) practice masking requirements set out in section 18.2 unless exempt under section 18.3; and
- (d) if attending a drive-in or parking lot faith service, or drive-in movie theatre, all persons in attendance must adhere to the following:
 - (i) the service or movie is conducted over speakers or by remote radio
 - (ii) there is no contact between cars and no transfer of things between cars:
 - (iii) participants remain in their respective vehicle while "attending" the service, unless accessing a washroom facility or concession stand and while doing so, must adhering to protocol for gathering limits, physical distancing, and masking requirements set out in this Order;
 - (iv) participants practice physical distancing among vehicles, such that vehicles must be at least 2 metres (6 feet) apart from each other; and
 - (v) participants in the same car adhere to the gathering limits of section 13.2.
- 15.1 Notwithstanding sections 15, where any of the activities listed in section 15 are hosted by an existing business or organization legally operating in Nova Scotia, then the business or organization:
 - (a) may host up to 50% of legal capacity of the indoor establishment in which the event or activity is held, to an overall maximum of 150 persons, or up to a maximum of 250 persons if the event or activity is outdoors;

but

- (b) must adhere to the self-quarantine requirements established by the Chief Medical Officer of Health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Protocol-for-child-custody.pdf if a parent or child develops symptoms or tests positive for COVID-19.
- 13.4 For the purpose of section 13.5, an "illegal public gathering" is defined as a gathering that does not comply with the requirements of this Order, including:
 - (a) the attendance limits applicable to gatherings, whether indoors or outdoors:
 - (b) physical distancing requirements; and
 - (c) masking requirements.
- 13.5 For greater certainty, persons are prohibited from:
 - (a) organizing an in-person gathering, including requesting, inciting, or inviting others to attend an illegal public gathering;
 - (b) promoting an illegal public gathering via social media or otherwise; or
 - (c) attending an illegal public gathering of any nature, whether indoors or
- 14. Notwithstanding section 13.1, persons may engage in outdoor physical activity while adhering to the gathering limitations set out in section 13.2 or the gathering limitations while participating in an outdoor fitness class and other recreational or leisure activities as set out in sections 28 and 28.1, and in doing so, must adhere to physical distancing requirements of 2 metres (6 feet) as set out in section 13.
- 15. Persons may engage in the following events and activities, whether indoors or
 - (a) social events:
 - (b) arts and culture events;
 - (c) sports, recreation and physical activity events;
 - (d) special events;
 - (e) festivals;
 - (f) weddings and receptions;

10

- (b) must ensure all persons in attendance practice physical distancing and masking requirements set out in this Order;
- (c) must have a COVID-19 Prevention Plan;
- (d) may submit specific organizational plans to nseconomy@novascotia.ca for large venues such as Scotiabank Centre, Halifax Exhibition Centre, Halifax Convention Centre, Centre 200 and Speedworld. Plans will be reviewed by CCH and LAE staff and approved by the Chief Medical Officer of Health.
- 15.1.1 Business and organizations may host events listed in section 15 virtually while adhering to the following:
 - (a) the capacity limits set out in section 15.1(a) to support the delivery of the event; and
 - (b) their COVID-19 Workplace Prevention Plan to carry out virtual production in accordance with section 26
- 15.1.2 For greater clarity, persons may participate in dancing when attending an event listed in section 15 but must adhere to the applicable public health requirements when the event is hosted by a person informally, or hosted by a recognized business or organization.
- 15.2 Notwithstanding section 15, players, participants, and officials
 - (a) may engage in organized sport practice and training, games and regular league play recreational, amateur, and professional), up to a maximum of 25 persons indoors and 50 persons outdoors without physical distancing, or in multiple groups of 25 persons indoors or 50 persons outdoors with physical distancing between groups;
 - (b) where competition is not possible with the number restrictions stated in (a), Sport Nova Scotia, in consultation with CCH, may, through an application process, provide approval to the Provincial Sport Organization, for additional numbers in the field of play. In addition, registered organizations may host competitions as part of an event while adhering to the standard sport participation numbers on the field;
 - (c) may engage in organized arts and culture (amateur) rehearsals and performances, up to a maximum of 25 persons indoors and 50 persons outdoors without physical distancing, or in multiple groups of 25 persons indoors or 50 persons outdoors with physical distancing between groups;
 - (d) are recommended to wear a mask

but

- (e) may only engage in tournaments when affiliated with or hosted by a Provincial Sport Organization following their respective Return to Sport Plan; and
- (f) spectators are permitted with a limit of 50% of indoor capacity of the facility up to a maximum of 150 spectators or up 250 spectators outdoors, when the organized sport practice, training and competition (recreational, amateur, and professional), or the organized arts and culture rehearsals (amateur) is hosted by a recognized business or organization. Plans for hosting multiple groups, each group meeting the gathering limits, are permitted if following the guidance outlined by CCH.
- 15.3 Notwithstanding sections 15, 15.1 and 15.2, professional artists, musicians, dancers and actors may engage in rehearsal and performances up to a maximum of 25 persons indoors and 50 persons outdoors, without physical distancing while adhering to the following:
 - (a) the masking and spectator requirements set out in sections 15.2(d) and (f);
 - (b) an appropriate COVID-19 Workplace Prevention Plan in accordance with section 26: and
 - (c) the Covid-19 Guidance for Musicians (Vocalists and Instrumentalists) located at: https://novascotia.ca/coronavirus/docs/COVID-19-Guidance-for-Vocalists-and-Instrumentalists.ddf.
- 15.4 Any person, business or organization that hosts an event authorized by sections 15 to 15.3 is responsible for maintaining oversight of the gathering and for ensuring that all persons in attendance comply with the requirements of this Order.
- 15.5 For greater clarity, the maximum number of persons requirement set out in section 15.1 applies to meetings and training hosted by, including but not limited to:
 - (a) provincial and municipal governments;
 - (b) private businesses and organizations;
 - (c) first responder organizations;
 - (d) organized clubs; and
 - (e) mental health and addictions support group meetings,

all persons in attendance must adhere to the masking requirements set out in section 18.2 unless exempt under section 18.3.

- 15.6 Notwithstanding sections 13 and 15.5, any person, volunteer or employee who provides emergency first response are not required to adhere to the physical distancing requirement or gathering limits during an emergency, when carrying out duties or when necessary during training.
- 16. For greater clarity, physical distancing and the person limit rule set out in sections 13, 13.1 and 13.2 do not apply to the following organizations, activities, persons, or places:
 - (a) hospitals as defined in the *Hospitals Act* and a health authority as defined in the *Health Authorities Act*;
 - (b) profit, not-for-profit or government operated Department of Community Services funded organizations or representatives that are covered under the Homes for Special Care Act and the Children and Family Services Act including places of safety for children and youth, and customized placements for persons with disabilities:
 - (c) profit or not-for-profit Department of Health and Wellness funded long-term care facilities licensed under the *Homes for Special Care Act* or home care agencies funded under the *Homemaker Services Act*;
 - (d) persons providing care under the self-managed care program, supportive care program, caregiver benefit program funded by the province of Nova Scotia;
 - (e) persons providing support under the Independent Living Support, Supported Apartment and Supervised Apartment Programs funded by the Department of Community Services;
 - (f) homeless shelters receiving operational grants from the Department of Municipal Affairs and Housing, and those operated by religious and other voluntary organizations;
 - (g) unlicensed child-care facilities;
 - (h) a place designated or established under the authority of the Correctional Services Act or the Youth Criminal Justice Act (Canada) for the supervision or custody of offenders and includes community-based correctional services;
 - (i) any administrative tribunal, arbitration proceeding or court operating essential services in the Province under the authority of any provincial or federal enactment, including but not limited to, a justice centre or courthouse under the authority of the *Judicature Act* or a provincial court under the authority of the *Provincial Court Act* or the *Family Court Act*,

14

- (ia) notwithstanding clause (i), the Labour Board must still comply with section 13 when conducting a vote under the *Trade Union Act*.
- (j) Emergency Medical Care Incorporated;
- (k) persons providing, servicing or repairing medical equipment, such as wheelchairs, red cross beds/equipment, home oxygen equipment;
- (I) private not-for-profit community transportation providers;
- (m) food production plants; and
- (n) fishing vessels.
- 16.1 Physical distancing requirements and the person limit set out in sections 13, 13.1 and 13.2 do not apply to the following municipal entities and their contractors when carrying out their work duties:
 - (a) Police and Fire Services;
 - (b) Municipal Utilities such as water, wastewater and stormwater;
 - (c) Maintenance of utilities and municipal facilities;
 - (d) Transportation;
 - (e) Road maintenance/repair;
 - (f) Municipal ICT systems and services;
 - (g) Public Transit;
 - (h) Solid Waste, garbage and litter collection and disposal;
 - (i) Urban Forestry; and
 - (j) Municipal logistic, distribution, storage, inventory and repair services.
- 16.2 Physical distancing requirements and the person limit rule set out in sections 13, 13.1 and 13.2 do not apply to the following provincial entities and their contractors when carrying out their work duties:
 - (a) Transportation and Active Transit;
 - (b) Infrastructure and Housing
 - (c) Road maintenance/repair;

- (d) Government building construction and/or repair; and
- (e) Enforcement or compliance officers authorized by their statutory appointments or delegated authority to inspect, investigate and/or enforce provincial legislation while carrying out their powers pursuant to the relevant statutory authority.
- 17. Recognized businesses and organizations:
 - (a) may host day camps, indoors or outdoors, while adhering to the following:
 - (i) up to a maximum of 30 persons per group excluding staff and volunteers without distancing, or in multiple groups of 30 persons with physical distancing between groups; and
 - (ii) must follow the COVID-19 Return to Day Camp Guidelines established by the Chief Medical Officer of health, located at: https://novascotia.ca/coronavirus/docs/COVID-19-Return-to-Day-Camp-Guidelines.pdf.
 - (b) may host overnight camps, while adhering to the following
 - (i) up to a maximum of 15 persons per group excluding staff and volunteers without distancing, or in multiple groups of 15 persons with physical distancing between groups; and
 - (ii) must follow the COVID-19 Return to Overnight Camp Guidelines established by the Chief Medical Officer of Health, located at: covid-19-overnight-camping-guidelines-June14.pdf (novascotia.ca).
- 18. For the purpose of sections 18.2 -18.7, a "mask" means a commercial medical or non-medical mask or a home-made mask made as per the PHAC instructions located at: https://www.canada.ca/en/public-health/services/diseases/2019-novel-coronavirus-infection/prevention-risks/sew-no-sew-instructions-non-medical-masks-face-coverings.html, that covers the nose and mouth.
- 18.1 For the purpose of section 18.2, a "public place" means the part of the following places accessible to the public,

insofar as it is enclosed:

(a) a retail business, a shopping centre, or a building or room of a business where personal care services are provided;

The Royal Gazette, Wednesday, July 21, 2021

- (b) a restaurant or a liquor licensed establishment, including the kitchen and preparatory space of a restaurant or a liquor licensed establishment;
- (c) a place of worship or faith gathering;
- (d) a place where activities or services of a cultural or entertainment nature are offered:
- (e) a place where sports are played, fitness, recreational or leisure activities are carried on:
- (f) a rental hall or other place used to hold events, including conventions and conferences, or to hold receptions;
- (g) a place where municipal or provincial government services are available to the public;
- (h) a common area, including an elevator, of a tourist accommodation establishment:
- (i) a lobby, reception area or elevator in an office building;
- (j) a common area or public space on a university or college campus;
- (k) a train or bus station, a ferry terminal, or an airport;
- (m) common areas of a multi-residential building;
- (n) all common areas of private indoor workplaces;
- (o) private indoor workplaces where there are:
 - (i) interactions with the public;
 - (ii) areas with poor ventilation; or
 - (iii) areas where physical distance as set out in section 13 cannot be maintained,

and insofar as it is outdoors:

- (p) all serviced areas of a restaurant or a liquor licensed establishment, including their patios but excluding holes on the golf course that are licensed.
- 18.2 All persons must wear a mask that covers their nose and mouth while present in a public place.

- 18.3 Notwithstanding section 18.2, a person is exempt from the requirement to wear a mask in a public place if the person:
 - (a) is less than 2 years of age or age 2 to 4 years and their caregiver cannot persuade them to wear a mask;
 - (b) for whom the wearing of a mask is not possible because of the person's medical condition:
 - (c) is reasonably accommodated by not wearing a mask in accordance with the Nova Scotia Human Rights Act;
 - (d) is in the public place receiving care or being provided a service or while participating in a physical or other activity requiring the mask be removed, in which case the person may remove the mask for the duration of the care, service or activity;
 - (e) removes the mask momentarily for identification or ceremonial purposes;
 - (f) is in a courtroom, jury room or secured area in a courthouse, or room where a
 proceeding or meeting of an administrative tribunal established by legislation is
 being held;
 - (g) is an officiant in the course of performing activities requiring vocalization such as talking at the event or activity as described in sections 15, 15.1, 15.2 or 15.3; or
 - (h) is 12 years of age or less, attending a day camp, overnight camp, or
- 18.4 All persons must wear a mask that covers their nose and mouth as defined in section 18 while travelling on vehicles providing transportation to the public, including:
 - (a) any municipally operated public transit, including municipally operated buses and ferries:
 - (b) any public passenger vehicle licensed under the *Motor Carrier Act*, including community transit vehicles, commercial vehicles (shuttle vans), and vehicles providing charters and/or tours;
 - (c) any school buses licensed under the *Motor Carrier Act* and any vehicles of any capacity operated by private schools recognized by the Minister of Education and Early Childhood Development;

18

- (d) commuter vehicles and courtesy vehicles as defined under the *Motor Carrier Act*, vans, mini-buses, or buses of any passenger capacity providing services to the public;
- (e) any vehicles serving residents and staff of facilities listed in section 20.1; and
- (f) taxicabs regulated by municipalities under the authority of the *Motor Vehicle Act*.
- 18.5 Notwithstanding section 18.4, the following persons are exempt from the requirement to wear a mask while travelling on vehicles providing transportation to the public:
 - (a) a person is less than 2 years of age or age 2 to 4 years and their caregiver cannot persuade them to wear a mask;
 - (b) a person for whom the wearing of a mask is not possible because of the person's medical condition; and
 - (c) a person who is reasonably accommodated by not wearing a mask in accordance with the Nova Scotia Human Rights Act.
- 18.6 A person may remove the mask momentarily for identification purposes when boarding any public transit set out in section 18.4.
- 18.7 For greater certainty, the requirement to wear a mask, as set in sections 18.2 and 18.4, is the minimum standard that persons and businesses must adhere to, and where the business's plan approved in accordance with section 26.1 of this Order imposes a greater standard, then that standard applies.
- 19. Childcare facilities and family childcare homes regulated under the Early Learning and Child Care Act:
 - (a) may remain open; and
 - (b) directors, caregivers, staff and visitors must wear a mask as defined in section 18, unless subject to one of the exemptions as set out in section 18.3.

PART III LONG TERM CARE FACILITIES AND OTHER VULNERABLE POPULATIONS

- 20. Effective 8:00a.m. July 14, 2021, except where otherwise stated in this Order, the restrictions on long term care facilities and other vulnerable populations apply to all persons present and residing in Nova Scotia.
- 20.1 Notwithstanding section 16(c) and subject to section 20.2, all for-profit or not-for-profit Department of Health and Wellness funded long-term care facilities licensed under the Homes for Special Care Act must comply with Schedule "A", "COVID-19 Management Long term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at:

https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf:

20.2 All long term care facilities licenced by the Department of Health and Wellness under the *Homes for Special Care Act*, and all residents of such facilities must comply with the process for isolating COVID-19 positive long term care residents set out in Schedule "A", "COVID-19 Management Long Term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at:

https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf.

- 20.3 All long term care facilities licensed by the Department of Health and Wellness under the *Homes for Special Care Act* are open to family for visits, contracted service providers performing non-essential long term care work, 2 designated caregivers and volunteers, in accordance with the terms and conditions set out in Schedule "A", "COVID-19 Management Long term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at: https://novsscotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf.
- 20.4 All residents in homes licensed by the Minister of Community Services under the Homes for Special Care Act may have visitors so long as approved by the home in which they reside and comply with all general public health measures required under this Order.
- 20.5 All residents in homes licensed by the Minister of Health and Wellness are permitted to have community access in accordance with the terms and conditions set out in Schedule "A", "COVID-19 Management Long term Care Facilities Directive Under the Authority of the Chief Medical Officer of Health", dated December 21, 2020, as amended from time to time and located at:

https://novascotia.ca/dhw/ccs/documents/COVID-19-Management-in-Long-Term-Care-Facilities-Directive.pdf.

- 20.6 All residents in homes licensed by the Minister of Community Services under the Homes for Special Care Act are permitted to have community access as long as they comply with all general public health measures required under this Order.
- 20.7 For the purpose of section 20.8, "Adult Day Program" means a planned program of activities in a professional care setting designed for older adults who require supervised care during the day, or those who are isolated and lonely. Adult day care centers enable seniors to socialize and enjoy planned activities in a group setting, while still receiving needed health services. At the same time, they offer family caregivers respite from caregiving duties while knowing that their loved one is in a safe place.
- 20.8 All senior's Adult Day Programs are open.
- 20.9 For greater clarity, nothing in this Order prevents the:
 - (a) discharge of a COVID-19 patient from a hospital to a long-term care or residential care facility;
 - (b) transfer of a COVID-19 patient from community to a long-term care or residential care facility; or
 - (c) return of a COVID-19 patient who has left a long-term care or residential care facility for healthcare services back to that facility after receiving treatment at a hospital.

PART IV BUSINESSES. ORGANIZATIONS AND PROFESSIONS

- 21A Effective 8:00a.m. July 14, 2021, except where otherwise stated in this Order, the restrictions on businesses, organizations and professions apply to all established businesses, organizations and professions carrying on business and operating in Nova Scotia.
- 21.1The Occupational Health and Safety Act, 1996, c.7, s.1, is hereby incorporated by reference and must be followed by all employers, contractors, constructors, suppliers, owners and employees and each shall take every precaution that is reasonable in the circumstances to ensure the health and safety of all persons at or near a workplace.
- 22. All not-for-profit and for-profit businesses and organizations operating or carrying on business in Nova Scotia may not carry out COVID-19 Point of Care Screening Tests (PCTs) unless the business or organization:

- (a) has obtained prior approval from the Office of the Chief Medical Officer of
- (b) complies with any direction issued by me, as Chief Medical Officer of Health.
- 23. Subject only to specific closures or limitations directed elsewhere in this Order, all not-for-profit and for-profit businesses and organizations operating or carrying on business in Nova Scotia may continue to operate but must implement physical distancing of 2 metres (6 feet) within all workplaces and meeting spaces, except where otherwise provided in this Order.
- 23.1 Any not-for-profit or for-profit business or organization carrying on business in Nova Scotia that cannot maintain the physical distancing requirements set out in section 13 of this Order must limit the number of customers or clients on its premises to no more than 25 persons at a time.
- 24. Restaurants and liquor licensed establishments may provide in-person dining while adhering to the following:
 - (a) all restaurants may offer in-person dining to a maximum of full legal capacity of the restaurant provided the operator:
 - (i) ensures no group of patrons shall have more than 25 persons at a
 - (ii) maintains a minimum physical distance of 2 metres (6 feet) between seats or a physical barrier between tables, booths and single seats;
 - (iii) ensures food is delivered to the table by staff;
 - (iv) ensures patrons remain seated at their table, except while obtaining bar service, waiting to be seated, going to the bathroom, and getting ready to leave;
 - (v) ensures patrons wear masks when not consuming food or alcohol; and
 - (vi) ensures staff (servers and kitchen) wear masks while carrying out their work duties,

and

- (b) all liquor licensed establishments, including liquor manufacturers (as defined in the *Liquor Control Act*) may offer in-person dining to a maximum of full legal capacity of the restaurant provided the operator provided the licensee:
 - (i) ensures no group of patrons have more than 25 persons per table.

21

- (ii) maintains a minimum physical distance of 2 metres (6 feet) between seats or a physical barrier between tables, booths and single seats;
- (iii) provides food and alcohol consumption only for in-seat service;
- (iv) ensures food is delivered to the table by staff;
- (v) ensures food and alcohol remains at the table;
- (vi) ensures patrons remain seated at their table, except while obtaining bar service, waiting to be seated, going to the bathroom, participating in a game/activity being held in the establishment and getting ready to leave;
- (vii) ensures patrons comply with the physical distancing requirements set out in section 13;
- $\mbox{(viii)}$ ensures patrons wear masks when not consuming food or alcohol;
- (ix) ensures staff (servers and kitchen) wear masks while carrying out their
- 24.1 Bingos that hold a bingo license must adhere to the same provisions as set out in section 24(b) for liquor licensed establishments and liquor manufacturers, and the Communities, Culture and Heritage Re-opening Guide for Bingo Operations, located at: https://novascotia.ca/coronavirus/docs/Bingo-operations-reopening-quidelines.pdf.
- 24.2 Liquor boutiques, distilleries, breweries, and wineries
 - (a) may operate at the maximum allowable occupancy of their retail space for purchase and sale for offsite consumption, while adhering to public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements;
 - (b) may offer in-person tastings and samplings in accordance with their approved Workplace COVID-19 Prevention Plan set out in section 26, including in their place of business and public markets; and
 - (c) may offer in-person dining in accordance with section 24(a)
- 24.3 Restaurants, liquor licensed establishments, liquor boutiques, distilleries, breweries, and wineries offering in-person dining must:
 - (a) collect and maintain records of each patron contact information that includes the following minimum data:

- (i) name(s);
- (ii) contact telephone number;
- (iii) date; and
- (iv) time

that the patron was at the restaurant or liquor licensed establishment;

- (b) retain information collected under section 24.3 for a period of 30 calendar days and destroy it on the following day immediately thereafter;
- (c) provide the information collected under section 24.3 to the Chief Medical Officer of Health or his delegate in electronic form; and
- (d) only use the information collected under section 24.3 for the purpose of provision to the Chief Medical Officer of Health in compliance with this Order and may not use it for any other purpose.
- 24.4 Notwithstanding other provisions of this Order, restaurants, liquor licensed establishments, liquor boutiques, distilleries, breweries, and wineries:
 - (a) may have musicians (professional or amateur) perform and the musicians must adhere to the gathering limits for professional musicians set out in section 15.3, and the Covid-19 Guidance for Musicians (Vocalists and Instrumentalists) located at: https://novascotia.ca/coronavirus/docs/COVID-19-Guidance-for-Vocalists-and-Instrumentalists.pdf; and
 - (b) allow patrons to participate in dancing, while ensuring patrons' adherence to the public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements.
- 24.5 Owner/operators of liquor licensed establishments and non-liquor licensed establishments, including community centres, charities or organized clubs, may host activities such as darts, pool, cards or bowling but in doing so, must adhere to requirements established by the Chief Medical Officer of Health located at: https://novascotia.ca/coronavirus/docs/COVID-19-Games-and-activities-licensed-establishments.pdf.
- 25. Regulated and unregulated health professions practicing may continue to practice and provide services within their scope of practice if they have a Workplace COVID-19 Prevention Plan that has been approved by the Chief Medical Officer of Health.

- 26. All not for-profit, for-profit businesses, organizations and professions named in Part IV of this Order shall, as a requirement of their ongoing operations, develop and comply with a Workplace COVID-19 Prevention Plan, as amended from time to time.
- 26.1 In addition to the physical distancing requirements and person limit rule set out in sections 13 and 23.1, the Workplace COVID-19 Prevention Plan required by section 26 shall address the followino:
 - (a) how to work and interact with customers or clients;
 - (b) physical distancing in the workplace;
 - (c) cleaning;
 - (d) equipment;
 - (e) preparing employees to return to work;
 - (f) preparing for customers or clients; and
 - (g) monitoring and communicating of plan,

and may be substantially similar to those proposals listed in Schedule "B", located at: https://novascotia.ca/coronavirus/docs/Schedule-B-HPA-Order.pdf for similar businesses or health professionals and must be made available for review by the Chief Medical Officer of Health.

- 26.2 All businesses or regulated health professionals that are a member of an association listed in Schedule "B" must adopt as their Workplace COVID-19 Prevention Plan the proposal submitted by their association and must be made available for review by the Chief Medical Officer of Health.
- 26.3 For greater certainty, in the case of conflict between a Workplace COVID-19 Prevention Plan and this Order, the more stringent provision applies.
- 26.4 Effective 8:00a.m. July 14, 2021, all Provincial Park campgrounds may operate at 100% capacity while adhering to the public health requirements set out in this Order.
- 26.5 Effective 8:00a.m. July 14, 2021, all privately operated campgrounds may operate at 100% capacity while adhering to the public health requirements of this Order and must operate in accordance with the terms and conditions as approved by the Chief Medical Office of Health referenced in Schedule "B" located at: https://novascotia.ca/coronavirus/docs/Schedule-B-HPA-Order.pdf.
- 27. Casino Nova Scotia (Halifax and Sydney locations) and Video Lottery Terminals may operate at 100% capacity while adhering to their approved Workplace COVID-19

Prevention Plan in accordance with section 26, and those that hold a liquor license may operate at normal hours of business as per their license.

- 28. All not-for-profit and for-profit fitness facilities such as gyms and yoga studios, sport and recreational facilities such as arenas, tennis courts, and large multipurpose recreation facilities may open and:
 - (a) operate at maximum capacity that can be achieved while adhering to the public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements;
 - (b) offer outdoor fitness classes and other recreational activities up to a maximum capacity of 25 persons; and
 - (c) operate indoor and outdoor pools at maximum that can be achieved while adhering to the public health requirements set out in this Order, including but not limited to the physical distancing and masking requirements.
- 28.1 Businesses and organizations that offer indoor recreation and leisure activities such as indoor play areas, arcades, climbing facilities, dance classes and music lessons may open and operate at 100% indoor capacity, while ensuring two metres (six feet) distance between participants and participants wear masks except during an activity that makes it difficult to wear a mask.
- 28.2 All not-for-profit and for-profit fitness facilities, recreation and leisure businesses, and organized clubs may host lessons for recreational activities or offer personal training, while adhering to the gathering limits set out in section 28.
- 28.3 Notwithstanding section 28 and 28.1, golf courses may operate at full capacity while ensuring players practice physical distancing between each group on the golf course.
- 29. All personal service establishments such as hair salons, barber shops, spas, nail salons and body art establishments may open for provision of personal services by appointment or drop-in, while operating within their approved Workplace COVID-19 Prevention Plan in accordance section 26 and adhering to the physical distancing requirements and person limit rule set out in this Order.
- 30. The Art Gallery of Nova Scotia and all museums and public libraries may open and operate at 100% capacity, while adhering to the physical distancing and other public health requirements set out in this Order.
- 31. For the purpose of sections 31.1 to 31.2, "retail business" means a business operating on an ongoing basis at a fixed location primarily selling goods or products for use or consumption by individual purchasers.
- 31.1 A retail business that engages in the retail sale, or rental, of items to the public

25

26

may operate at maximum capacity that can be achieved while adhering to the public health requirements set out in this Order, including but not limited to compliance with the following conditions:

- (a) in accordance with section 13, members of the public must be able to maintain a physical distance of at least 2 metres (6 feet) from every other person in the place of business, except those persons residing in the same household;
- (b) no patrons may be permitted to line up inside or line up or congregate outside of the establishment unless they are,
 - (i) maintaining a physical distance of at least 2 metres (6 feet) from other persons inside or outside the establishment; and
 - (ii) wearing a mask or face covering in a manner that covers their nose and mouth, unless they are entitled to any of the exceptions set out in section 18.3.

and

(c) the person responsible for the place of business must ensure that music is not played at the place of business at a decibel level that exceeds the level at which normal conversation is possible, and in any event no greater than 50 db.

PART V EXCEPTIONS

32. Under exceptional circumstances and under the authority granted to me as the Chief Medical Officer of Health under Part I of the *Health Protection Act*, I may exercise discretion to grant an exception to any term and condition of this Order.

PART VI PENALITIES

- 33. Any direction provided by a medical officer of health to a person, business, organization or other entity pertaining to COVID-19 and the terms and conditions of this Order must be followed.
- 34.Failure to comply with this health protection order may be considered a breach of this Order issued under the *Health Protection Act* and may result in penalties under the *Act*.

This Order remains in effect until notice is provided by myself, as Chief Medical Officer of Health, under the authority granted under Part I of the *Health Protection Act* and will be updated from time to time.

Signed

Dr. Robert Strang Chief Medical Officer of Health

War

Nova Scotia Department of Health and Wellness

cc The Honourable Iain Rankin, Premier of Nova Scotia
The Honourable Zach Churchill, Minister of Health and Wellness
Laura Lee Langley, Deputy to the Premier and Clerk of Executive Council
Dr. Kevin Orrell, Deputy Minister of Health and Wellness
Dr. Shelley Deeks, Deputy Chief Medical Officer of Health
Tina M. Hall, Legal Counsel, Nova Scotia Dept. of Justice